Investigation of Desiccant Air-Conditioning and Evaporative Cooling Systems for Agricultural Storage Applications

ムハンマド, ハミド, マフムード

https://doi.org/10.15017/1866337

出版情報:九州大学, 2017, 博士(工学), 課程博士 バージョン: 権利関係: **INVESTIGATION OF DESICCANT AIR-CONDITIONING** AND EVAPORATIVE COOLING SYSTEMS FOR AGRICULTURAL STORAGE APPLICATIONS

MUHAMMAD HAMID MAHMOOD



June 2017

Department of Energy and Environmental Engineering Interdisciplinary Graduate School of Engineering Sciences

KYUSHU UNIVERSITY

JAPAN

INVESTIGATION OF DESICCANT AIR-CONDITIONING AND EVAPORATIVE COOLING SYSTEMS FOR AGRICULTURAL STORAGE APPLICATIONS

A THESIS SUBMITTED IN PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR THE AWARD OF THE DEGREE OF

DOCTOR OF ENGINEERING (Dr.Eng.)

By

MUHAMMAD HAMID MAHMOOD



June 2017

Supervisor Assoc. Prof. Takahiko Miyazaki

Department of Energy and Environmental Engineering Interdisciplinary Graduate School of Engineering Sciences

KYUSHU UNIVERSITY

JAPAN

DECLARATION

I hereby declare that the work, which is being presented in the thesis entitled "Investigation of Desiccant Air-Conditioning and Evaporative Cooling Systems for Agricultural Storage Applications" submitted in partial fulfillment of the requirements for the award of the degree of Doctor of Engineering (Dr.Eng.), Interdisciplinary Graduate School of Engineering Sciences, Kyushu University, Japan is an authentic record of my own research work.

The research work presented in the thesis has not been submitted by me for the award of any other degree in this or any other university.

Muhammad Hamid Mahmood

June 2017

ACKNOWLEDGMENT

ALLAH the Almighty never spoils any effort. Every piece of work is rewarded according to the nature of devotion. I invoke ALLAH's blessings, peace for His Holy Prophet "MUHAMMAD" (Peace Be Upon Him), the messenger of ALLAH, the most perfect and exalted among and of even born.

The work presented in this thesis was accomplished under the inspiring guidance and dynamic supervision of my kind supervisor **Professor Takahiko Miyazaki**. I express my deepest sense of gratitude and indebtedness to him for his precious guidance and inspiration throughout this research work.

I am also indebted to **Prof. Shigeru Koyama**, **Prof. Bidyut B. Saha**, and **Prof. Kazuhide** Ito for evaluating this work and for their valuable comments and suggestions.

I express my gratitude to **Dr. Muhammad Sultan** for his valuable assistance, encouragement and guidance in my research.

I am thankful to all my laboratory members and staff for their help and cooperation. I am grateful to Student Support Center of Kyushu University for their kind support and assistance. I would also like to thanks **MEXT**: Japanese Ministry of Education, Culture, Sports, Science and Technology, for providing fully funded scholarship throughout the study.

I am grateful to all of my teachers, friends, and colleagues for their kind support and motivation during this study.

I am thankful and would like to express my gratitude to my beloved Father Haji Ismat Ullah and Brothers Haji Khalid Bhai, Shahid Bhai, Hafiz Zahid Bhai for their love, prayers and everything they have done for me. Also, I wish to thank my beloved wife Engr. Tahira for her moral support, patience, prayers and encouragement during my study. I wish to share my emotions to Hasan (son) and Fatima (daughter) for making my life beautiful. I also express my feelings to my affectionate Grandparents, Sisters and In-Laws for their love and prayers for me.

I could never express my emotions to my lovely brother **Rashid** who passed away during my college life. To me, you're a symbol of hardworking, goodness and thankfulness. At the end, I dedicate my PhD thesis to my mother **Mrs. Mumtaz Begum** - The most loving and caring mother ever. You left me alone during my doctoral studies, I love you and miss you Maa Jee..! May ALLAH (SWT) showers His blessings on both of you and grants the Jannah.

Muhammad Hamid Mahmood

CONTENTS

Summary	X
List of Figures	xii
List of Tables	xviii

Chapter 1.	Background of the Study1
1.1	Introduction
1.2	Insights of Storage
1.3	Motivation of the Study10
1.4	Scope and Objectives
1.5	Thesis Outline
1.6	Nomenclature
1.7	References
Chapter 2.	Review of Literature
2.1	Introduction
2.2	Storage Techniques
	2.2.1 Drying
	2.2.2 Natural and Artificial Farm Storage
	2.2.3 Iced and Ventilated Storage
	2.2.4 Refrigerated Cold Storage
	2.2.5 Controlled and Modified Atmosphere Storage
	2.2.6 Hypobaric and Hyperbaric Storage
	2.2.7 Heat Treatment Options
	2.2.8 Ultraviolet-C Irradiation Treatments
2.3	Vapor Compression based Air-Conditioning

2.4	Evaporative Cooling based Air-Conditioning	
2.5	Desiccant based Air-Conditioning	
	2.5.1 Single-Stage Desiccant Air-Conditioning	
	2.5.2 Multi-Stage Desiccant Air-Conditioning	37
	2.5.3 Solar Energy Operated Desiccant Air-Conditioning	
2.6	Hybrid System	41
	2.6.1 Evaporative Cooling based Hybrid System	41
	2.6.2 Desiccant based Hybrid System	42
2.7	Nomenclature	44
2.8	References	44
Chanter 3	Advances in Evanorative Cooling Systems	53
Chapter 5.	Auvances in Evaporative Cooming Systems	
3.1	Introduction	
3.2	Wet-Bulb Evaporative Cooling	55
	3.2.1 Direct Evaporative Cooling	
	3.2.2 Indirect Evaporative Cooling	
3.3	Dew-Point Evaporative Cooling	
3.4	Maisotsenko Cycle based HVAC Systems	
	3.4.1 Standalone Maisotsenko Cycle Air-Conditioning	60
	3.4.1.1 Chilled Ceiling and Displacement Ventilation	63
	3.4.2 Hybrid Maisotsenko Cycle Air-Conditioning	64
	3.4.3 Ejector Maisotsenko Cycle Air-Conditioning	66
	3.4.4 Desiccant Maisotsenko Cycle Air-Conditioning	
	3.4.4.1 Solid Desiccant System	
	3.4.4.2 Liquid Desiccant System	71
3.5	Nomenclature	74
3.6	References	74
Chapter 4.	Experimental Performance Evaluation of Desiccant Unit	
4.1	Introduction	
4.2	Experimental Section	90

	4.2.1 Adsorbent/Desiccant	
	4.2.2 Experimental Setup	
	4.2.3 Development of Equilibrium Conditio	ns93
	4.2.4 Adjustment of Operating Conditions	
	4.2.4.1 Regeneration Air Flow	
	4.2.4.2 Process Air Flow	
	4.2.4.3 Air Flow Rate	
	4.2.5 Experimental Procedure	
	4.2.6 Uncertainty Analysis	
4.3	Theory and Methods	
4.4	Results and Discussion	
	4.4.1 Performance Evaluation of Desiccant	Unit 102
	4.4.2 Development of Correlation	
4.5	Conclusions	
4.6	Nomenclature	
4.7	References	
Chapter 5	Optimization of Desiccant Air-Conditionin	g Systems for Storage of
	Food Products	
5.1	Introduction	
5.2	Development of Ideal Storage Zones	
5.3	Description of Proposed Systems	
5.4	Methods and Analyses	
5.5	Results and Discussion	
	5.5.1 Performance Evaluation of System-I a	nd II 141
	5.5.1 Performance Evaluation of System-I a5.5.2 Performance Evaluation of System-III	nd II141 and IV144
	5.5.1 Performance Evaluation of System-I a5.5.2 Performance Evaluation of System-III5.5.3 Performance Evaluation of System-V	nd II
	 5.5.1 Performance Evaluation of System-I a 5.5.2 Performance Evaluation of System-III 5.5.3 Performance Evaluation of System-V 5.5.4 Storage Systems 	nd II
5.6	 5.5.1 Performance Evaluation of System-I a 5.5.2 Performance Evaluation of System-III 5.5.3 Performance Evaluation of System-V 5.5.4 Storage Systems Concluding Remarks 	nd II
5.6 5.7	 5.5.1 Performance Evaluation of System-I a 5.5.2 Performance Evaluation of System-III 5.5.3 Performance Evaluation of System-V 5.5.4 Storage Systems Concluding Remarks Nomenclature 	nd II

Chapter 6.	General Conclusions	
6.1	General Conclusions	
6.2	Future Work	164
Appendix A	Scope of Nano/Micro Polymeric Materials in Farm Air Co	nditioning
	Applications	
A.1	Introduction	
A.2	Scope of the Study	
A.3	Materials and Methods	
	A.3.1 Materials	
	A.3.2 Data Analysis and Procedure	
A.4	Results and Discussion	
A.5	Conclusions	
A.6	Nomenclature	
A.7	References	
Appendix B	Strengths, Weaknesses, Opportunities, Threats (SWOT) A	nalysis of
	Low Cost Cooling Technologies for Agricultural Product S	Storage 184
B.1	Introduction	
B.2	Procedural Details	
B.3	Results and Discussion	
B.4	Conclusions	
B.5	Nomenclature	
B.6	References	
Appendix C	Energy Recovery in Condensers and Cooling Towers	
C.1	Maisotsenko Cooling Tower	
C.2	Maisotsenko Condenser	
C.3	Nomenclature	
C.4	References	

Appendix D	Advances of Psychrometric Renewable Energy in Power Turbines20)4
D.1	Maisotsenko Cycle Conception in Gas Turbines)5
D.2	Maisotsenko Humid Air Turbine Cycle20)8
D.3	Maisotsenko Air Bottoming Cycle	10
D.4	Maisotsenko Sub-Atmospheric Brayton Cycle2	12
	D.4.1 Compressor based System	13
	D.4.2 Ejector based System	14
D.5	Nomenclature	16
D.6	References	16

SUMMARY

Energy efficient and low-cost agricultural storage is a burning issue of 21st century where postharvest losses (PHL) are ranging from 25-50% of net agricultural produce. Lots of storage techniques have been adopted worldwide in this regard but these are either expensive or inefficient. Therefore, Desiccant Air-Conditioning (DAC) and Evaporative Cooling (EC) systems are investigated in this study in order to establish low-cost, energy-efficient and environmentally benign system. The study focuses on theory of agricultural storage and consequently performs series of experiments for the performance evaluation of desiccant unit. Simplified mathematical models of evaporative cooling are employed in order to opt-in these technologies into the proposed cooling system. The key features of the study are highlighted as follows:

Theory of agricultural storage as well as insights of storage are explained in the very first chapter keeping in view the complex mechanism of respiration, transpiration and fermentation. The PHL in existing agricultural/food production system are elaborated. Factors affecting the PHL in terms of products quality, quantity, storage life and mal/nutrition are correlated. Studied technologies are motivated vis-à-vis typical heating, ventilation and air-conditioning (HVAC) systems, and thermodynamics limitations are discussed. In the second chapter extensive literature has been reviewed for existing and advanced storage practices. In addition to the proposed storage, following storage techniques are reviewed: drying, ice, ventilation, refrigeration, HVAC, hypobaric, hyperbaric, heat treatment, ultraviolet irradiation, and controlled and modified atmosphere.

The chapter three is devoted for evaporative cooling technologies. Direct and indirect evaporative cooling (DEC/IEC) technologies are explained and compared. As all the conventional evaporative cooling is based on wet-bulb cooling conception therefore the present study emphasis on dew-point based advanced indirect evaporative cooling (named as Maisotsenko cycle or M-cycle cooling) for agricultural storage application. In addition to subjected application, the applicability of M-Cycle has also been studied and analyzed for various applications which include cooling tower, condenser, heat recovery from turbines using various cycles (see appendices). It has been found that direct evaporative cooling is only applicable for dry regions or climates where the prime objective is to control the temperature irrespective of air enthalpy. On the other hand indirect evaporative cooling including M-Cycle cooling can be utilized sensibly in order to reduce the air enthalpy and dry-bulb temperature simultaneously. It is worth mentioning that the benefits of M-Cycle integration are obvious due to its dew-point cooling approach.

In contrary to evaporative cooling systems, desiccant air-conditioning is investigated in chapter four for humid climates/regions where the prime objective is to reduce the humidity. An open-cycle experimental apparatus was setup for the performance evaluation of hydrophilic polymer based desiccant blocks. Series of experiments are conducted for various: ambient air conditions, regeneration conditions, cycle time, and switching time. Generalized root sum of squares method is used to calculate the experimental uncertainty, and experimental data can be reproduced within $\pm 2-3\%$ error. It is examined that when humid ambient air passes through the

desiccant unit deep dehumidification occurs. It shows huge potential of air dehumidification at low regeneration temperature (50-60°C). It is determined that under the relatively dry ambient air conditions equivalent heat of adsorption (q_{eq}) profile approaches to net value equals zero. The average effective dehumidification slightly increases under humid ambient air conditions by changing the switching time ratio from 1:1 to 2:3 due to higher process air relative humidity. However, it keeps decreasing with increase in switching time ratio for relatively dry ambient conditions. It has been found that the switching time depends on dehumidification amount, nature of application and operating conditions. From the bunch of experiments, it has been concluded that the switching time ratio of 1:2, 2:3 and 1:1 can be selected for the operation of DAC system for high, medium and low humidity operating conditions, respectively. The desiccant dehumidification process should follow isenthalpic line on psychrometric chart in an ideal scenario when there is no adsorption heat. The slope of dehumidification line on psychrometric chart is therefore modified for the realization of real desiccant dehumidification process based on experimental data. Consequently a simplified correlation is developed by which real desiccant dehumidification process can be predicted on psychrometric chart for polymeric desiccant. In addition the study also highlights the applicability of few polymer- and carbon- based nano and micro sorbents for agricultural storage (see appendices). The correlation leads toward the stead-state analysis of DAC systems.

In chapter five, six kinds of desiccant air-conditioning systems are proposed and analyzed for the summer conditions of Fukuoka (Japan). In each system latent load of AC was accomplished by desiccant unit whereas sensible load of AC was accomplished by M-Cycle evaporative cooling. Storage compatibility of agricultural products is crucial to maintain the quantitative and qualitative attributes. In this regard, three different compatible groups of postharvest agricultural products (fruits and vegetables) according to their temperature and relative humidity (RH) are established for the psychrometric presentation of ideal storage zones. In case of dried fruits the ideal temperature and RH zones are established using Guggenheim, Anderson and De-Boer adsorption model. The effects of temperature on the storage life of dried fruits are realized by which it is found that shelf life reduces 50% by increase of 10°C temperature. It has been found that the system without preevaporative cooling on regeneration side, require less regeneration heat due to the provision of regeneration air stream to the heat exchanger (HX) at higher dry-bulb and lower dew-point temperature. It also provides higher dehumidification and wet-bulb/dew-point effectiveness. Similarly system pre-cooling does not give positive outcomes even when HX is not integrated in to the systems. On the other hand, system with pre-evaporative cooling on regeneration side in the absence of MEC enables higher cooling capacity, supply air relative humidity and COP.

The study concludes that the evaporative cooling (preferably M-Cycle) systems should be considered on top priority for agricultural storage applications wherever these are thermodynamically and meteorologically applicable. When these are not applicable, thermally driven DAC systems could yield advance agricultural storage system which can control temperature and humidity distinctly irrespective to conventional compressor based AC systems. Therefore, integration of evaporative cooling unit(s) into DAC system will lead towards energy-efficient and reliable low-cost AC systems for various applications. However optimum operational conditions will need to be determined and regulated for particular application. Herein it has been concluded that one or other DAC system could be efficiently utilized for the storage of agricultural products.

LIST OF FIGURES

Figure 1.1 Typical postharvest chain of agricultural products (reproduced from Kader and
Rolle, 2004; Mishra and Gamage, 2007)3
Figure 1.2 Key factors affecting the product quality (reproduced from Mishra and Gamage,
2007)
Figure 1.3 The pictorial representation of mechanism of photosynthesis, transpiration and
respiration (reproduced from Burden and Wills, 1989)
Figure 1.4 Generalized overview of stability map of agricultural products (reproduced from
Labuza, 1975; Taoukis et. al., 1997)
Figure 1.5 Psychrometric comparison between working principle of VAC and DAC 12
Figure 2.1 Self-explanatory explanation of controlled atmosphere storage representing the
stored product respiration and controlling gadgets (Hoehn et al., 2009)27
Figure 2.2 Pictorial view of UV-C irradiation treatment for smooth surfaced pear fruit
(Washington State University, 2015)
Figure 2.3 Schematics of heat pump cycles for: (a) cooling mode; and (b) heating mode
(www.kendallcountyair.com)
Figure 2.4 Schematic of typical desiccant air conditioning system (reproduced from
Miyazaki, et. al., 2010)
Figure 2.5 Psychrometric comparison between single-stage & multi-stage desiccant
dehumidification and air conditioning
Figure 2.6 Schematic of solar powered desiccant air-conditioning system installed at
Freiburg, Germany (Henning, 2007)40
Figure 2.7 Schematic of solar thermal and electric energy operated desiccant air-
conditioning system installed at Tohoku University, Japan (Enteria et al., 2009)41
Figure 2.8 Schematics of single-stage solar hybrid desiccant air-conditioning system
installed at Central Queensland University, Australia (Baniyounes et al., 2013)

Figure 3.1 Principle operation of conventional evaporative cooling techniques showing
their cooling limit for: (a) DEC, and (b) IEC
Figure 3.2 Schematic diagram of Maisotsenko Cycle for: (a) old M-Cycle, (b) modified M-
Cycle, (c) psychrometric representation, and (d) sequential temperature decrement in wet-
channel
Figure 3.3 Overview of the Maisotsenko Cycle applications
Figure 3.4 Effect of inlet air conditions on the performance of standalone M-Cycle AC unit
for: (a) supply air temperature (reproduced from Anisimov et al., 2014a; Anisimov et al.,
2014b; Pandelidis and Anisimov, 2015a; Pandelidis et al., 2015a); and (b) dew-point
effectiveness (reproduced from Anisimov et al., 2014a)
Figure 3.5 Schematic and geometric representation of solar chimney and M-Cycle in a
passive cooling system (reproduced from Miyazaki et al., 2010a; Miyazaki et al., 2011a)
Figure 3.6 Schematic diagram of the hybrid M-Cycle AC (H-MAC) system (Anderson et
al., 2011; Duan, 2011; Kozubal and Slayzak, 2010)65
Figure 3.7 Schematic diagram of the ejector M-Cycle AC system (E-MAC) (reproduced
from Buyadgie et al., 2011)
Figure 3.8 Performance comparison between EAC and E-MAC systems at different
ambient conditions (reproduced from Buyadgie et al., 2011)67
Figure 3.9 Schematic diagram of solid desiccant M-Cycle AC (D-MAC) system
(reproduced from Miyazaki et al., 2011b)
Figure 3.10 Cooling performance of solid D-MAC system (reproduced from Miyazaki et
al., 2011b)
Figure 3.11 Schematic diagram of liquid desiccant M-Cycle AC (D-MAC) system
(reproduced from Gao et al., 2015b)
Figure 3.12 Effect of air temperature on the performance of the liquid D-MAC system
(reproduced from Gao et al., 2015b)
Figure 4.1 Pictorial and magnifying view of the desiccant block
Figure 4.2 Detailed illustration of experimental setup
Figure 4.3 Illustration of flow control unit of experimental setup

Figure 4.4 Experimental profiles of ambient air conditions on desiccant dehumidification
performance at 60°C regeneration temperature: (a) case-A, and (b) case-B103
Figure 4.5 Experimental profiles of ambient air conditions on desiccant dehumidification
performance at 40°C regeneration temperature: (a) case-A, and (b) case-B104
Figure 4.6 Influence of regeneration temperature on equivalent adsorption heat for: (a)
case-A, and (b) case-B106
Figure 4.7 Effect of regeneration temperature on net dehumidification and regeneration
performance for: (a) case-A, and (b) case-B. Negative and positive values indicate
regeneration and dehumidification cycle, respectively
Figure 4.8 Effect of regeneration temperature on average dehumidification of desiccant
unit109
Figure 4.9 Experimental performance evaluation of desiccant unit under case-C: (a)
temperature and relative humidity profiles (a) resulted desorption and adsorption profiles.
Figure 4.10 Experimental performance evaluation of desiccant unit under case-D: (a)
temperature and relative humidity profiles (a) resulted desorption and adsorption profiles.
Figure 4.11 Experimental performance evaluation of desiccant unit under case-E: (a)
temperature and relative humidity profiles (a) resulted desorption and adsorption profiles.
Figure 4.12 Effect of switching time ratios on net dehumidification
Figure 4.13 Temporal variation in wet-bulb and dew-point temperature depression115
Figure 4.14 The resulted dehumidification comparison at different switching time ratios.
Figure 4.15 Experimental investigation of slope of dehumidification
Figure 4.16 Experimental investigation of slope of dehumidification under varying
regeneration temperature
Figure 5.1 Simplified scheme of adopted approach for storage of food products
Figure 5.2 Psychrometric comparison among agricultural products, dried fruits, dried
foods/feed storage, greenhouse AC and humans' AC130
Figure 5.3 Effects of temperature on storage life of dried fruits

Figure 5.4 Schematics of proposed DAC systems along with psychrometric representation
(a)-(f) represents system (I)-(VI), respectively
Figure 5.5 Schematics elaborated the methodology adopted for the analysis of proposed
systems
Figure 5.6 Ambient temperature and relative humidity profiles of Fukuoka, Japan
Figure 5.7 Profiles of ambient temperature and relative humidity for the month of July. 140
Figure 5.8 Effect of regeneration temperature on supply air temperature and relative
humidity of S-I and S-II
Figure 5.9 Net dehumidification performance profiles of S-I and S-II under varying
ambient conditions
Figure 5.10 Effect of regeneration temperature on system's (S-I, S-II) thermal COP and
heat input
Figure 5.11 Dew-point and wet-bulb effectiveness comparison of MEC integrated in S-I
and S-II
Figure 5.12 Effect of regeneration temperature on supply air temperature and relative
humidity of S-III and S-IV
Figure 5.13 Net dehumidification performance profiles of S-III and S-IV under varying
ambient conditions145
Figure 5.14 Effect of regeneration temperature on system's thermal COP and heat input.
Eigene 5.15 Deve agint and wat hulk affectiveness comparison of MEC integrated in S.I.
Figure 5.15 Dew-point and wet-build effectiveness comparison of MEC integrated in S-I,
S-11, S-111 and S-1V
Figure 5.16 Effect of regeneration temperature on supply air temperature and relative
$\mathbf{F} = \mathbf{F} + $
Figure 5.17 Comparison of net dehumidification performance of all proposed systems. 148
Figure 5.18 Effect of regeneration temperature on system's (S-V, S-VI) thermal COP and
heat input
Figure 5.19 Psychrometric evaluation of supply air conditions of S-II for different
applications
Figure 5.20 Psychrometric evaluation of supply air conditions of S-IV for different
applications151

Figure 5.21 Psychrometric evaluation of supply air conditions of S-V for different
applications152
Figure A.1 Few particles microscopic image of (a) PS-I, and (b) PS-II (Sultan et al., 2015b,
Sultan et al., 2016a)
Figure A.2 Psychrometric representation of different humidity applications for determining
the adsorbent to air mass fraction173
Figure A.3 Schematic of the DAC system with psychrometric representation of ideal DAC
cycle for medium humidity applications
Figure A.4 Comparison of adsorption isotherms at 30°C for (a) PS-I, PS-II, ACP, ACF,
PSS-FS, PSS-MS; (b) PS-I and PS-II with Silica-gel
Figure A.5 Adsorption uptake fraction of adsorbents over silica-gel
Figure A.6 Comparison of isosteric heat of adsorption of PS-I and PS-II with silica-gel. 177
Figure A.7 The adsorbent to air mass fraction under varying regeneration temperature for
(a) high; (b) medium; and (c) low humidity conditions
Figure B.1 Hierarchy of SWOT analysis of three air-conditioning dilemma for agricultural
products storage187
Figure C.1 Schematic diagram for: (a) general cooling tower flow scheme, (b) CCT, (c)
close circuit MCT, and (d) open circuit MCT (reproduced from Anisimov et al., 2014c;
Gillan et al., 2011b; Khalatov et al., 2011; Morosuk et al., 2012)
Figure C.2 Psychrometric representation of cooling tower operation for: (a) CCT; and (b)
close/open circuit MCT (reproduced from Anisimov et al., 2014c; Gillan et al., 2011b)
Figure C.3 Experimental setup for performance comparison between the air-cooled and M-
Condenser (reproduced from Gillan et al., 2011a; Idalex Technologies, 2006; Maisotsenko,
Condenser (reproduced from Gillan et al., 2011a; Idalex Technologies, 2006; Maisotsenko, 2006)
Condenser (reproduced from Gillan et al., 2011a; Idalex Technologies, 2006; Maisotsenko, 2006)
Condenser (reproduced from Gillan et al., 2011a; Idalex Technologies, 2006; Maisotsenko, 2006)
Condenser (reproduced from Gillan et al., 2011a; Idalex Technologies, 2006; Maisotsenko, 2006)
Condenser (reproduced from Gillan et al., 2011a; Idalex Technologies, 2006; Maisotsenko, 2006)

Figure D.2 Maisotsenko humid air turbine (M-HAT): (a) simplified schematic diagram of	
M-HAT cycle; (b) operational scheme of M-Cycle shell and tube air saturator (Gillan and	
Maisotsenko, 2003) in M-HAT cycle	
Figure D.3 Schematic diagram of the Maisotsenko air bottoming cycle (M-ABC)	
(reproduced from Saghafifar and Gadalla, 2015b)211	
Figure D.4 Schematic diagram of the compressor based M-SAB cycle (reproduced from	
Khalatov et al., 2015; Maisotsenko et al., 2004; Maisotsenko et al., 2006)214	
Figure D.5 Schematic diagram of the ejector based M-SAB cycle (reproduced from	
Buyadgie et al., 2015)	

LIST OF TABLES

Table 4.1 Parametric description of desiccant block (Yoshida, 2014). 91
Table 4.2 Close and open cycle kinetics parameters values of desiccant block (Sultan et al.,
2014a)
Table 4.3 Data recording sheet. 98
Table 4.4 Experiments under varying regeneration temperatures and constant switching
time ratio105
Table 4.5 Experiments under varying switching time ratios and constant regeneration
temperature
Table 5.1 Compatible storage groups (I, II and III) of the agricultural products (ASHRAE,
2010; Kitinoja and Kader, 2002)129
Table 5.2 Recommended moisture contents of dried fruits for safe storage, shipping and
buying/consumption (Dauthy, 1995)130
Table A.1 The particle diameter of studied adsorbents. 170
Table B.1 SWOT analysis of air-conditioning technologies. 189

CHAPTER 1

BACKGROUND OF THE STUDY

Page | 1

CHAPTER 1

BACKGROUND OF THE STUDY

This chapter describes the fundamentals background about the necessity of agricultural products storage and utilization of desiccant air-conditioning system for the purpose. The key factors responsible for postharvest losses (PHL) are determined. The role of temperature, relative humidity for storage of agricultural products for extended period is highlighted. The complex mechanism of respiration, transpiration and fermentation of agricultural products is explained in order to maintain the maximum quantity and quality of agricultural products and to avoid malnutrition. Motivation towards the use of desiccant air-conditioning and evaporative cooling systems for product storage instead of conventional vapor compression systems is explained from thermodynamics and product quality point of view.

1.1 Introduction

World population has crossed 7.3 billion by 2015 (FAOSTAT, 2015), which will reach to 9.1 billion by 2050 (FAO, 2009). In this scenario, additional agricultural produce has to be grown to combat food shortage (Hodges et al., 2011). In contrary, there are huge postharvest losses (PHL) in the existing agricultural/food production system. According to a study (Lal Basediya et al., 2013), higher moisture contents (60-95%) are usually the key reason for PHL which ultimately shorten the shelf life. The PHL are the losses of quality and quantity of the agricultural products. It is important to mention that the level of postharvest losses is different in different countries. However, 20-30% can be set an average as reported by Atanda et al., 2011 and El-Ramady et al., 2015. In case of developing countries (particularly along the tropical belt)



Figure 1.1 Typical postharvest chain of agricultural products (reproduced from Kader and Rolle, 2004; Mishra and Gamage, 2007).

these losses may exceed from half of the actual produce (Atanda et al., 2011; Burden and Wills, 1989; El-Ramady et al., 2015; Sanzani et al., 2016). Such losses may occur at any stage throughout the postharvest chain such as shown in Figure 1.1 (Kader and Rolle, 2004; Mishra and Gamage, 2007). However, the PHL can be minimized by cooling and storing the agricultural products on-farm and/or ex-farm right after their harvest that ultimately will increase their shelf/storage life. In this context, it has been reported that the storage/shelf life of the products can be enhanced by avoiding the postharvest cooling delays such as the shelf life of lettuce can be extended to 12 days by keeping it under optimal environmental conditions within one hour

after the harvest (Agüero et al., 2014). Contrary, it has also been reported in the literature that the cooling delay of 3 hours at 37°C or 6 hours at 24°C after harvesting lowers the marketable quality and weight of the Japanese eggplants (Cantwell et al., 2012). Likewise, the cooling delay of just 3 hours after the harvest reduces the shelf life of the broccoli because of floret openings which ultimately affect its visual appearance (Brennan and Shewfelt, 1989) and cooling delay of 48 hours at 20°C shortened the storability of European plum due to increased internal breakdown (Guerra and Casquero, 2009).



Figure 1.2 Key factors affecting the product quality (reproduced from Mishra and Gamage, 2007).

From the above prospective, the key factors responsible for postharvest losses are determined as shown in Figure 1.2 (El-Ramady et al., 2015; Mishra and Gamage, 2007). It is important to mention that the preharvest and harvest factors cannot be avoided/minimized after the harvesting, however, the postharvest factors can be controlled to slow down the decay process in the agricultural products.

The management of the postharvest factors is crucial for extending the shelf/storage life of the harvested products with maximum quantity and quality. Otherwise, the inadequate consumption and/or availability of fruits and vegetables are causing about 2.7 million deaths per year globally, and contribute 1.8 % worldwide diseases (Lock et al., 2005). The rationale about insights of agricultural product storage is described in detail in heading 1.2.

1.2 Insights of Storage

Postharvest losses mainly depend on temperature and relative humidity (Mahmood et al., 2016). The postharvest agricultural products perform respiration and require certain level of oxygen as described by Eq. (1.1) (ASHRAE, 2010).

$$C_6H_{12}O_6 + 6O_2 \to 6CO_2 + 6H_2O + respiration heat$$
 (1.1)

During this process ambient air oxygen reacts with the reserve sugar/starch of the harvested products and breakdown it into carbon dioxide (CO₂), water and heat energy (2667 kJ) is released during this reaction. The water produced during this reaction (Eq. (1.1)) remained within the product tissue, however the CO₂ releases and resulted in 3-5% weight loss of the product (Rao, 2015). The ratio between the volume of carbon dioxide liberated to the volume of oxygen absorbed is known as respiratory quotient (R_q). The value of the respiratory quotient for aerobic respiration varies from 0.7 to 1.3 depending upon the type of substrate being oxidized (e.g. in case of carbohydrates $R_q = 1$) (Rao, 2015). Moreover, the rapid removal of respiration heat is always required from the package and/or cold storage. It is worthy to mention that every agricultural product enables specific respiration rate which can be calculated from the following relationship (ASHRAE, 2010).

$$R = \frac{10.7 \,\theta_1}{36 * 10^5} \,(1.8 \,T + 32)^{\theta_2} \tag{1.2}$$

where R and T are respiratory heat generation rate [W/g] and product temperature $[^{\circ}C]$, respectively. The parameters θ_1 , θ_2 are respiratory coefficients which varies product to product (ASHRAE, 2010; Becker and Fricke, 1996). It can be notice from Eq. (1.2) that respiration rate mainly depends on the temperature which influences the decay/aging process of the products. In simple words, higher the respiration rate, shorter will be the product shelf life, and lower the respiration rate, longer will be the product shelf life. The postharvest respiration rate in almost all fresh vegetables (instead of root crops) remained high initially for 1-2 days. Later, it quickly lowers and achieves the equilibrium rate. On the other hand, the fresh fruits which do not ripen during storage (e.g. citrus fruits, grapes etc.) have fairly constant respiration rate. However, the ripening of fruits (e.g. apples, pears etc.) during storage increases the respiration rate. It is important to mention that when fruits are stored at relatively higher temperature (about 10-15°C) the respiration rate first increases due to ripening and later decreases, contrary, at low storage temperatures (0°C) no ripening takes place and accordingly almost no respiration heat produces by the products (ASHRAE, 2010). So, the temperature is most important factor in extending the postharvest shelf life of the agricultural products as the respiration rate is the function of the temperature.

It is important to mention that good supply of fresh air (with oxygen concentration about 20%) is crucial for normal respiration process in the agricultural products (Burden and Wills, 1989). The oxygen concentration (<10%) may control the respiration rate and slow down the aging (Rao, 2015) but the sufficient level of the oxygen (>2%) is always required to avoid the fermentation (Burden and Wills, 1989; Mishra and Gamage, 2007; Rao, 2015). Fermentation is an anaerobic respiration ($R_q > 1.3$) process in which sugar from agricultural product breaks down into ethanol, CO₂, and heat energy (92 kJ) is liberated during this reaction (Eq. 1.3). The poor ventilation of the cold storage due to restricted fresh supply air may promote the fermentation. Fermentation causes the unpleasant flavor, decay and early aging in agricultural products (Mishra and Gamage, 2007; Rao, 2015). The excessive accumulation of the CO₂ around the agricultural products (due to poor ventilation) also causes the other physiological changes/disorders in them (Burden and Wills, 1989; Mishra and Gamage, 2007; Rao, 2015). In

this regard, black heart potato disease due to higher carbon dioxide and lower oxygen concentration during storage/shipment is well known in the literature (Davis, 1926; Boyd, 1951).

$$C_6 H_{12} O_6 \rightarrow 2C_2 H_5 OH + 2CO_2 + fermentation heat$$
(1.3)

In order to maintain the quality of agricultural products, the relative humidity is also an important parameter during transpiration of the postharvest products. The transpiration is loss of water from the postharvest agricultural products. It involves the transport of moisture through the skin, evaporation, and convective mass transport of moisture to the surroundings (Becker et al., 1996). The endothermic evaporation at product surface cools its surface, lowers the vapor pressure and thus resulting in reducing the transpiration. On the other hand, the respiration within the agricultural product increases the surface vapor pressure due to raising product temperature and consequently the transpiration increases (Gaffney et al., 1985). The loss of moisture affects the products' physical appearance (wilting, shriveling), flavor, texture (softening, juiciness, limpness, flaccidity, crispness etc.), net weight and ultimately nutritional value (Lal Basediya et al., 2013; Mishra and Gamage, 2007). Five to ten percent reduction in fresh weight of agricultural product due to excessive transpiration (moisture loss) damages the product quality (Burden and Wills, 1989). The high relative humidity cannot inhibit the moisture loss if the product temperature is not close to the air temperature (Paull, 1999). The net transpiration rate varies from product to product and can be influenced by air temperature, relative humidity, flow rate, surface air to volume ratio, nature of surface coating, atmospheric pressure, mechanical damage etc. (Mishra and Gamage, 2007; Rao, 2015). However, it varies mainly with relative humidity, air flow rate and skin mass transfer coefficient. In other words, water vapor pressure deficit (VPD) between the product and surrounding is the primary driving force for transpiration (Sultan et al., 2016). The transpiration rate (q) [ng/kg·sec] is a linear function of VPD and can be calculated by Eq. (1.4) (ASHRAE, 2010).

$$q = k_t \left(P_{\nu s} - P_a \right) \tag{1.4}$$

where the parameters P_{vs} [Pa], P_a [Pa] and k_t [ng/kg·sec·Pa] represents the saturated water vapor pressure, actual water vapor pressure and transpiration coefficient, respectively. The

transpiration coefficient (k_t) is considered constant for particular product to simplify the calculations. The pictorial representation of the mechanism involved in photosynthesis and transpiration in the growing plant, and the respiration and transpirations processes in the harvested product is shown in Figure 1.3 (Burden and Wills, 1989).

In case of dried agricultural products the respiration rate is very low as compare to fresh agricultural products. The dried products are more sensitive to the surrounding moisture conditions in the cold storage. If the moisture contents in the storage environment increases the product will absorb more moisture and becomes highly prone to the deterioration. Contrary, if the moisture contents decrease the product will lose moisture and its weight will be reduced. The reduction in product weight will result in economic loss (Pahlevanzadeh and Yazdani, 2005). In this regard, it is crucial to know about the equilibrium moisture content (MCe) of the product in order to identify the required storage conditions (T and RH). The MCe depends upon the temperature, water activity and the product type. The water activity (a_w) is a temperature dependent intrinsic property of the product. It represents the decimal form of the equilibrium relative humidity (RH_e) at given temperature and moisture content. The RH_e is the property of the atmosphere/storage environment (i.e. extrinsic property) in equilibrium with the product. The water activity becomes equal to the decimal form of the relative humidity at equilibrium moisture content condition (Wilhelm et al., 2004). In short, the main controlling factor towards the preservation of dried agricultural products is water activity. In this context, the dried fruits become unacceptably hard upon losing their moisture below 0.5-0.7 a_w (Kochhar and Rossell, 1982; Taoukis et. al., 1997). The self-explanatory stability map of agricultural products as a function of water activity is shown in Figure 1.4 (Labuza, 1975; Taoukis et. al., 1997). It can be seen from Figure 1.4 that the typical products (like starch based snack food, cake mixes, crackers etc.) at 0.3 a_w are more stable against the deterioration/spoilage due to lower rate of lipid oxidation, nonenzymatic browning, enzymatic activity and consequently no growth of molds, yeasts and bacteria. The further increase in the water activity results in higher probability of the food deterioration due to growth of certain microorganisms. The dried fruits in the range of 0.6-0.7 a_w can be affected by the microorganisms (some molds and yeasts). The fresh agricultural products possess higher water activity ($a_w \ge 0.95$) and their quality is mostly affected by the bacteria (Wilhelm et al., 2004).

From the above detail discussion, it is generally concluded that the temperature and relative humidity are the key controlling factors for the stable storage of fresh and dried agricultural products for extended period.



Figure 1.3 The pictorial representation of mechanism of photosynthesis, transpiration and respiration (reproduced from Burden and Wills, 1989).



Figure 1.4 Generalized overview of stability map of agricultural products (reproduced from Labuza, 1975; Taoukis et. al., 1997).

1.3 Motivation of the Study

The conventional vapor compression refrigeration (VCR) and/or air-conditioning (VAC) systems are usually being used for storage of agricultural products. Such systems can regulate temperature precisely whereas humidity is controlled indirectly (Paull, 1999; Sharkey and Peggie, 1984). Moreover, anaerobic respiration may be prompted due to unavailability of the oxygen as discussed in heading 1.2. Therefore, conventional systems are difficult to provide optimum conditions for the storage of agricultural products. It is mainly due to the uneven control of humidity and/or ventilation. These systems can also cause chilling injury, off-flavor and discoloration in agricultural products e.g. mango, banana, tomato, leafy vegetables etc. (Ndukwu and Manuwa, 2015; Olosunde et al., 2016). Chilling injury can cause internal browning, surface pitting and also give rise to decay vulnerability in tropical root and tuber crops

(Kitinoja and Kader, 2002). Apart from other quality losses, the loss of "Vitamin C" from the product become greater due to higher temperature, lower relative humidity and chilling injury during storage (Lee and Kader, 2000; Paull, 1999). Moreover, ninety percent of "Vitamin C" in human diet is supplied through agricultural products (Lee and Kader, 2000). It is crucial for prevention of scurvy (low red blood cells, gum diseases, skin bleeding), arteriosclerosis (restriction of blood flow to organs and tissues), cardiovascular diseases, some forms of cancer and maintenance of healthy skin, gums and blood vessels (Harris, 1996; Lee and Kader, 2000). In this regards, conventional vapor compression air-conditioning (VAC) might be considered in order to maintain the storage conditions. However, VAC systems consume primary energy and have certain thermodynamic limitations which include: limitation of fresh air, poor ventilation (excessive CO₂) etc. In addition the VAC system cannot control the temperature and relative humidity distinctly as depicted on Figure 1.5 (Sultan et al., 2015; Sultan et al., 2016). In this perspective, low-cost environmental friendly evaporative cooling technologies like direct evaporative cooling (DEC), indirect evaporative cooling (IEC)/M-Cycle evaporative cooling (MEC) have shown potential to provide storage conditions for particular agricultural products. However, these technologies cannot be used effectively for the storage of agricultural products under largely varying ambient conditions (particularly humid) due to limited cooling performance. The scope of DEC and IEC/MEC (for detail see chapter 2 and chapter 3) for the storage of wide range of agricultural products under varying environmental conditions can be extended by the integration of desiccant dehumidification. The desiccant air-conditioning (DAC) comprises of desiccant dehumidification cum evaporative cooling has ability to deal the latent and sensible load of airconditioning distinctly. Furthermore, the DAC system can regulate relative humidity by means of desiccant whereas in case of VAC system it is only possible by cooling below dew-point temperature. Furthermore, DAC system is an environmental friendly technology and enables zero global warming and ozone depletion potential. It is operated on thermal energy preferably solar heat and/or biogas/mass available at the farm. Its design possesses better ventilation and consequently may help to control respiration and transpiration. The existing DAC system have shown the worth for various applications e.g. buildings (Enteria et al., 2009; Enteria et al., 2010); supermarkets, schools, hotels, ice arenas, cold warehouses, hospitals, theaters (Dabrowski, 1998); greenhouses (Longo and Gasparella, 2015; Sultan et al. 2014); automobiles (Nagaya et al., 2006); marine ships (Guojie et al., 2012; Zhu and Chen, 2014), museums (Ascione et al.,

2009; Ascione et al., 2013); wet markets (Lee and Lee, 2013) product storage and preservation (Sultan et al. 2016). Therefore, the present study investigates the DAC and Evaporative Cooling systems for the storage of agricultural products.



Figure 1.5 Psychrometric comparison between working principle of VAC and DAC.

1.4 Scope and Objectives

The scope and objectives of the work presented in this thesis are as follows

- To investigate the practical feasibility of Maisotsenko Cycle based evaporative cooling systems for farm air-conditioning applications based on following conceptions:
 - Wet-bulb and dew-point evaporative cooling
 - Sensible load control limited to dry conditions
- To develop simplified methodology for performance evaluation of desiccant unit.
 - Various ambient and regeneration air conditions (five cases)
 - Desiccant response analysis and correlation formulation
- To investigate various desiccant air-conditioning systems' configurations for humid areas for agricultural applications.
 - Six DAC systems' configuration and analyses
 - Selection of regeneration temperature and system COP
 - Overall systems' applicability for different applications

1.5 Thesis Outline

This thesis comprises of seven chapters and four supporting appendices in order to achieve the scope and objectives of the presented work. The summary of each chapter is as follows

Chapter 1 describes the fundamentals background about the necessity of agricultural products storage and utilization of desiccant air-conditioning system for the purpose. The key factors responsible for postharvest losses (PHL) are determined. Importance of reducing the PHL on the extension of products storage/shelf life is highlighted and typical postharvest chain of agricultural products right from agricultural farm to the end user is presented. The complex mechanism of respiration, transpiration and fermentation of agricultural products is explained in order to maintain the maximum quantity and quality of agricultural products and to avoid malnutrition. The motivation towards the use of DAC for product storage instead of conventional vapor compression refrigeration machines is explained from thermodynamic and product quality point of view.

Chapter 2 presents the extensive literature review about the agricultural product storage techniques. First part of the chapter covers the conventional and advanced storage techniques. The comparative analysis is made between different types of environmentally benign solar dryers. The methodology involved in natural, artificial, iced and ventilated storage is briefly reviewed along with merit/demerits of each technique. Salient features of advanced storage techniques are also explained along with their limitations. Second part of chapter includes the insight review about the vapor compression refrigeration and/or air-conditioning, evaporative cooling based AC, thermally driven desiccant based AC systems for agricultural products storage. The working principle and features of these systems are elaborated in this regards. Moreover, the solar energy operated single-stage DAC, multi-stage DAC and hybrid systems are also reviewed in order to extend the scope of DAC system for product storage in multi-climatic regions.

Chapter 3 briefly explains the principle and features of advanced dew-point Maisotsenko Cycle evaporative cooling (MEC) in comparison with conventional evaporative cooling. The potential applicability of low cost environmental friendly evaporative cooling technologies like conventional direct and indirect evaporative cooling and M-Cycle evaporative cooling for agricultural products storage is discussed in conjunction with Appendix B. The broad spectrum applications of M-Cycle are also presented in this chapter. Various modifications in MEC design are discussed in order to investigate its applicability in humid regions. It is found that desiccant based MEC systems enable huge energy saving potential to achieve the sensible and latent load of AC in humid regions.

Chapter 4 presents the setup of an open-cycle experimental apparatus for desiccant AC applications. Honeycomb like cubical shaped desiccant blocks are used for experimental dehumidification analyses. The orientation of the each unit of the experimental setup is explained briefly along with pictorial representation. The operation of each unit of the experimental setup is also explained distinctly. The efficacy of the supplementary heating unit to the increase regeneration air stream temperature is highlighted. The procedure adopted during regeneration and process air experimentation is described and experimental data recoding sheet is also presented in this chapter. The generalized root of sum of squares method is used to calculate the experimental uncertainty. It is carried out for measured and/or calculated variables in order to determine the accuracy of the experimental data. The experiments are performed under different ambient conditions, and various regeneration air temperatures and switching time ratios (five cases). The net and average effective dehumidification performances of desiccant unit are determined and presented. Influence of regeneration temperature on equivalent heat of adsorption is ascertained. The temporal variations in wet-bulb and dew-point temperature differences are presented. The optimized switching time ratio between regeneration and dehumidification of desiccant unit is suggested. Various scenarios of steady-state adsorption are explored and possible options are figure out for the development of simplified correlation for desiccant performance evaluation. In this regard a novel correlation is developed on the conception of modification of isenthalpic slope of dehumidification line on psychrometric chart. The experimental validation of slope of dehumidification under varying regeneration temperature is also investigated.

Chapter 5 presents the experimental investigation of the combined effect of desiccant dehumidification and sensible cooling (via MEC or HX) for the storage of agricultural products. The ideal storage zones of agricultural products, dried fruits, dried foods & feeds in comparison with greenhouse growth and humans thermal comfort zones are established on the psychrometric chart. Six different configurations of DAC systems are proposed and their performance evaluation is made under the ambient conditions of Fukuoka-Japan. A simplified methodology is

developed for performance evaluation of proposed DAC systems under varying regeneration temperatures. The parametric and thermodynamic analysis of all the system configurations (S-I to S-VI) is made to investigate that which configurations could yield better system performance. The psychrometric evaluation of three optimized DAC systems is performed for the storage of agricultural products, dried fruits, dried foods & feeds in comparison with greenhouse and humans AC.

Chapter 6 includes the general conclusions of this study and suggested future work.

1.6 Nomenclature

a _w	water activity [-]
DAC	desiccant air-conditioning
DEC	direct evaporative cooling
IEC	indirect evaporative cooling
k _t	transpiration coefficient [ng/kg·sec·Pa]
MC _e	equilibrium moisture content [%]
MEC	Maisotsenko cycle evaporative cooling
Pa	actual water vapor pressure [Pa]
PHL	postharvest losses
$P_{\nu s}$	saturated water vapor pressure [Pa]
q	transpiration rate [ng/kg·sec]
R	respiratory heat generation rate [J/sec/g]
RH	relative humidity [%]
R _q	respiratory quotient [-]
Т	temperature [°C or K]
VAC	vapor compression air-conditioning
VCR	vapor compression refrigeration

1.7 References

Agüero, M.V., Viacava, G.E., Moreira, M.R., Roura, S.I., 2014. Delayed cooling or suboptimal storage temperatures reduce butterhead lettuce shelf-life. Int. J. Veg. Sci. 20, 59-77.

- Ascione, F., Bellia, L., Capozzoli, A., 2013. A coupled numerical approach on museum airconditioning: Energy and fluid-dynamic analysis. Appl. Energy. 103, 416-427.
- Ascione, F., Bellia, L., Capozzoli, A., Minichiello, F., 2009. Energy saving strategies in airconditioning for museums. Appl. Therm. Eng. 29, 676-686.
- ASHRAE, 2010. Handbook-Refrigeration. American Society of Heating, Refrigerating and Airconditioning Engineers, Inc., Atlanta.
- Atanda, S.A., Pessu, P.O., Agoda, S., Isong, I.U., Ikotun, I., 2011. The concepts and problems of post-harvest food losses in perishable crops. African J. Food Sci. 5, 603-613.
- Becker, B.R., Fricke, B.A., 1996. Simulation of moisture loss and heat loads in refrigerated storage of fruits and vegetables, in: Refrigeration Science and Technology Proceedings, New Developments in Refrigeration for Food Safety and Quality, Lexington, Kentucky, USA, pp. 210-221.
- Becker, B.R., Misra, A., Fricke, B.A., 1996. Bulk refrigeration of fruits and vegetables Part I: theoretical considerations of heat and mass transfer. HVAC&R Res. 2, 122-134.
- Boyd, A.E.W.,1951. The internal blackening of potatoes caused by bruising. J. Hortic. Sci. 26, 148-156.
- Brennan, P.S., Shewfelt, R.L., 1989. Effect of cooling delay at harvest on broccoli quality during postharvest storage. J. Food Quality. 12, 13-22.
- Burden, J., Wills, R.B.H., 1989. Prevention of postharvest food losses: fruits, vegetables and root crops. Food and Agriculture Organization of the United Nations, Rome, Italy. http://www.fao.org/docrep/t0073e/t0073e00.htm.
- Cantwell, M.I., Thangaiah, A., 2012. Acceptable cooling delays for selected warm season vegetables and melons, in: Acta Hort. (ISHS) 934, Lisbon, Portugal, pp. 77-84. Doi: 10.17660/ActaHortic.2012.934.6
- Dabrowski, A., 1998. Adsorption and its applications in industry and environmental protection studies in surface science and catalysis. Elsevier Sci. Appl. Ind. 120, 79-916.
- Davis, W.B., 1926. Physiological investigation of black heart of potato tuber. Botanical Gazette, University of Chicago. 81, 323-338.
- El-Ramady, H.R., Domokos-Szabolcsy, É., Abdalla, N.A., Taha, H.S., Fári, M., 2015. Postharvest management of fruits and vegetables storage, in: Lichtfouse, E., Sustainable

Agriculture Reviews, Springer, Switzerland, pp. 65-152. Doi: 10.1007/978-3-319-09132-7_2.

- Enteria, N., Yoshino, H., Mochida, A., Takaki, R., Satake, A., Yoshie, R., Mitamura, T., Baba, S., 2009. Construction and initial operation of the combined solar thermal and electric desiccant cooling system. Sol. Energy. 83, 1300-1311.
- Enteria, N., Yoshino, H., Satake, A., Mochida, A., Takaki, R., Yoshie, R., Baba, S., 2010. Development and construction of the novel solar thermal desiccant cooling system incorporating hot water production. Appl. Energy, 87, 478-486.
- FAO, 2009. How to feed the world in 2050. http://www.fao.org/fileadmin/templates/wsfs/docs/expert_paper/How_to_Feed_the_Worl d_in_2050.pdf
- FAOSTAT, 2015. http://faostat3.fao.org/browse/O/*/E
- Gaffney, J.J., Baird, C.D., Chau, K.V., 1985. Influence of airflow rate, respiration, evaporative cooling, and other factors affecting weight loss calculations for fruits and vegetables. ASHRAE Trans. 91, 690-707.
- Guerra, M., Casquero, P.A., 2009. Influence of delayed cooling on storability and postharvest quality of European plums. J. Sci. Food Agric., 89, 1076-1082.
- Guojie, Z., Chaoyu, Z., Guanghai, Y., Wu, C., 2012. Development of a new marine rotary desiccant air-conditioning system and its energy consumption analysis. Energy Procedia. 16, 1095-1101.
- Harris, J.R., 1996. Subcelluar Biochemistry, Ascorbic Acid: Biochemistry and Biomedical Cell Biology (vol. 25), Plenum Press, New York.
- Hodges, R.J., Buzby, J.C., Bennett, B., 2011. Postharvest losses and waste in developed and less developed countries: opportunities to improve resource use. J. Agri. Sci.149, 37-45.
- Kader, A.A., Rolle, R.S., 2004. The role of postharvest management in assuring the quality and safety of horticultural produce.152, FAO-UN, Rome.
- Kitinoja, L., Kader, A.A., 2002. Small-scale postharvest handling practices: a manual for horticultural crops, fourth ed. Postharvest Technology Research and Information Center, University of California, Davis.
- Kochhar, S.P., Rossell, J.B., 1982. A vegetable oiling agent for dried fruits. Int. J. Food Sci. Technol. 17, 661-668.
- Labuza, T. P., 1975. Sorption phenomena in foods: theoretical and practical aspects, in: Rha, C.K. (Eds.), Theory, Determination and Control of Physical Properties of Food Materials, Springer, Netherlands, pp. 197-219.
- Lal Basediya, A., Samuel, D.V.K., Beera, V., 2013. Evaporative cooling system for storage of fruits and vegetables-a review. J. Food Sci. Technol. 50, 429-442.
- Lee, S.H., Lee, W.L., 2013. Site verification and modeling of desiccant-based system as an alternative to conventional air-conditioning systems for wet markets. Energy. 55, 1076-1083.
- Lee, S.K., Kader, A.A., 2000. Preharvest and postharvest factors influencing vitamin C content of horticultural crops. Postharvest Biol. Technol. 20, 207-220.
- Lock, K., Pomerleau, J., Causer, L., Altmann, D.R., McKee, M., 2005. The global burden of disease attributable to low consumption of fruit and vegetables: implications for the global strategy on diet. Bulletin of the World Health Organization, 83, 100-108.
- Longo, G.A., Gasparella, A., 2015. Three years experimental comparative analysis of a desiccant based air-conditioning system for a flower greenhouse: Assessment of different desiccants. Appl. Therm. Eng. 78, 584-590.
- Mahmood, M.H., Sultan, M., Miyazaki, T., Koyama, S., 2016. Desiccant Air-Conditioning System for Storage of Fruits and Vegetables: Pakistan Preview. Evergr. Jt. J. Nov. Carbon Resour. Sci. Green Asia Strategy. 3, 12-17.
- Mishra, V.K., Gamage T.V., 2007. Postharvest physiology of fruit and vegetables, in: Rahman, M.S., (Eds.), Handbook of Food Preservation, CRC press, Taylor and Francis Group, Boca Raton, USA, pp. 19-50.
- Nagaya, K., Senbongi, T., Li, Y., Zheng, J., Murakami, I., 2006. High energy efficiency desiccant assisted automobile air-conditioner and its temperature and humidity control system. Appl. Therm. Eng. 26, 1545-1551.
- Ndukwu, M.C., Manuwa, S.I., 2015. Impact of evaporative cooling preservation on the shelf life of fruits and vegetable in South Western Nigeria. Res. Agric. Eng. 61, 122-128.
- Olosunde, W.A., Aremu, A.K., Onwude, D.I., 2016. Development of a solar powered evaporative cooling storage system for tropical fruits and vegetables. J. Food Process Preserv.40, 279-290. DOI: 10.1111/jfpp.12605. https://dx.doi.org/10.1111/jfpp.12605.

- Pahlevanzadeh, H., Yazdani, M., 2005. Moisture adsorption isotherms and isosteric energy for almond. J. Food Process Eng. 28, 331-345.
- Paull, R., 1999. Effect of temperature and relative humidity on fresh commodity quality. Postharvest Biol. Technol. 15, 263-277.
- Rao, C.G., 2015. Engineering for Storage of Fruits and Vegetables: Cold Storage, Controlled Atmosphere Storage, Modified Atmosphere Storage. Academic Press (Elsevier), USA.
- Sanzani, S.M., Reverberi, M., Geisen, R., 2016. Mycotoxins in harvested fruits and vegetables: Insights in producing fungi, biological role, conducive conditions, and tools to manage postharvest contamination. Postharvest Biol. Technol. 122, 95-105.
- Sharkey, P.J., Peggie, I.D., 1984. Effects of high-humidity storage on quality, decay and storage life of cherry, lemon and peach fruits. Sci. Hortic., 23, 181-190.
- Sultan, M., El-Sharkawy, I.I., Miyazaki, T., Saha, B.B., Koyama, S., 2015. An overview of solid desiccant dehumidification and air-conditioning systems. Renew. Sustain. Energy Rev. 46, 16-29.
- Sultan, M., Miyazaki, T., Koyama, S., Saha, B.B., 2014. Utilization of desiccant air-conditioning system for improvement in greenhouse productivity: a neglected area of research in Pakistan. Int. J. Environ. 4, 1-10.
- Sultan, M., Miyazaki, T., Saha, B.B., Koyama, S., 2016. Steady-state investigation of water vapor adsorption for thermally driven adsorption based greenhouse air-conditioning system. Renew. Energy. 86, 785-795.
- Taoukis, P.S., Labuza, T.P., Saguy I.S., 1997. Kinetics of food deterioration and shelf-Life Prediction, in: Valentas, K.J., Rotsten, E., Singh, R.P. (Eds.), Handbook of Food Engineering Practice, CRC press, Boca Raton, New York, pp. 367-409.
- Wilhelm, L.R., Suter, D.A., Brusewitz, G.H., 2004. Food & Process Engineering Technology. American Society of Agricultural Engineers (ASAE), St. Joseph, Michigan.
- Zhu, J., Chen, W., 2014. Energy and exergy performance analysis of a marine rotary desiccant air-conditioning system based on orthogonal experiment. Energy. 77, 953-962.

CHAPTER 2

REVIEW OF LITERATURE

CHAPTER 2

REVIEW OF LITERATURE

This chapter presents the extensive literature review about the agricultural product storage techniques, conventional vapor compression based AC, evaporative cooling based AC, thermally driven desiccant based AC and hybrid AC systems. The salient features of advanced storage techniques are described in depth along with their limitations. The insights review about various air-conditioning systems for agricultural products storage is made. Moreover, the solar energy operated single-stage DAC, multi-stage DAC and hybrid systems are also reviewed in order to extend the scope of DAC system for product storage in multi-climatic regions.

2.1 Introduction

The agricultural products (fruits and vegetables) are perishable in nature and possess short shelf life under ambient conditions. These products are cultivated and harvested during specific period/duration in a year. However, their market demand mostly remains throughout the year. In this scenario it is required to store the excess production in order to meet the supply and demand gap. The storage is also made to get higher/reasonable returns in off-season. Storage period can be short, mid or long term depending on product type, usage, perishability, internal characteristics, physiology etc. (Camelo, 2004; Rao, 2015). The highly perishable agricultural products like raspberries and other cultivars of berries can only be stored for short period. The products (e.g. mangoes, bananas, cabbages, brinjals, tomatoes etc.) can be stored for midterm period (few weeks) depending on the

produce quality, type and the market demand. However, the long term storage of onions, potatoes, sweet-potatoes, garlic, squash, pumpkins, oranges, apples can be made in order to attain the good market price. Moreover, the remaining surplus products, after fulfilling storage requirements, can be preserved by drying. The shelf life of dried agricultural products can be extended from six months to one year or even more by providing optimal storage conditions (temperature and relative humidity). Such controlled conditions are essential because dried agricultural products require certain percentage of storage moisture for technical, practical; and commercial point of view (Dauthy, 1995). The details about these storage/preservation techniques are explained under subsequent headings.

2.2 Storage Techniques

2.2.1 Drying

The drying is considered as an oldest technique for preservation of agricultural products (Dauthy, 1995). The simultaneous heat and mass transfer during the drying process result in removal of moisture from the product (El-Sebaii, 2012). The dried agricultural products possess longer shelf life and can be transported easily due to lighter weight. These products also required less space for storage (Dauthy, 1995; Ertekin, 2004).

The surplus amounts of postharvest fresh agricultural products are traditionally preserved by exposing directly to the sun. It is widely used common method of preservation in the developing countries. The products are spread over the floors and exposed to the natural solar energy directly for certain period of time depending on the product type, climatic conditions etc. (Greig and Reeves, 1985; Proctor, 1994). However, the adverse climatic conditions, attack of birds, insects/pests can deteriorate the quality and quantity of the naturally dried agricultural products (El-Sebaii, 2012). In this regards, advancements have been made in the solar drying methods in parallel to the traditional natural solar drying in order to minimize the losses (qualitative, quantitative, nutritive etc.). These drying methods can be broadly classified in to four different types (as follow) depending on the mode of energy supplied to the agricultural products for the removal of moisture (El-Sebaii, 2012; Furlan et al., 1983; Kumar et al. 2016).

- Natural Solar Dryer
- Direct Solar Dryer
- Indirect Solar Dryer
- Mixed/Hybrid Type Solar Dryer

In natural solar drying the agricultural products are placed on the ground or above ground surface. The plastic/tarpaulin sheets are usually placed beneath the products (Greig and Reeves, 1985). The products are exposed directly to ambient environmental conditions (solar energy, wind, rain) and infestation of birds, insects/pests etc. This method of drying is quite cheap but it causes higher loss of product quality and quantity (El-Sebaii, 2012; Fudholi et al., 2010). However, in direct solar dryer the products are placed in small wooden/metallic chambers/cabinets. The drying chamber is covered with transparent (polyethylene) sheet/cover. It is used to avoid the products from adverse environmental conditions, infestation of birds, insects/pests etc. (El-Sebaii, 2012; Fudholi et al., 2010). The products which are going to dry are placed inside the dyer chamber. The solar radiations pass though the sheet (polyethylene) and absorbed by products. Thus, the absorption of solar energy by the products results in removal of moisture from products and moved away due to air circulation. Though such dryers are cheaper and simple in construction but cannot be used widely due to limited product drying capacity and damaging its quality due to overheating (El-Sebaii, 2012; Kumar et al. 2016).

The indirect solar dryer overcome the demerits associated with the direct solar dryer. It consists of solar collector, drying unit, air ducting, and, blower (in case of active indirect solar dryer) (Esper and Mühlbauer, 1998). The solar collector and drying units are separately placed and connected with each other through ducting unit. The air heated by the solar collector is moved to the drying unit/chamber through air ducting unit. The better operating conditions (higher temperature, air mass flow rate etc.) can be achieved by this type of dryer.

In mixed/hybrid type solar dryer the products receive direct solar energy (as in direct solar dryer) and pre-heated air is also supplied to the product. The supply of pre-heated air is made by solar/electric or bio-mass/bio-gas heater etc. (Kumar et al. 2016). The rapid drying of agricultural products can be made by such type of dryers. In this regard, Prasad et al., 2006, developed a hybrid solar cum biomass energy operated dryer for the

drying of turmeric. This hybrid system dried the turmeric rhizomes in only 1.5 days, whereas traditional natural solar drying took 11 days for such drying.

2.2.2 Natural and Artificial Farm Storage

In natural storage method the crops belong to roots and tubers group are left underground for several weeks/months. Some of root and tuber crops include carrot, sweetpotato, cassava, yam, garlic and potato etc. (Rao, 2015). These crops are harvested before the start of rainy season and/or when the demand arising from the markets in order to get the good market value (price) of the produce (Camelo, 2004). This storage technique is very cheap because it does not require the separate storage structure/building. So it does not involve any fixed and/or variable cost of operation. However, the products quality is affected by insects/pests and disease attack.

In the artificial storage method the products (e.g. onion, potato, sugar beet, sweetpotato, carrot turnip etc.), unlike the natural farm storage, are first harvested from plant and then store on farm. These products are stored by making heap/mound like structure and/or by digging a trench/pit. The products are protected from field moisture and weather conditions by covering with straw, plastic-sheets, tarpaulin (polyethylene), and soil (Rao, 2015). The extra available bin/barrel filled with products can also be buried in the soil for the purpose (Kitinoja and Kader, 2002). Another alternative of this method is to place the products container/bins on top of each other in the field (Camelo, 2004). The topmost container is finally protected from weather. This alternative facilitates the mechanized handling of the containers.

2.2.3 Iced and Ventilated Storage

In iced storage method, the ice was used (in the warehouse) for the preservation of meat, fish and perishable agricultural products. The ice was harvested in winter from the frozen lakes/ponds. It was stored in the ice storehouses and utilized as refrigerant in the product's cold storages when needed. The ice starts melting by absorbing product heat. The water discharge due to ice melting is a major drawback of this storage technique. About 325 kJ heat is absorbed due to melting of 1 kg ice (Rao, 2015). The small size ice box, now

mostly used for transporting food/medicine at domestic and/or commercial level, is an advance form of iced storage.

The ventilation of the stored products is utmost important for the removal of the heat and humidity produced due to respiration/transpiration. In this regard natural or forced air ventilation can be employed in the storage structure. The storage structure can be constructed on ground and underground. Cellar is an example of underground ventilated storage structure. Cellar provides good storage conditions in the cold and snowy regions. It is mostly built for storage of products like potato, onion, turnip, garlic, sugar beet etc. It can be constructed into hillside with shingle roofs covered by sods and soil (Bubel and Bubel, 1979; Hobson, 1981; Rao, 2015).

2.2.4 Refrigerated Cold Storage

Cold storages are the structures (buildings) equipped with refrigeration system along with necessary facilities and gadgets. The conventional vapor compression refrigeration (VCR) systems are mostly used in cold storages facilities. The main purpose of cold storage is to provide the storage conditions (temperature, relative humidity, ventilation) to the agricultural products for specific period. Usall et al., 2015 stated that storing the agricultural products at low temperature and high relative humidity in the cold storage is not an antifungal treatment; however, it can directly and indirectly influence the host and pathogen. According to Usall et al., 2015, the age delay is mainly due to: (i) loss of pathogens growth and corresponding enzymatic functions (straightforward influence) and (ii) metabolism reduction in the product (indirect influence). This fact keeps the resistance against infections particularly the caused by fungus. It is the most adopted physical method for the storage of agricultural products (Eckert and Sommer, 1967).

The cold storage building can be constructed using different materials like concrete, metal, wood etc. The walls, floors and ceilings should be well insulated. The insulation materials (e.g. polyurethane, cork etc.) can be used for the purpose. It is generally common that 75-80% of the total surface area of the building is used for products storage (Camelo, 2004). Rest of the surface area is usually occupied by walkways and aisles etc. The height of the product chamber depends on product type and stacking pattern. It varies from 3 meters to 6 meters for hand stacking and forklift stacking, respectively.

2.2.5 Controlled and Modified Atmosphere Storage

The separate storage chambers are provided in the conventional cold storage for controlled atmosphere provisions to the produce. In controlled atmosphere storage (CAS) the low temperature air with varying gas composition (O_2 and CO_2) is supplied to the fruits and vegetables. The low (1-3%) and high (4-5%) concentration levels of oxygen and carbon dioxide are usually provided around the products, respectively. The levels of these air gases mostly depend upon the product type and its variety too. CAS facilities are equipped with specialized controlled equipment like gas generators and carbon dioxide scrubbers in order to provide the different ratios of O₂ and CO₂ such as shown in Figure 2.1 (Hoehn et al., 2009). Rao, 2015 stated that the higher concentration of carbon dioxide is considered as antiseptic against the vital activity of phytopathogenic microorganism. The antiseptic properties of CO_2 appear at very high concentration in the air. However, the concentration of CO₂ higher than permissible level may result in browning of product tissue. It has been reported that CAS provides better storage conditions as compare to conventional cold storage particularly to apple fruit (Nilsson et al., 1956). It is due to reduction in postharvest decay (Nilsson et al., 1956), and considerably less existence of blue molds (Sams and Conway, 1987).

The modified atmosphere storage/packaging (MAP) is used to extend the storage life of the fresh and minimally processed agricultural products. In this technique the produce is packed in pouches/bags and containers made by polymeric material. The air composition inside the bag around the product is changed to another composition depending upon the product type, storage temperature and packaging material (Rao, 2015). The higher level of CO_2 and lower level of O_2 in the modified atmosphere package decreases the product respiration rate, consequently the product with extended shelf life can be transported easily for long distances (Barkai-Golan, 2001). In this regard, the export of sweet cherries (from turkey) through this storage technique has been reported in the literature (Barkai-Golan, 1990). Such storage technique preserve the fruit quality (color, firmness) and quantity (water loss, shriveling) (Karabulut et al., 2010). The products packed in modified atmosphere packaging are stored in conventional cold storage. Therefore it does not involve any extra cost for the construction of cold storage building (Rao, 2015).

The controlled and modified atmosphere storage techniques both can extend the particular products shelf life by delaying the postharvest decay. The modified atmosphere packaging can be used for the storage of variety of products. However, the main limiting factors in the adoptability of controlled atmosphere storage are the requirement of skilled labors/technicians and specialized facilities (equipments). Finally the cost benefit ratio plays the crucial role in deciding the storage technique (Usall et al., 2015). Such factors limits it applicability for few fruits like apples, kiwifruit, persimmon, and pomegranate etc. (Tonutti, 2015).



Figure 2.1 Self-explanatory explanation of controlled atmosphere storage representing the stored product respiration and controlling gadgets (Hoehn et al., 2009).

2.2.6 Hypobaric and Hyperbaric Storage

Hypobaric storage is also termed as low pressure or sub-atmospheric storage in the literature (Thompson, 2015). The hypobaric pressure (0-100 kPa absolute) is applied to the fruits and vegetable for short period of time (Goyette et al., 2007; Romanazzi et al., 2001). Such pressure application induces the natural resistance in agricultural products against the

postharvest decay. Workman et al., (1957) reported that the respiration rate of tomatoes when stored at 20°C and 10.67 kPa was reduced as compared to tomatoes stored at 20°C under atmospheric pressure. He is probably the first who records the hypobaric conditions which were applied to the agricultural products (Thompson, 2015). The hypobaric chambers, used for products storage, always required proper ventilation control in order to maintain the permissible oxygen level. It is because of continues respiration and ethylene production by fruit and/or vegetables. Otherwise, rising levels of ethylene can rapidly deteriorate the quality of stored products. Therefore, the hypobaric chambers are equipped vacuum pump, which continuously evacuate the chamber air and replenish it with outside fresh air. Thompson, 2015 stated that in order to apply the hypobaric storage technology, the store structure must be constructed strong enough (e.g. with steel sheet) so that it can withstand the range of low pressure without any collapse. Furthermore, the air supply to the store must be saturated (relative humidity about 100%) otherwise application of low pressure may result in dehydration (rapid loss of moisture) of agricultural products.

Chen et al., 2013, experimentally investigated the effects of hypobaric storage on the quality of bamboo shoots. In this regard, the bamboo shoots were placed in the storage chamber of hypobaric storage system for 35 days at pressure and temperature of 50 kPa and $2(\pm 1)^{\circ}$ C, respectively. Chen et al., 2013, reported that the hypobaric pressure and low temperature storage of bamboo shoots not only maintain their quality till the end of storage period (35 days), nonetheless, some attributes were also improved by this storage technique.

In hyperbaric storage, unlike the hypobaric pressure, higher pressure than atmospheric pressure is applied to the stored agricultural products. The provision of pressure slightly higher than atmospheric pressure is not much explored yet, however, its higher range application (about 405 to 1215 MPa) has been broadly studied for agricultural products (Thompson, 2015). The hyperbaric pressure is also applied for short period of time to the agricultural products. Romanazzi et al., 2008 investigated that the postharvest storage/shelf life of sweet cherries and table grapes can be extended by exposing them to pressure of 152 kPa for 4 and 24 hours. The hyperbaric chambers are mainly made for medical and diving industries; however, they are also used for extending shelf life of agricultural products (Thompson, 2015).

2.2.7 Heat Treatment Options

The heat treatment is leading non-damaging physical treatment to avoid the postharvest decay in agricultural products due to insects, pests, and diseases (Ji et al., 2012; Lurie and Pedreschi, 2014). The heat treatments can extend the shelf life of agricultural products due to its effects on product ripening, senescence, and chilling injury (Sisquella et al., 2014). Lurie, 2008 stated that the heat treatments can inhibit the ethylene production in climacteric fruits which consequently delay the ripening, softening and discoloration. It can also reduce the softening rate and colour development in case of climacteric fruits (ripening does not depend on ethylene production). The yellowing of the green vegetables due to senescence (aging) can be reduced by heat treatments because of its effect on the physiology of product. Moreover, the heat treatments induce the resistance in subtropical agricultural products against the susceptibility to chilling injury. Thus the subtropical agricultural products after such treatments can be stored at low temperature, and transported through ships.

The heat treatments can be performed by hot air (vapor heat and forced air), hot water (hot water dipping and brushing), and dielectric heating (radio frequency and microwave) (Lurie, 2008; Usall et al., 2015). In case of hot air treatment, air at temperature and relative humidity higher than 30°C and 90% respectively, is usually provided to the fruits (Usall et al., 2015). The hot air treatment also reported in the literature for curing citrus fruit against rind/peel wound and green mold (Ben-Yehoshua and Porat, 2005; Hopkins, and Loucks, 1948). However, the hot water treatment is considered advantageous over hot air treatment due to the provision of efficient heat transfer with less time. It is also cheaper than other options of the heat treatments (Jacobi et al. 2001; Sivakumar, and Fallik, 2013). The water at temperature higher than 40°C is used in hot water treatment for certain period of time depending on several factors (Usall et al., 2015). Small scale hot water treatment plant was proposed under trade related technical assistance programme (TRTA II) in Pakistan for the treatment of fruits fly infestation of mangoes (www.trtapakistan.org).

The dielectric heating is also used for rapid heat transfer to the agricultural products, though the limited literature is available in this regard (Casals et al., 2010).

2.2.8 Ultraviolet-C Irradiation Treatments

The ultraviolet-C (UV-C) irradiation of wavelength ranging from 190 nm to 280 nm has shown promising results against the postharvest decay in agricultural products (Romanazzie et al., 2006; Usall et al., 2015). It is naturally present in the sunlight, but absorbed completely by ozone layer and earth atmosphere (Washington State University, 2015). It induces the resistances in the host against the pathogens. However, such induced resistance starts decreasing with the ripening of fruits (D'Hallewin et al., 1999). This treatment has been experimentally tested in order to inhibit the fungal growth and/or aging in different types of fruits and vegetables from last few decades (Usall et al., 2015). The application of low dose of UV-C irradiation has resulted in effective control over postharvest diseases in sweet potatoes (Stevens et al., 1990), peaches and apples (Lu et al., 1991), carrots (Mercier et al., 1993), grapefruits (Droby et al., 1993), strawberries (Nigro et al., 2000), and tomatoes (Liu et al., 2012) etc. as explained in the respective literature (Usall et al., 2015). Moreover, the delayed aging of peach fruit treaded with this treatment has also been reported (Yang et al., 2014).



Figure 2.2 Pictorial view of UV-C irradiation treatment for smooth surfaced pear fruit (Washington State University, 2015).

Adhikari et al., 2015 experimentally investigated the effectiveness of UV-C irradiation treatment on the surface of various fresh fruits against the foodborne pathogens. In this regard, the smooth surface (apples, pears) and rough surface (strawberries, red raspberries, cantaloupe melons) fresh fruits were exposed to the UV-C irradiation for certain period of time (10 sec-840 sec) depending on different factors (e.g. fruit type, UV-C dose, etc.). The study concludes that UV-C irradiation treatment is highly effective against the food borne pathogens in case of smooth surface fruits. However, this treatment is found less effective for rough surface fruits because these fruits provide more hiding places to the pathogens. The pictorial view of UV-C irradiation treatment on pears is shown in Figure 2.2 (Washington State University, 2015).

2.3 Vapor Compression based Air-Conditioning

The importance and necessity of the conventional vapor compression refrigeration and/or air-conditioning for various applications/processes (at industrial, institution commercial and domestic etc. levels) is well known in the literature. The air-conditioning meanly deals with the provisions of conditioned air for human thermal comfort. However, its broad spectrum application also includes the conditioning of air for industrial processes, food processing, storage of agricultural products and materials (Khurmi, Gupta, 2008), and livestock thermal comfort. The conditioning of air means its temperature, relative humidity, purity and circulation.

The conventional vapor compression refrigeration (VCR) system mainly consists on an evaporator, condenser, compressor, expansion valve and fan (blower). In case of agricultural product storage, the evaporator coils and blowers are integrated into single unit and placed inside the storage room in order to provide the cold forced air circulation throughout the storage facility (Rao, 2015). The air becomes hot (due to heat of respiration) as it passes through the stored products. This air is then returns and passes over the evaporator to exchange the heat. The refrigerant (e.g. ammonia, freon etc.) inside the evaporator coils absorb this heat and cold air is again deliver to the stored produce. The phase changed refrigerant (liquid to vapor) then passes through the compressor. The compressor compresses the refrigerant (vapor), increased it pressure and temperature and transported it to the outside condenser in order to release the product heat to the atmosphere. The refrigerant (vapor to liquid) at high pressure enters to expansion valve. Now the refrigerant (vapor) is throttled down at low pressure to the evaporator in order to complete the refrigeration cycle. During this cyclic process the expansion valve controls the evaporation and flow of refrigerant (Camelo, 2004).



Figure 2.3 Schematics of heat pump cycles for: (a) cooling mode; and (b) heating mode (www.kendallcountyair.com).

The conventional vapor compression air-conditioning (VAC) systems might be considered in order to maintain the storage conditions for agricultural products. However, these systems consume primary energy and have certain thermodynamic limitations which include: limitation of fresh air, poor ventilation (excessive CO_2) etc. Moreover, such systems also cannot provide the optimal storage conditions (temperature and relative humidity) particularly to the tropical agricultural products and dry fruits (Mahmood et al., 2016a, Mahmood et al., 2016b). In addition the conventional vapor compression airconditioning cannot control the temperature and relative humidity distinctly as compare to desiccant air-conditioning (Sultan et al., 2015; Sultan et al., 2016).

As far the as the thermal comfort of human and livestock is concerned, it is uneconomical to have two separate systems for cooling (air conditioners) and heating (heat pumps). The heat pumps and air conditioner have same components; therefore one system can be used for both cooling (in summer) and heating (in winter) (Cengel and Boles, 2006). The heating and cooling modes of a heat pump system are shown in self-explanatory Figure 2.3(a) and 2.3(b), respectively (www.kendallcountyair.com). Cengal and Boles, 2006, stated that heat pumps are more economical/competitive for areas of high cooling and less heating loads and vice versa.

2.4 Evaporative Cooling based Air-Conditioning

Evaporative cooling is low cost environmental friendly air-conditioning system (Jha and Chopra, 2006; Zahra and John, 1996), which works on the basis of induced processes of heat and mass transfer (Camargo, 2007). In this system water and air are used as working fluids (Camargo, 2007). The details about advances in evaporative cooling systems are explained in chapter 3. Evaporative cooling systems have been practically studied for extending the shelf/storage life of fruits and vegetables (Dadhich et al., 2008; Jha and Chopra, 2008; Odesola and Onyebuchi, 2009; Zhao et al., 2008). Moreover, it has also been proved successful for on-farm short term storage of agricultural products in hot and dry areas/locations (Jha and Chopra, 2008). Rao, 2015, explained two types of evaporative cooling systems for the storage of agricultural products. The main difference between the Fan & Pad type and Mist/Spray type evaporative cooling systems is the provision of water

supply to the incoming hot air. In Fan and Pad type evaporative cooling system water is supplied to the pad (made of plastic coir/ wood fibers) and the hot ambient air is passed through wetted pad. On the other hand, water is sprinkled on the hot incoming air in case of Mist/Spray type evaporative cooling system. In such systems water evaporates by absorbing heat of vaporization from incoming hot air, therefore the amount of heat removed from the air remains equal to the amount of heat absorbed by evaporated water (ASHRAE, 2007).

Rao, 2015, proposed the commercial type evaporative cooler for storage of agricultural products that works on the principle of Fan and Pad type evaporative cooling system as explained above. In order to achieve the efficient and uniform cooling, the containers of agricultural products in cold store/room must be stacked in a way that supply air can move all around the product's containers at minimum velocity.

The different types of evaporative cooling systems depending on various types of construction materials (bricks, sands, ceramics, woods, metals, fibers, zeolites etc.) and energy sources (solar, electric etc.) have been experimentally studied for the storage of agricultural products in literature (Habibunnisa et al., 1988; Roy, 1984; Roy and Khurdiya, 1982). In this regard, Chouksey, 1985, has studied the design aspects and performance of a 20 tonne capacity solar cum wind aspirator evaporative cooling system for the storage of potatoes and other semi perishable agricultural products. This system maintained the storage/supply air temperature between 21-25°C (80-90% RH) when ambient air temperature was 40-42°C (30-35% RH) with the ventilation rate of 24m³/min. Moreover, the potatoes stored in evaporative cooling storage are more suitable for making potato chips due to less physiological losses, and lower reduction in sugar contents as compare to refrigerated cold storage (Lal Basediya et al., 2013). However, the requirement of continuous water supply, big system size, and limited cooling performance in high humidity conditions are the major drawbacks of the evaporative cooling systems.

2.5 Desiccant based Air-Conditioning

The desiccant air-conditioning (DAC) deals the latent and sensible load of airconditioning distinctly. Such distinct control of sensible and latent load of AC makes it more versatile and flexible for wide range of applications. Desiccant materials (e.g. silica gel, activated alumina, molecular sieve, composite materials, liquid and polymer desiccants, bio-desiccant and activated carbons, bentonite clay etc.) as hygroscopic in nature possess water loving ability (Sultan et al., 2015). The main driving force for moisture adsorption by desiccant material is the difference in vapor pressure between air and desiccant surface.

The water loving ability of desiccants makes them able to be used in applications where the control over relative humidity is crucial. The general applications of the DAC include buildings (Enteria et al., 2009; Enteria et al., 2010), supermarkets, schools, ice arenas, cold warehouse, hotels, theaters, hospitals (Dabrowski, 1998), automobiles (Nagaya et al., 2006), wet markets (Lee and Lee, 2013), marine ships (Guojie et al., 2012; Zhu and Chen, 2014), museums (Ascione et al., 2009; Ascione et al., 2013) etc. However, its agricultural based applications may include greenhouses (Longo and Gasparella, 2015; Sultan et al. 2014, Sultan et al., 2016), products storage and preservation (Mahmood et al., 2015, Mahmood et al., 2016, Sultan et al. 2016), and Livestock cooling etc. In DAC, latent load is achieved through desiccant dehumidification while the sensible load is achieved by sensible heat exchanging through heat exchanger and/or evaporative cooler.

The thermally driven DAC systems can be broadly classified into solid DAC system, liquid DAC system and hybrid DAC system. However, the selection and performance of typical solid DAC system depends on the ambient condition (e.g. temperature, relative humidity, sensible to latent heat ratio), demand conditions (e.g. temperature, relative humidity, flow/ventilation rate) and the type of available regeneration heat source (e.g. electric, gas, solar, waste heat etc.) (Sultan et al., 2015).

The working principle of typical solid DAC system can be explained by Figure 2.4 (Miyazaki, et. al., 2010). The outdoor ambient/process air when pass through desiccant rotor becomes dehumidified and its temperature increases due to the heat of adsorption. The dehumidified air then enters to the heat recovery devices (e.g. heat exchanger) for heat exchanging to the regeneration air. This warm dehumidified process air can be supplied for particular applications. However, in most of cases cooling of the process air is required in order to meet the desired levels of temperature and relative humidity. Therefore, cooling of the process air is accomplished through cooling devices (direct/indirect evaporative cooler) for supplying cool air. Moreover, the regeneration of the desiccant dehumidifier is always

required for its continuous cyclic operation. It can be done by supplying hot regeneration air to the desiccant dehumidifier. The heat energy can be supplied to regeneration air through renewable/non-conventional and conventional energy sources (Enteria and Mizutani, 2011).



Figure 2.4 Schematic of typical desiccant air conditioning system (reproduced from Miyazaki, et. al., 2010).

2.5.1 Single-Stage Desiccant Air-Conditioning

A typical DAC system is mainly consists of desiccant, cooling and regeneration units. In this regard, La et al., 2010, reviewed the different type of single-stage rotary desiccant air-conditioning systems, presented their system configurations and psychrometric representation. These systems include, Pennington/ventilation cycle, Modified ventilation cycle, Recirculation cycle, Dunkle cycle, SENS cycle, REVERS cycle, DINC cycle and Staged regeneration cycle. It is important to mention that the most of the existing DAC cycles are mainly originated from three basic cycle configurations: Ventilation, Recirculation, and Dunkle cycles (La et al. 2010). Therefore, only the basic cycles are briefly discussed here; however the details about all listed cycles can be found from cited reference. Pennington, 1955, patented the first rotary DAC cycle. The working principle of typical Pennington/Ventilation cycle can be explained as: the ambient air when passes through the desiccant wheel (DW) it becomes dehumidified. The temperature of dehumidified air increases due to heat of adsorption. The dehumidified air is then sensibly cooled through heat exchanger. The further cooling is achieved by direct evaporative cooler (DEC) in order to supply the air to conditioned space. On the regeneration side, room return air is passed through the DEC. The DEC cooled and humidified the regeneration air. The regeneration air is then passed through the heat exchanger for sensible precooling of process air and heating itself. The regeneration air is further heated by supplying heat through heat source. This hot regeneration air is used to regenerate the desiccant wheel and finally exhausted to the environment. The recirculation cycle as a variation of ventilation cycle is developed for increasing the cooling capacity. In recirculation cycle the room returns air is used as process air and ambient air is used as regeneration air (La et al. 2010). The rest of system operation is similar to ventilation cycle. The thermal COP of recirculation cycle is not usually more than 0.8 (Waugaman et al., 1993).

The ventilation and recirculation cycles are suitable for different particular applications. The ventilation cycle is suitable for applications/spaces which required more fresh air whereas recirculation cycle enables less supply of fresh air. The Dunkle cycle combines the merits of both the ventilation cycle and regeneration cycle. However, its design configuration requires two heat exchangers (La et al. 2010). The applicability of the Dunkle cycle is also restricted due to supply of less fresh air like recirculation cycle (La et al., 2010).

2.5.2 Multi-Stage Desiccant Air-Conditioning

In multi-stage desiccant air-conditioning system lower regeneration is required as compare to single stage desiccant air-conditioning system (Sultan et al., 2015). The multistage dehumidification promotes the isothermal dehumidification which ultimately results in the requirement of less regeneration temperature. Lower regeneration temperature benefitted to economic system design and operation. It promotes the utilization of low grade waste heat and solar energy etc. (Huan, and Jianlei, 1999). A comparison between single-stage and multi-stage dehumidification is shown in Figure 2.5. The regeneration temperature required for multi-stage process air dehumidification (1-2a-3a-4a-5a) is T_1 and T_2 , whereas it is T_3 in case of single-stage process air dehumidification (1-6a-7a). Therefore, regeneration temperature (T_1, T_2) of multi-stage dehumidification is much lower than regeneration temperature (T_3) of single-stage dehumidification as shown in Figure 2.5.

Two-stage desiccant unit was developed for fast-food restaurants by Gershon Meckler Associates, P.C. (Mei et al., 1992). The thermal COP achieved by this system was 0.89. However, the annual energy cost and electricity use by the system was reported as 40% and 60% less, respectively, when compare with vapor compression air-conditioning.



Dry-bulb temperature [°C]

Figure 2.5 Psychrometric comparison between single-stage & multi-stage desiccant dehumidification and air conditioning.

Two types of two-stage rotary desiccant cooling systems have been experimentally tested by authors (Ge et al., 2008) and (Ge et al., 2009). These systems named as two-stage rotary desiccant cooling system (TSDCS) and one-rotor two-stage rotary desiccant cooling system (OTSDCS). The main difference between theses system was in the divisions of cross section of desiccant wheel. In TSDCS the desiccant wheel was divided into two parts: one for process and other for regeneration air, while in OTSDCS desiccant wheel was divided in four parts: two for process and two for regeneration air (Ge et al., 2008). Silicagel lithium chloride based composite was used as desiccant material. It was concluded in

these studies that both systems can achieve thermal COP greater than 1.0 and can be driven by heat sources of above 50°C. Moreover, design configuration of one-rotor two-stage rotary cooling desiccant reduces the system size about half of two-stage rotary desiccant cooling (Ge et al., 2008; Ge et al., 2009; La et al., 2010).

2.5.3 Solar Energy Operated Desiccant Air-Conditioning

A solar powered desiccant air-conditioning system is installed in the building of Chamber of Trade and Commerce, Freiburg/Germany for the air-conditioning of seminar room and cafeteria (Henning, 2007). System consists of desiccant/sorption wheel (silicagel), heat recovery wheel (heat exchanger, solar collector and humidifiers etc.) as shown in Figure 2.6 (Henning, 2007). Solar air collectors of 100 m² were used for the regeneration of desiccant wheel. Desiccant wheel handles the air flow rate of 10,200 m³/h. The installed system maintains the indoor conditions properly, though its COP was less than expected design value (Henning, 2007).

Enteria et al., 2009, developed a solar thermal and electric energy operated desiccant air-conditioning system at Tohoku University, Japan. The experimental system consists of two subsystems: thermal energy subsystem and desiccant cooling subsystem as shown in Figure 2.7 (Enteria et al., 2009). The objective of thermal energy subsystem is to collect the thermal energy for desiccant regeneration either from solar radiation or electric heater. The thermal energy subsystem was consists of five collector panels, an electric heater for night-time thermal energy storage and for lower solar radiation period, and a thermal storage tank with water as working fluid. In order to avoid the freezing of water in solar collector during winter operation, an automatic compressed air mechanism is attached to drain the water as soon the solar collector surface temperature approaches to water freezing point. The desiccant cooling subsystem was mainly consists of a silica-gel desiccant wheel, two cross-flow heat exchangers, and an evaporative cooler. The system design allows the thermal energy storage during night-time in order to use it for early daytime operation. The reason to store the energy at nighttime is to make the system cost effective by using relatively lower price of electricity. The system was experimentally tested for standard: outdoor air conditions (T: 30°C, RH: 60%) and indoor return air conditions (T: 26°C, RH: 55%) in Japan. The outdoor and return indoor air flow rates were

maintained as 200 m³/h and 100 m³/h, respectively. The calculated total COP of the system was quite low (0.25) due to utilization of energy by auxiliary devices (i.e. control systems, instrumentations, etc.). However, system can work in both day and night time (Enteria et al., 2009). The study was further extended by the same authors (Enteria et al., 2010) in order to check the improvement in system performance and COP by increasing the regeneration temperature. The desiccant dehumidification performance increases by increasing the regeneration temperature from 60°C to 75°C but COP decreases accordingly from 0.44 to 0.35. Moreover, the supply air temperature also increases by increasing the regeneration temperature. Therefore, it can be concluded that thermal energy supplied to the desiccant air-conditioning system is not proportional the produced cooling. It was found in further analyses of the system that maximum energy and exergy loss occur in solar collectors (Enteria et al., 2013).



Figure 2.6 Schematic of solar powered desiccant air-conditioning system installed at Freiburg, Germany (Henning, 2007).



Figure 2.7 Schematic of solar thermal and electric energy operated desiccant airconditioning system installed at Tohoku University, Japan (Enteria et al., 2009).

2.6 Hybrid System

Hybrid system combines the features of coupled air-conditioning/cooling systems and mitigates the problems associated with their air-conditioning/cooling processes (Kojok et al., 2016). The properly selected hybrid systems for particular applications may reduce the energy consumption and improve the system COP.

2.6.1 Evaporative Cooling based Hybrid System

Cengal and Boles, 1998, stated that one degree centigrade (1°C) increase in condenser temperature of conventional vapor compression air-conditioning (VAC) system results in 2-4% decrease in system performance. In this regard, evaporative coolers particularly in hot and dry climatic conditions can be used to pre-cool the ambient air before it passes over the condenser coil of VAC system.

Delfani et al., 2010, constructed a hybrid system consists of indirect evaporative cooler (IEC), conventional vapor compression air conditioning system and other necessary instrumentation. The purpose of IEC was to supply the pre-cool air to the VAC unit. The

system performance was experimentally investigated for different climatic conditions of Iran. It was concluded that the coupling of IEC with VAC unit results in saving of 55% electrical energy consumption.

Hajidavalloo and Eghtedari, 2010, conducted an experimental study by connecting the direct evaporative cooler (DEC) to the air-cooled condenser of existing split air-conditioner. The performance of retrofitted DEC condenser and air-cooled condenser of existing air-conditioner are compared. It was concluded that system performance can be improved about 50%, and power consumption can be reduced about 20%, when employing retrofitted DEC condenser as compared to air-cooled condenser.

2.6.2 Desiccant based Hybrid System

The concept of hybrid desiccant air-conditioning using DAC system with assistance of VAC got so much attention because of its high performance, feasibility, and reliability as compared to standalone DAC system. Usually when the standalone DAC system could not achieve the demand conditions fully, it is suggested to switch on hybrid system. The hybrid DAC system can work effectively in almost all climatic conditions without any interruption (day and night) for different applications.

Burns et al., 1980, studied the three different configurations of hybrid desiccant airconditioning system for high latent load applications (i.e. supermarkets). These are one of the earliest major studies done for solid desiccant based hybrid air-conditioning system. In these studies comparison was made between hybrid DAC and VAC systems. The hybrid DAC systems, for specified design conditions (ambient and indoor air conditions), resulted in total air-conditioning energy savings of about 57-66% (Burns et al., 1980).

Baniyounes et al., 2013, installed a single-stage solar hybrid desiccant airconditioning system at Central Queensland University, Australia. The system mainly consists of three units: solar thermal unit, DAC and VAC units as shown in Figure 2.8 (Baniyounes et al., 2013). The subcomponents of the solar thermal unit incudes flat plate solar collectors (10 m²), hot water buffer tank, two backup heaters (4.8 kW and 9 kW) and valves etc. However, the DAC and VAC units had typical subcomponents. The system performance was measured both by experimentation and simulation. The experimentally measured annual energy savings achieved by hybrid DAC system was 18% as compared to standalone conventional VAC system. However, its simulated value was found as 23%. The COP of the system was measured as high as 0.83 (simulated: 0.728) in the month of December. The detail month wise trends of COP and energy savings can be seen from cited reference. In another study, El-Agouz and Kabeel, 2014, simulated that the COP of single-stage hybrid DAC system can be achieved maximum up to 1.03.

La et al., 2011, configured and experimentally investigated the two-stage solar hybrid DAC system in Jiangsu, China. The performance of the system was theoretically analyzed. The detail system components and schematics can be found from cited reference. It is determined that system can achieve the designed cooling load (i.e. about 10 kW) under particular local conditions with thermal and electric COP of 1.0 and 10, respectively. Moreover, the system is feasible for its successful operation in wide range of climatic conditions like extremely humid (e.g. Hong Kong), humid (e.g. Shanghai), and temperate (Beijing) climatic conditions



Figure 2.8 Schematics of single-stage solar hybrid desiccant air-conditioning system installed at Central Queensland University, Australia (Baniyounes et al., 2013).

2.7 Nomenclature

AC	air-conditioning
CAS	controlled atmosphere storage
COP	coefficient of performance
DAC	desiccant air-conditioning
DEC	direct evaporative cooler
DW	desiccant wheel
IEC	indirect evaporative cooler
MAP	modified atmosphere packaging
UV-C	Ultraviolet-C
VAC	vapor compression air-conditioning

2.8 References

- Adhikari, A., Syamaladevi, R.M., Killinger, K., Sablani, S.S., 2015. Ultraviolet-C light inactivation of Escherichia coli O157: H7 and Listeria monocytogenes on organic fruit surfaces. Int. J. Food Microbiol. 210, 136-142.
- Ascione, F., Bellia, L., Capozzoli, A., 2013. A coupled numerical approach on museum air conditioning: Energy and fluid-dynamic analysis. Appl. Energy. 103, 416-427.
- Ascione, F., Bellia, L., Capozzoli, A., Minichiello, F., 2009. Energy saving strategies in airconditioning for museums. Appl. Therm. Eng. 29, 676-686.
- ASHRAE, 2007. Handbook-HVAC applications. American Society of Heating, Refrigerating and Air-conditioning Engineers, Inc., Atlanta.
- Baniyounes, A.M., Rasul, M. G., Khan, M.M.K., 2013. Experimental assessment of a solar desiccant cooling system for an institutional building in subtropical Queensland, Australia.Energy Build. 62, 78-86.
- Barkai-Golan, R., 1990. Postharvest disease suppression by atmospheric modifications. In: Calderon, M., Barkai-Golan, R. (Eds.), Food Preservation by Modified Atmospheres. CRC Press, Boca Raton, pp. 237–264.
- Barkai-Golan, R., 2001. Postharvest Diseases of Fruits and Vegetables: Development and Control. Elsevier, pp (418)

- Ben-Yehoshua, S., Porat, R., 2005. Heat treatments to reduce decay. In: Ben-Yehoshua, S. (Eds.), Environmentally Friendly Technologies for Agricultural Produce Quality. CRC Press, Boca Raton
- Bubel, M., Bubel, N., 1979. Root cellaring: the simple. no-processing way to store fruits and vegetables. Emmaus, PA: Rodale Press. pp. 297.
- Burns, P.R., Mitchell, J.W., Beckman, W.A., 1985. Hybrid desiccant cooling systems in supermarket applications. ASHRAE Trans. 91(part 1B), 457-468.
- Camargo J.R., 2007. Evaporative cooling: water for thermal comfort. An interdisciplinary J. Applied Sci. 3, 51–61.
- Camelo, A.F.L., 2004. Manual for the preparation and sale of fruits and vegetables: from field to market. FAO agricultural Services Bulletin No. 151. Food and Agriculture Organization of the United Nations, Rome (Italy).
- Casals, C., Viñas, I., Landl, A., Picouet, P., Torres, R., Usall, J., 2010. Application of radio frequency heating to control brown rot on peaches and nectarines. Postharvest Biol. Technol. 58, 218–224.
- Cengel, Y.A., Boles, M.A., 1998. Thermodynamics: an engineering approach, 3rd Ed., McGraw-Hill. New York, USA.
- Cengel, Y.A., Boles, M.A., 2006. Thermodynamics: an engineering approach, 5th Ed., McGraw-Hill. New York, USA.
- Chen, H., Ling, J., Wu, F., Zhang, L., Sun, Z., Yang, H., 2013. Effect of hypobaric storage on flesh lignification, active oxygen metabolism and related enzyme activities in bamboo shoots. LWT-Food Sci. Technol. 51, 190-195.
- Chouksey R.G., 1985. Design of passive ventilated and evaporatively cooled storage structures for potato and other semi perishables. In: Proc. Silver jubilee convention of ISAE held at Bhopal, India, 29–31 October, pp 45–51.
- D'Hallewin, G., Schirra, M., Manueddu, E., Piga, A., Ben-Yehoshua, S., 1999. Scoparone and scopoletin accumulation and ultraviolet-C induced resistance to postharvest decay in oranges as influenced by harvest date. J. Am. Soc. Hortic. Sci. 124, 702– 707.

- Dabrowski, A., 1998. Adsorption and its applications in industry and environmental protection studies in surface science and catalysis. Elsevier Sci. Appl. Ind. 120, 879-916.
- Dadhich S.M., Dadhich, H., Verma, R.C., 2008. Comparative study on storage of fruits and vegetables in evaporative cool chamber and in ambient. Int. J. Food Eng. 4, 1-11.
- Dauthy, M.E., 1995. Fruit and vegetable processing. FAO agricultural Services Bulletin No. 119. Food and Agriculture Organization of the United Nations, Rome (Italy).
- Delfani, S., Esmaeelian, J., Pasdarshahri, H., Karami, M., 2010. Energy saving potential of an indirect evaporative cooler as a pre-cooling unit for mechanical cooling systems in Iran. Energy Build. 42, 2169-2176.
- Droby, S., Chalutz, E., Horev, B., Cohen, L., Gaba, V., Wilson, C.L., Wisniewski, M.E., 1993. Factors affecting UV-induced resistance in grapefruit against the green mould decay caused by penicillium digitatum. Plant Pathol. 42, 418–425.
- Eckert, J.W., Sommer, A.F., 1967. Control of diseases of fruits and vegetables by postharvest treatment. Annu. Rev. Phytopathol. 5, 391-428.
- El-Agouz, S.A., Kabeel, A.E., 2014. Performance of desiccant air conditioning system with geothermal energy under different climatic conditions. Energy Convers. Manag. 88, 464-475.
- El-Sebaii, A.A., Shalaby, S.M., 2012. Solar drying of agricultural products: A review. Renew. Sustain. Energy Rev. 16, 37-43.
- Enteria, N., Mizutani, K., 2011. The role of the thermally activated desiccant cooling technologies in the issue of energy and environment. Renew. Sustain. Energy Rev. 15, 2095-2122.
- Enteria, N., Yoshino, H., Mochida, A., Takaki, R., Satake, A., Yoshie, R., Mitamura, T. Baba, S., 2009. Construction and initial operation of the combined solar thermal and electric desiccant cooling system. Sol. Energy. 83.1300-1311.
- Enteria, N., Yoshino, H., Satake, A., Mochida, A., Takaki, R., Yoshie, R. Baba, S., 2010. Development and construction of the novel solar thermal desiccant cooling system incorporating hot water production. Appl. Energy. 872, 478-486.

- Enteria, N., Yoshino, H., Takaki, R., Yonekura, H., Satake, A. and Mochida, A., 2013. First and second law analyses of the developed solar-desiccant air-conditioning system (SDACS) operation during the summer day. Energy Build. 60, 239-251.
- Ertekin, C., Yaldiz, O., 2004. Drying of eggplant and selection of a suitable thin layer drying model. J. Food Eng. 63, 349-359.
- Esper, A., Mühlbauer, W., 1998. Solar drying-an effective means of food preservation. Renew. Energy. 15, 95-100.
- Fudholi, A., Sopian, K., Ruslan, M.H., Alghoul, M.A., Sulaiman, M.Y. 2010. Review of solar dryers for agricultural and marine products. Renew. Sustain. Energy Rev. 14, 1-30.
- Furlan G., Mancini N.A., Sayigh A.A.M., 1983. Non-conventional energy sources. Miramare-Triest, Italy.
- Ge, T.S., Dai, Y.J., Wang, R.Z., Li,Y., 2008. Experimental investigation on a one-rotor two-stage rotary desiccant cooling system. Energy. 33, 1807-1815.
- Ge, T.S., Li, Y., Wang, R.Z., Dai, Y.J., 2009. Experimental study on a two-stage rotary desiccant cooling system. Int. J. Refrig. 32, 498-508.
- Goyette, B., Charles, M.T., Vigneault, C., Raghavan, V., 2007. Pressure treatment for increasing fruit and vegetable qualities. Stewart Postharvest Rev. 3, 1-6.
- Greig, D. J., Reeves, M., 1985. Prevention of post-harvest food losses: a training manual. Food and Agriculture Organization of the United Nations, Rome (Italy).
- Guojie, Z., Chaoyu, Z., Guanghai, Y., Wu, C., 2012. Development of a new marine rotary desiccant air conditioning system and its energy consumption analysis. Energy Procedia. 16, 1095-1101.
- Habibunnisa, E.A., Aror, E., Narasimham, P., 1988. Extension of storage life of the fungicidal waxol dip treated apples and orange under evaporative cooling storage conditions. J. Food Sci. Technol. 25, 75-77.
- Hajidavalloo, E., Eghtedari, H., 2010. Performance improvement of air-cooled refrigeration system by using evaporatively cooled air condenser. Int. J. Refrig. 33, 982-988.
- Henning, H.M., 2007. Solar assisted air-conditioning of buildings-an overview. Appl. Therm. Eng. 27, 1734-1749.

- Hobson, P., 1981. Build your own underground root cellar. Storey Communications, Inc. pp. 32.
- Hoehn, E., Prange, R.K., Vigneault, C., 2009. Storage technology and applications, in: Yahia, M.E. (Eds), Modified and controlled atmospheres for the storage, transportation, and packaging of horticultural commodities. CRC Press, Taylor & Francis Group, Boca Raton, US, pp. 17-50.
- Hopkins, E.F., Loucks, K.W., 1948. A curing procedure for the reduction of mold decay in citrus fruits. Bull. Univ. Florida Agric. Exp. Station, pp. 450.
- Huan, Z., Jianlei, N., 1999. A two-stage desiccant cooling system using low-temperature heat. Build. Serv. Res. Technol. 20, 51-55.
- Ippolito, A., Lattanzio, V., Nigro, F., Di Venere, D., Lima, G., Castellano, M.A., Salerno, M., 1994. Improvement of kiwifruit resistance to Botrytis storage rot by curing. Phytopathol. Mediterr. 33, 132-136.
- Jacobi, K.K., MacRae, E.A., Hetherington, S.E., 2001. Postharvest heat disinfestation treatments of mango fruit. Sci. Hortic. 89, 171–193.
- Jha S.N., Chopra, S., 2006. Selection of bricks and cooling pad for construction of evaporatively cooled storage structure. Inst Engineers (I) (AG). 87, 25-28.
- Ji, L., Pang, J., Li, S., 2012. Application of new physical storage technology in fruit and vegetable industry. African J. Biotech. 11, 6718-6722.
- Karabulut, O.A., Smilanick, J.L., Crisosto, C.H., Palou, L., 2010. Control of brown rot of stone fruit by brief heated water immersion treatments. Crop Prot. 29, 903–906.
- Khurmi, R.S., Gupta, J.K., 2008. A textbook of thermal engineering. S. Chand Co. Ltd. India.
- Kitinoja, L., Kader, A.A., 2002. Small-scale postharvest handling practices: a manual for horticultural crops, fourth ed. Postharvest Technology Research and Information Center, University of California, Davis.
- Kojok, F., Fardoun, F., Younes, R., Outbib, R., 2016. Hybrid cooling systems: A review and an optimized selection scheme. Renew. Sustain. Energy Rev. 65, 57-80.
- Kumar, M., Sansaniwal, S.K., Khatak, P. 2016. Progress in solar dryers for drying various commodities. Renew. Sustain. Energy Rev. 55, 346-360.

- La, D., Dai, Y., Li, Y., Ge, T., Wang, R., 2011. Case study and theoretical analysis of a solar driven two-stage rotary desiccant cooling system assisted by vapor compression air-conditioning. Sol. Energy. 85, 2997-3009.
- La, D., Dai, Y.J., Li, Y., Wang, R.Z., Ge, T.S., 2010. Technical development of rotary desiccant dehumidification and air-conditioning: a review. Renew. Sustain. Energy Rev. 14, 130-147.
- Lal Basediya, A., Samuel, D.V.K., Beera, V., 2013. Evaporative cooling system for storage of fruits and vegetables-a review. J. Food Sci. Technol. 50, 429-442.
- Lee, S.H., Lee, W.L., 2013. Site verification and modeling of desiccant-based system as an alternative to conventional air-conditioning systems for wet markets. Energy. 55, 1076-1083.
- Liu, C., Cai, L., Lu, X., Han, X., Ying, T., 2012. Effect of postharvest UV-C irradiation on phenolic compound content and antioxidant activity of tomato fruit during storage.J. Integr. Agric. 11, 159–165.
- Longo, G.A., Gasparella, A., 2015. Three years experimental comparative analysis of a desiccant based air conditioning system for a flower greenhouse: Assessment of different desiccants. Appl. Therm. Eng. 78, 584-590.
- Lu, J.Y., Stevens, C., Khan, V.A., Kabwe, M., 1991. The effect of ultraviolet irradiation on shelf-life and ripening of peaches and apples. J. Food Qual. 14, 299–305
- Lurie, S., 2008. Heat treatment for enhancing postharvest quality, in: G. Paliyath, D.P. Murr, A.K. Handa, S. Lurie (Eds.), Postharvest Biology and Technology of Fruits, Vegetables, and Flowers, Wiley-Blackwell, pp. 246-258.
- Lurie, S., Pedreschi, R., 2014. Fundamental aspects of postharvest heat treatments. Horticulture research. 1. 1-7. doi:10.1038/hortres.2014.30
- Mahmood, M.H., Sultan, M., Miyazaki, T., Koyama, S., 2016a. Desiccant Air-Conditioning System for Storage of Fruits and Vegetables: Pakistan Preview. Evergr. Jt. J. Nov. Carbon Resour. Sci. Green Asia Strategy. 3, 12-17.
- Mahmood, M.H., Sultan, M., Miyazaki, T., Koyama, S., 2016b. Theoretical and experimental analysis of desiccant air conditioning system for storage of agricultural products. 16th International refrigeration and air conditioning conference at Purdue, July 11-14, 2016. Purdue University, USA.

- Mehta, A., Kaul, H.N., 1997. Physiological weight loss in potatoes under non refrigerated storage: contribution of respiration and transpiration. J. Indian Potato Assoc. 24, 106-113.
- Mei, V.C., Chen, F.C., Lavan, Z., Collier Jr, R.K., Meckler, G., 1992. An assessment of desiccant cooling and dehumidification technology (No. ORNL/CON-309). Oak Ridge National Lab., TN, United States.
- Mercier, J., Arul, J., Julien, C., 1993. Effect of UV-C on phytoalexin accumulation and resistance to botrytis cinerea in stored carrots. J. Phytopathol. 139, 17–25.
- Miyazaki, T., Oda, T., Ito, M., Kawasaki, N., Nikai, I., 2010. The possibility of the energy cost savings by the electricity driven desiccant system with a high performance evaporative cooler. Int. Symp. Innov. Mater Process Energy Syst.
- Nagaya, K., Senbongi, T., Li, Y., Zheng, J., Murakami, I., 2006. High energy efficiency desiccant assisted automobile air-conditioner and its temperature and humidity control system. Appl. Therm. Eng. 26, 1545-1551.
- Nigro, F., Ippolito, A., Lattanzio, V., Di Venere, D., Salerno, M., 2000. Effect of ultraviolet-c light on postharvest decay of strawberry. J. Plant Pathol. 82, 29–37.
- Nilsson, R., Nyhlen, A., Nilsson, R., Stoy, V., & Nilson, T. 1956. Carbon dioxide storage of apples and pears 1951–1954. J. R. Swed. Acad. Agric. 93, 318-347.
- Odesola, I.F., Onyebuchi, O, 2009. A review of porous evaporative cooling for the preservation of fruits and vegetables. Pacific J. Sci. Technol. 10, 935-941.
- Pennington, N.A., 1955. Humidity changer for air-conditioning. U.S. Patent 2,700,537.
- Prasad, J., Vijay, V.K., Tiwari, G.N., Sorayan, V.P.S., 2006. Study on performance evaluation of hybrid drier for turmeric (Curcuma longa L.) drying at village scale. J. Food Eng. 75, 497-502.
- Proctor, D.L., 1994. Grain storage techniques: Evolution and trends in developing countries. FAO agricultural Services Bulletin No. 119. Food and Agriculture Organization of the United Nations, Rome (Italy).
- Rao, C.G., 2015. Engineering for storage of fruits and vegetables: cold storage, controlled atmosphere storage, modified atmosphere storage. Academic Press (Elsevier), USA.

- Romanazzi, G., Gabler, F.M., Smilanick, J.L., 2006. Preharvest chitosan and postharvest UV irradiation treatments suppress gray mold of table grapes. Plant Dis. 90, 445-450.
- Romanazzi, G., Nigro, F., Ippolito, A., 2008. Effectiveness of short hyperbaric treatment to control postharvest decay of sweet cherries and table grapes. Postharvest Biol. Technol. 49, 440–442.
- Romanazzi, G., Nigro, F., Ippolito, A., Salerno, M., 2001. Effect of short hypobaric treatments on postharvest rots of sweet cherries strawberries, and table grapes. Postharvest Biol. Technol. 22, 1-6.
- Roy, K.S., Khurdiya, D.S., 1982. Keep vegetables fresh in summer. Indian Hort. 27, 5-6.
- Roy, S.K., 1984. Postharvest storage of fruits and vegetables in a specially designed built in space. In: Proc. Intl. Workshop on Energy conservation in buildings. CBRI, Roorkee, UP, India, 2–7 April, pp 190–193.
- Sams, C.E., Conway, W.S., 1987. Additive effects of controlled-atmosphere storage and calcium chloride on decay firmness retention, and ethylene production in apples. Plant Dis. 71, 1003–1005.
- Sisquella, M., Picouet, P., Viñas, I., Teixidó, N., Segarra, J., Usall, J., 2014. Improvement of microwave treatment with immersion of fruit in water to control brown rot in stone fruit. Innov. Food Sci. Emerg. Tech. 26, 168-175.
- Sivakumar, D., Fallik, E., 2013. Influence of heat treatments on quality retention of fresh and fresh-cut produce. Food Rev. Int. 29, 294–320.
- Sommer, N.F., 1985. Role of controlled environments in suppression of postharvest diseases. Can. J. Plant Pathol. 7, 331–339.
- Stevens, C., Khan, V.A., Tang, A.Y., Lu, J.Y., 1990. The effect of ultraviolet radiation on mold rots and nutrients of stored sweet potatoes. J. Food Prot. 53, 223–226.
- Sultan, M., Miyazaki, T., Koyama, S., Saha, B.B., 2014. Utilization of desiccant airconditioning system for improvement in greenhouse productivity: a neglected area of research in Pakistan. Int. J. Environ. 4, 1-10.
- Sultan, M., El-Sharkawy, I.I., Miyazaki, T., Saha, B.B., Koyama, S., 2015. An overview of solid desiccant dehumidification and air-conditioning systems. Renew. Sustain. Energy Rev. 46, 16-29.

- Sultan, M., Miyazaki, T., Saha, B.B., Koyama, S., 2016. Steady-state investigation of water vapor adsorption for thermally driven adsorption based greenhouse air-conditioning system. Renew. Energy. 86, 785-795.
- Thompson, A.K., 2015. Fruit and vegetable storage: hypobaric, hyperbaric and controlled atmosphere. Springer Briefs in Food, Health, and Nutrition. Springer, London, pp. 37-114.
- Tonutti, P., 2015. The technical evolution of CA storage protocols and the advancements in elucidating the fruit responses to low oxygen stress. Acta Hortic. 1079, 53–60.
- Usall, J., Ippolito, A., Sisquella, M., Neri, F., 2016. Physical treatments to control postharvest diseases of fresh fruits and vegetables. Postharvest Biol. Tech. 122, 30-40.
- Washington State University, 2015. UV light can kill foodborne pathogens on certain fruits: Research increases options for organic fruit growers. Retrieved from www.sciencedaily.com/releases/2015/07/150728091741.htm
- Waugaman, D.G., Kini, A., Kettleborough, C.F., 1993. A review of desiccant cooling systems. J. Energy Resour. Technol. 115, 1-8.
- www.kendallcountyair.com/learn/heat-pumps/
- www.trtapakistan.org/archives/4680
- Yang, Z., Cao, S., Su, X., Jiang, Y., 2014. Respiratory activity and mitochondrial membrane associated with fruit senescence in postharvest peaches in response to UV-C treatment. Food Chem. 161, 16–21.
- Zahra G, John A.B., 1996. A passive evaporative cooling system by natural ventilation. Build. Environ. 31, 503-507
- Zhao, X., Shuli, L., Riffat, S.B., 2008. Comparative study of heat and mass exchanging materials for indirect evaporative cooling systems. Build. Environ. 43, 1902-1911.
- Zhu, J., Chen, W., 2014. Energy and exergy performance analysis of a marine rotary desiccant air-conditioning system based on orthogonal experiment. Energy. 77, 953-962.

CHAPTER 3

Advances in Evaporative Cooling Systems
CHAPTER 3

Advances in Evaporative Cooling Systems

This chapter briefly explains the principle and features of advanced dew-point Maisotsenko Cycle evaporative cooling (MEC) in comparison with conventional evaporative cooling. The potential applicability of low cost environmental friendly evaporative cooling technologies like conventional direct and indirect evaporative cooling and M-Cycle evaporative cooling for agricultural products storage is discussed. Various modifications in MEC design are discussed in order to investigate its applicability in humid regions.

3.1 Introduction

The most of the cold storages available today are equipped with conventional vapor compression refrigeration and/or air-conditioning. Such systems are not only degrading the environment and consuming high primary energy but also difficult to provide the optimal storage conditions (temperature and relative humidity) particularly to the tropical agricultural products (Mahmood et al., 2015; Mahmood et al., 2016a, Ndukwu and Manuwa, 2015; Olosunde et al., 2015). It is because of diversified nature of agricultural products, chilling injury, and complex mechanism of transpiration, respiration and fermentation etc. In this perspective, low cost environmental friendly evaporative cooling technologies like conventional direct and indirect evaporative cooling and M-Cycle evaporative cooling have shown potential to provide storage conditions for particular

agricultural products (Mahmood et al., 2016b). Therefore, the principle and features of conventional direct and indirect evaporative cooling are shortly described. On the other hand, advance M-Cycle evaporative cooling due to its potential of dew point cooling (theoretically) is discussed in detail.

3.2 Wet-Bulb Evaporative Cooling

3.2.1 Direct Evaporative Cooling

The conventional direct evaporative cooling (DEC) works on the principle of isenthalpic cooling. The principle operation of DEC is shown in Figure 3.1(a). It can be seen that when ambient air (1) passes through the wet channel it becomes cool (2). The cooling process in the wet channel is considered as adiabatic. During this process sensible heat of the ambient air is converted in to latent heat. The cooling performance of the DEC is obtained by commonly known wet-bulb effectiveness (ε_{wb}) as given in Eq. (3.1) (Anisimov et al., 2014a; Bruno, 2011).

$$\varepsilon_{wb} = \frac{T_1 - T_2}{T_1 - T_{1,wb}} \tag{3.1}$$

The parameters of Eq. (3.1) are similar as explained on Figure 3.1. The wet-bulb effectiveness achieved by DEC is usually ranges between 0.75-0.95 (Sultan, 2015b). However, it can be ideally equal to 1 (Mahmood et al., 2016b).

3.2.2 Indirect Evaporative Cooling

Conventional indirect evaporative cooling (IEC) works on the principle of sensible cooling that means the humidity ratio of ambient and supply air remained same. The working principle of IEC is shown in Figure 3.1(b). It consists of two channels namely dry-channel and wet-channel. The wet-channel remains wetted with moisture/water. The product/process air flows in the dry-channel, and working air flows oppositely in the wet-channel. In this regards, it can be seen (from Figure 3.1b) that when ambient air (1) is flow through the dry-channel it becomes sensibly cooled (2) at constant humidity ratio by transferring its heat to the working air flowing in the wet-channel. The cooling performance

of the IEC is also obtained by commonly known wet-bulb effectiveness (Anisimov et al., 2014a; Bruno, 2011) as given in Eq. (3.1). The wet-bulb effectiveness achieved by IEC is usually ranges between 0.50-0.65 (Sultan, 2015b). However, it can be ideally equal to 1 (Mahmood et al., 2016b).



Figure 3.1 Principle operation of conventional evaporative cooling techniques showing their cooling limit for: (a) DEC, and (b) IEC.

3.3 Dew-Point Evaporative Cooling

The Maisotsenko Cycle (M-Cycle) is a thermodynamic process which captures energy from the air by utilizing the psychrometric renewable energy available from the latent heat of water evaporating into the air (Anisimov et al., 2014b; Anisimov and Pandelidis, 2014; Bruno, 2011; Cui et al., 2015). It combines thermodynamic processes of heat transfer and evaporative cooling to facilitate product temperature to reach the dewpoint temperature of the ambient air. In other words, it is also an advance indirect evaporative cooling (IEC) by which the air can be cooled to the dew-point temperature rather than wet-bulb temperature (Miyazaki et al., 2011a). However, apart from the cooled air the M-Cycle produces saturated hot air which is required by many applications. Thus, the M-Cycle is a heat recovery process (Alsharif et al., 2011; Buyadgie et al., 2015; Maisotsenko et al., 2004; Maisotsenko et al., 2005b; Maisotsenko et al., 2006) by which the system efficiency can be increased tremendously for various applications (see heading 3.4 and appendices).

The basic principle and features of the M-Cycle can be explained from Figure 3.2(a) and 3.2(b) representing the old and modified M-Cycle, respectively. The psychrometric representation of old and modified M-Cycle is shown in Figure 3.2(c). It consists of two kinds of primary channels named as wet and dry channels. The product as well as working channels are devoted for air flow in case of old M-Cycle (Figure 3.2(a)), whereas modified M-Cycle (Figure 3.2(b)) gives the freedom to recover the heat from any fluid/gas by using an additional dry-channel. For cooling and AC applications, the product/process air flows into the dry-channel whereas working air flows into the wet-channel. For example, ambient air (1) is flowed into the dry-channel where it is sensibly cooled at constant humidity to cycle point (2) by transferring the heat to the wet-channel. The operational principle of M-Cycle is based on diverting the cooled air (2) to the wet-channel in order to use as working air. It results in subsequently decrement of effective dry-bulb $(1 \rightarrow 2a; 2b; 2c; 2)$ and wetbulb (1w \rightarrow 2a,w; 2b,w; 2c,w; 2dp) temperatures of the working air in the wet-channel as shown in Figure 3.2(d). Sequential decrement of dry-bulb temperature in the wet-channel brings the effective wet-bulb temperature to be ideally equal to the dew-point temperature. Hence for an ideal heat transfer surface, the product air can be sensible cooled to the dewpoint temperature of the ambient air. Moreover, saturated hot air (3) is rejected from the wet-channel equivalent to the evaporated water and recovered heat. Depending upon the nature of M-Cycle application, the product and working channels can be interchanged in order to utilize the saturated hot air (see appendix D).

Using the air inlet and outlet conditions, the M-Cycle performance is usually estimated by dew-point effectiveness. It is the ratio of inlet and outlet dry-bulb temperature difference to the temperature difference between inlet dry-bulb and the corresponding dew-point temperature (Anisimov et al., 2015b; Hasan, 2012; Riangvilaikul and Kumar et al., 2010a). The dew-point effectiveness (ε_{dp}) is given by Eq. (3.2) as follows

$$\varepsilon_{dp} = \frac{T_1 - T_2}{T_1 - T_{1,dp}} \tag{3.2}$$



Figure 3.2 Schematic diagram of Maisotsenko Cycle for: (a) old M-Cycle, (b) modified M-Cycle, (c) psychrometric representation, and (d) sequential temperature decrement in wetchannel.

The parameters of Eq. (3.2) are similar as explained on Figure 3.2. The conventional DEC (i.e. isenthalpic cooling) and IEC (i.e. sensible cooling) processes are presented on Figure 3.1(a) and 3.1(b), respectively in order to compare with M-Cycle versatility. It can be noticed that the theoretical limit of DEC and IEC processes is wet-bulb temperature whereas it is dew-point temperature in case of M-Cycle (Anisimov et al., 2014b; Anisimov et al., 2014a; Cui et al., 2014a) as elaborated in Figure 3.2(c). The wet-bulb effectiveness achieved by the M-Cycle is up to 1.80 (Anisimov et al., 2014a; Anisimov et al., 2014b; Anisimov et al., 2015b; Anisimov and Pandelidis, 2014; Gillan, 2008; Zhan et al., 2011a; Zhan et al., 2011b; Zhao et al., 2008) whereas it can be ideally equal to 1.0 for DEC/IEC techniques. Unlike conventional evaporative cooling, the M-Cycle efficiency increases with the increase in temperature at $X_{inlet} \leq 11.2$ g/kg-DA

(Anisimov et al., 2014a; Khalatov et al., 2011), because of its dependency on dew-point temperature rather than wet-bulb temperature. In other words, at certain absolute humidity the wet-bulb temperature increases with the increase in dry-bulb temperature whereas the dew-point temperature remains constant.

The versatile applicability of the M-Cycle has been studied in the literature for various energy recovery applications. It can be categorized into three main sections based HVAC, cooling (Appendix C) and gas turbine power cycles (Appendix D). The coming headings will discuss M-Cycle HVAC. A brief overview of M-Cycle applications is presented by Figure 3.3 which shows its significance in thermal engineering sciences. Moreover, broad spectrum applications of M-Cycle can be found from Mahmood et al., 2016b.



Figure 3.3 Overview of the Maisotsenko Cycle applications.

3.4 Maisotsenko Cycle based HVAC Systems

The M-Cycle based HVAC technologies are getting much attention by the researchers because of the dew-point evaporative cooling potential. Many HVAC system designs are investigated in the literature in order to achieve the sensible and latent load of AC for various applications which include: conventional AC for residential and office buildings (Caliskan et al., 2011a; Caliskan et al., 2012b; Chua et al., 2013; Cui et al., 2014b; Pandelidis et al., 2015a; Pandelidis and Anisimov, 2015b; Riangvilaikul and Kumar

2010b; Rogdakis et al., 2014; Zhan et al., 2011a); chilled ceiling and/or displacement ventilation (Bruno, 2011; Itani et al., 2015); data center cooling (Weerts, 2011; Weerts, 2012; Weerts et al., 2012; FEMP report, 2012); gas turbine inlet air cooling (Saghafifar and Gadalla, 2015); greenhouse AC (Sultan, 2015b); electronic cooling (Maisotsenko and Reyzin, 2005); automobile batteries cooling (Khazhmuradov et al., 2011); manufacturing and storage processes (Miyazaki et al., 2010b); frost formation for energy recovery (Anisimov et al., 2015a) etc. The additional potential applications are highlighted in Figure 3.3. The coming sections will discuss the M-Cycle HVAC system designs in detail for many applications.

3.4.1 Standalone Maisotsenko Cycle Air-Conditioning

The standalone M-Cycle AC (MAC) (see Figure 3.2) unit provides two types of air flows simultaneously i.e. (1) sensibly cooled air and (2) saturated hot air, which can be used for cooling in summer and humidification cum heating in winter seasons (Anisimov et al., 2014a; Anisimov et al., 2014d; Lee and Lee, 2011; Miyazaki, 2010b, Miyazaki et al., 2011b; Sultan et al., 2015a). The basic working principle of the system is elaborated in section 3.2; however the detail design varies according to the cooling capacity and ambient conditions. The advantages of evaporative cooling over conventional vapor compression AC (VAC) are well known in the literature (Caliskan et al., 2011a; Duan et al., 2012; Jaber and Ajib, 2011; Khalatov et al. 2011). In particular, the standalone MAC possesses huge energy saving potential and consumes 10 times less primary energy as compared to typical VAC system (Anisimov and Pandelidis, 2011a; Khalatov et al., 2011; Maisotsenko and Treyger, 2011; Tertipis et al., 2015; Weerts, 2011). It provides the conditioned air to inhabitant without recirculating the indoor air (Anisimov et al., 2014b; Anisimov and Pandelidis, 2011b; Chua et al., 2013; Reznikov, 2011). On the other hand, typical VAC system recirculates about 85% of the indoor air in order to make the system cost effective (Khalatov et al., 2011). The fresh air intake is very obligatory for some of the AC applications e.g. hospitals/clinics where the patients are sensitive to indoor air quality (Sultan et al., 2015a), and greenhouses when the plants require sufficient amount of CO₂ for effective photosynthesis (Sultan et al., 2014). In this regard, the standalone MAC systems have been successfully studied for different kinds of applications (Alklaibi, 2015; Caliskan et al., 2011a; Caliskan et al., 2012b; Chua et al., 2013; Heidarinejad and Moshari, 2015; Itani et al., 2015; Lee and Lee, 2013; Miyazaki et al., 2011a; Pandelidis et al., 2015a; Pandelidis and Anisimov, 2015b; Weerts, 2011; Weerts, 2012; Weerts et al., 2012; Xuan et al., 2012; Zhan et al., 2011a).

A standalone MAC system has been experimentally investigated in detail for building AC (Anisimov et al., 2014a; Anisimov et al., 2014b; Anisimov et al., 2015b; Anisimov and Pandelidis, 2014; Anisimov and Pandelidis, 2015; Pandelidis et al., 2015a; Pandelidis et al., 2015b; Pandelidis and Anisimov, 2015a; Pandelidis and Anisimov, 2015b). In the cited literature the authors established energy and mass conservation balance equations, and performed mathematical simulation for heat and mass transfer of the standalone M-Cycle unit. The mathematical model was validated against the experimental result which gives the accurate agreement. It is the most accurate approximations from any available M-Cycle models as far as our understanding is concern. Effect of inlet air conditions on the M-Cycle supply air temperature and dew-point effectiveness is shown in Figure 3.4(a) and 3.4(b), respectively. It can be noticed from Figure 3.4(a) that the system delivers highly cooled air at relatively lower humidity ratio and higher temperature of inlet air, because of the corresponding lower dew-point temperature. Figure 3.4(b) gives more comprehensive outlook in order to optimize the ideal inlet air conditions for the system. The dew-point effectiveness of the system increases with the increase in inlet air temperature at $X_{inlet} = 11.2$ g/kg-DA. However, it starts decreasing when the humidity ratio exceeds from 11.2 g/kg-DA. It can be concluded that the standalone MAC is efficient when the ambient air humidity ratio is ≤ 11.2 g/kg-DA.

It is therefore concluded that the system can provide desired thermal comfort in residential and commercial buildings when inlet temperature and humidity ratio are up to 45°C and 11.2 g/kgDA, respectively (Anisimov et al., 2014b; Pandelidis et al., 2015a; Pandelidis and Anisimov, 2015a; Pandelidis and Anisimov, 2015b). On the other hand, the system cooling capacity increases at higher temperatures (Anisimov et al., 2014a) which also motivate its applicability in various industrial applications (Khalatov et al., 2011). Furthermore, the system is practically investigated for National Snow and Ice Data Center (NSIDC) in a project funded by National Science Foundation and NASA (Weerts, 2011; Weerts 2012; Weerts et al., 2012) through the complete retrofit of a conventional AC system, the cooling energy has been reduced up to 70% in summer and 90% in winter by



Figure 3.4 Effect of inlet air conditions on the performance of standalone M-Cycle AC unit for: (a) supply air temperature (reproduced from Anisimov et al., 2014a; Anisimov et al., 2014b; Pandelidis and Anisimov, 2015a; Pandelidis et al., 2015a); and (b) dew-point effectiveness (reproduced from Anisimov et al., 2014a).

means of M-Cycle unit. In another study (Caliskan et al., 2011a; Caliskan, 2011b; Caliskan et al., 2012b) the standalone MAC system is found more sustainable in comparison with three kinds of IEC system because of the high exergy efficiency at dead state temperature \geq 23°C. Dead state is a reference temperature at which thermodynamic system is in equilibrium with the environment. Therefore, it works more efficiently in hot and dry climatic conditions because higher inlet temperature results in more sensible heat transfer between dry and wet channels (Gillan and Maisotsenko, 2008; Maisotsenko and Treyger, 2011).

In addition to above mention scenarios, the standalone MAC can also be used as passive cooling technology for displacement ventilation as well as cooling load reduction. The coming sub-heading discusses the applicability of MAC as passive cooling technology.

3.4.1.1 Chilled Ceiling and Displacement Ventilation

The importance of passive cooling, chilled ceiling and displacement/natural ventilation is well-known in the literature (Ghaddar et al., 2013; Hao et al., 2007; Novoselac and Srebric, 2002; Rees and Haves, 2013; Taki et al., 2011). It helps in cooling load reduction and improves in indoor air quality by possessing high ventilation. The M-Cycle being an advance IEC system has been successfully considered for chilled ceiling displacement ventilation (Itani et al., 2015; Miyazaki et al., 2010a; Miyazaki et al., 2011a). The concept was originally recognized by (Miyazaki et al., 2010a; Miyazaki et al., 2011a) in which the authors performed the system dynamic simulation while using the solar chimney as a system driving force. The schematic diagram of the proposed system is shown in Figure 3.5 (Miyazaki et al., 2010a; Miyazaki et al., 2011a). The study concludes that the system is feasible as a solar energy driven cooling system because the sufficient air flow to the M-Cycle evaporative cooling channel could be induced by the solar chimney. The system can also achieve the radiative load of 40-50 W m⁻² without increasing the ceiling temperature. Furthermore, the system reduces 10% of the maximum cooling load by replacing quarter of the ceiling area with the proposed system. Similarly, another study (Itani et al., 2015) on M-Cycle based displacement ventilation system showed the improvement in sensible load removal of 18% to 72% for the supply air RH ranging from 90% to 10%, respectively.



Figure 3.5 Schematic and geometric representation of solar chimney and M-Cycle in a passive cooling system (reproduced from Miyazaki et al., 2010a; Miyazaki et al., 2011a).

From the above prospective it has been concluded that the standalone MAC can achieve the AC load of an active or passive AC system irrespective of temperature only when the humidity is not so high. On the other hand the system design can be modified by many ways in order to achieve the AC loads in humid regions. The details about such modifications are discussed in the coming sections.

3.4.2 Hybrid Maisotsenko Cycle Air-Conditioning

The hybrid M-Cycle AC (H-MAC) combines the features of vapor compression AC (VAC) and M-Cycle AC (MAC), though the operational scheme could be different depending upon the system design. A simple schematic of the system is shown in Figure 3.6 (Anderson et al., 2011; Duan, 2011; Kozubal and Slayzak, 2010). It can be seen that the MAC unit cools the air sensibly to a certain temperature whereas the VAC unit achieves the remaining sensible and latent load of AC by cooling below the dew-point. Furthermore, the return air from the conditioned space can be passed through the M-Cycle wet-channel followed by the condenser's vicinity in order to improve the overall system performance.



Figure 3.6 Schematic diagram of the hybrid M-Cycle AC (H-MAC) system (Anderson et al., 2011; Duan, 2011; Kozubal and Slayzak, 2010).

The H-MAC system was experimentally investigated by National Renewable Energy Laboratory (NREL) in 2009 (Kozubal and Slayzak, 2010). It has been reported that the system enables energy saving potential of 80% as compared to conventional VAC system (Anderson et al., 2011; Anisimov and Pandelidis, 2011a; Khalatov et al., 2011; Kozubal and Slayzak, 2010; Zube and Gillan, 2011). In another study the H-MAC system has been simulated for hot and humid climates (Cui et al., 2015). The authors proposed a particular H-MAC design and developed the numerical simulation model to investigate the overall system performance. The system cooled the ambient air below the dew-point, and in this regard condensation occurred in the dry-channels due to the lower temperature in wetchannel (working air) as compared to dry-channel dew-point temperature. The lower working air temperature was achieved by mixing the return air from the conditioned space. Results showed that the MAC unit successfully accomplished 40% to 47% of total cooling load depending upon the ambient conditions. Moreover, the system enables higher latent heat transfer rate at various relative humidity when compared with conventional IEC. It has been concluded that the MAC unit as a pre-cooling unit under humid climates will lead to huge energy saving with improved efficiency and reduced vapor compression cooling capacities (Cui et al., 2015). In a similar simulation based study (Dirkes II, 2011) the annual cooling energy saving of 11% to 35% was obtained by H-MAC system when operated in different climatic cities. Furthermore, 25% to 50% less compressor size was obtained when compared with conventional vapor compression system.

3.4.3 Ejector Maisotsenko Cycle Air-Conditioning

The ejector M-Cycle AC (E-MAC) combines the features of ejector AC (EAC) and M-Cycle AC (MAC). In other words, it replaces the compressor of H-MAC with the ejector and the associated assembly. The simplified schematic diagram of the E-MAC system is shown in Figure 3.7 (Buyadgie et al., 2011). The operational mechanism of E-MAC is similar to the H-MAC, however the ejector in the E-MAC system is applied as a jet compressor which is operated by thermal heat most preferably solar energy or low grade waste heat as shown in Figure 3.7. The details of the ejector working principle can be found from the references (Abdulateef et al., 2009; Chen et al., 2013; Chen et al., 2014; Sarkar, 2012). The conventional EAC system enables very low COP as compared to the VAC or other AC systems (Al-Zubaydi, 2011; Chen et al., 2014). Therefore, the combination of EAC with other AC systems (e.g. absorption, VAC, MAC etc.) have been studied with the aim to increase the overall system performance (Aphornratana and Eames, 1998; Buyadgie et al., 2011; Sun, 1997). However, the literature on E-MAC system is limited.

Buyadgie et al., 2011, conducted a study on E-MAC system in order to analyze the system performance for various climatic conditions. The proposed system uses the binary fluid because of the higher COP than the single fluid (Artemenko et al., 2013; Buyadgie et al., 2012a; Buyadgie et al., 2012b). The single and binary fluid based EAC are identical in terms of evaporator, refrigerant condenser, vapor generator, thermal pump, expansion valve and ejector, however the binary fluid based EAC possesses an additional fractionating condenser. The purpose of the fractionating condenser is to separate the mixed fluid (coming from the ejector) into the working and refrigerant fluids. The proposed system is similar to the one shown in Figure 3.7. It is important to mention that the refrigerant and fractionating condensers are presented as single unit on Figure 3.7 in order to avoid the complexity in understating the system operation. The performance of E-MAC and conventional EAC system has been compared at different ambient conditions as shown in Figure 3.8 (Buyadgie et al., 2011). It can be noticed that the E-MAC system possesses



Figure 3.7 Schematic diagram of the ejector M-Cycle AC system (E-MAC) (reproduced from Buyadgie et al., 2011).



Figure 3.8 Performance comparison between EAC and E-MAC systems at different ambient conditions (reproduced from Buyadgie et al., 2011).

higher COP than the conventional EAC system at RH less than 60% when the ambient air temperature is 30°C. On the other hand the E- MAC delivers higher COP throughout the RH range when the ambient air temperature is 43°C. It was concluded that the E-MAC in humid areas should be considered only if the ambient air temperature is more than 40°C. Furthermore, the system yields the minimum energy consumption for processing of unit air flow rate.

3.4.4 Desiccant Maisotsenko Cycle Air-Conditioning

The desiccant M-Cycle AC (D-MAC) system combines the features of desiccant AC (DAC) and M-Cycle AC (MAC). It has been recently studied by many researchers in order to establish an efficient AC system for humid climates (Amer et al., 2015; Gao et al., 2015b; Miyazaki et al., 2010b; Miyazaki et al., 2011b; Saghafifar and Gadalla, 2015c; Sultan et al., 2015a; Worek et al., 2012). The system can be supplement of standalone MAC in order to be feasible for all kinds of climates. It achieves the latent load of AC by desiccant dehumidification whereas sensible load is accomplished by evaporative cooling (Daou et al., 2006; La et al., 2010; Mei and Dai, 2008; Sultan et al., 2015a). In principle, M-Cycle gives better evaporative cooling as compared to conventional DEC/IEC techniques (Anisimov et al., 2015b), as explained in heading 3.3. It also helps to reduce the system cost by eliminating heat exchanger which costs about 45% of the conventional DAC system (Worek et al., 2012). Moreover, the concept of desiccant integrated MAC (as a single unit) protected by the US patent (Maisotsenko et al., 2002) sound more feasible in order to develop a compact system. The concept covers the variety of solid and liquid desiccants however the literature in this field is limited. The coming headings briefly discuss the potential of M-Cycle in solid and liquid based DAC applications.

3.4.4.1 Solid Desiccant System

The conventional solid DAC system has shown the potential for various kinds of AC applications e.g. residential and office buildings (Baniyounes et al., 2013; Enteria et al., 2009; Enteria et al., 2010, La et al., 2011); automobiles (Nagaya et al., 2006); wet markets (Lee and Lee, 2013); drying grains (Ismail et al., 1991); greenhouses (Sultan et al., 2014); marine ships (Guojie et al., 2012; Zhu and Chen, 2014); museums (Ascione et al., 2009; Ascione et al., 2013); hospitals; product storage and preservation etc. The M-Cycle being

an advance IEC can improve the existing DAC systems for different applications covering all kinds of climates. Moreover, the solid D-MAC system has the ability to utilize the exhaust waste heat from the M-Cycle wet-channel for desiccant regeneration, which will increase the system COP. Recent studies have shown its applicability for multiple applications e.g. thermal comfort (Anisimov and Pandelidis, 2011a; Gao et al., 2015a; Miyazaki et al., 2011b; Worek et al., 2012); turbines inlet air cooling (Saghafifar and Gadalla., 2015); greenhouses (Sultan, 2015b); manufacturing and storage processes (Miyazaki et al., 2010a).

Miyazaki et al., 2011b, investigated the potential of solid D-MAC system by means of theoretical and experimental analysis. The simple schematic of the proposed system is shown in Figure 3.9 (Miyazaki et al., 2011b). In addition to M-Cycle unit, the system was consisting of two desiccant beds, a heat exchanger, and a thermal heat unit. A numerical simulation model was developed by the author to simulate the system dynamic performance, and reasonable agreement was obtained between the experimental and simulation results. The dew-point effectiveness (ε_{dp}) by the M-Cycle unit (as given by Eq. 3.2) was found about 0.60-0.75 for supply air flow ratio of \leq 0.75, respectively. The instantaneous variation of final cooling effect of M-Cycle on dehumidified air was determined as shown in Figure 3.10. Results showed that the M-Cycle successfully cooled the dehumidified air from 50-80 °C to the supply air conditions of ~18°C.

According to an experimental study (Enteria et al., 2010), the coefficient of performance (COP) by the standalone DAC system is limited to 0.35-0.44 for regeneration temperature of 60-75°C. However, the theoretical COP by the solid D-MAC system is ranging from 0.60 to 1.10 for regeneration temperature of 70°C to 90°C, respectively (Miyazaki et al., 2011b). Another comparative study (Saghafifar and Gadalla 2015) on four different cooling system showed that the solid D-MAC is the most economically justified inlet cooling technology for a 50 MWe gas turbine power plant in UAE. It possesses life savings of 31.882 MUS\$ and life span of 25 years. Furthermore, a numerical simulation based analysis showed the potential of solid D-MAC for reduction in electricity consumption of various applications (Miyazaki et al., 2010b). A payback period of less than 5 years is obtained when the load hours was more than or equal to 4000.



Figure 3.9 Schematic diagram of solid desiccant M-Cycle AC (D-MAC) system (reproduced from Miyazaki et al., 2011b).



Figure 3.10 Cooling performance of solid D-MAC system (reproduced from Miyazaki et al., 2011b).

3.4.4.2 Liquid Desiccant System

Liquid DAC systems have been widely studied for AC in humid areas (Mohammad et al., 2013a; Mohammad et al., 2013b; Qi et al., 2014). Solar operated liquid DAC systems have shown a huge energy saving potential as compared to conventional AC system (Buker and Riffat, 2015; Mohammad et al., 2013c; Qi and Lu, 2014). However, the system performance is dependent on the sensible AC loads. The MAC can be a good supplement to achieve the sensible AC loads distinctly. Furthermore, the thermal energy from the M-Cycle exhaust air can be used to regenerate the desiccant economically. In this regard, a liquid D-MAC system was experimentally investigated for humid climates (Gao et al., 2015b). The system top view for one channel pair is presented in Figure 3.11 which shows the working principle of M-Cycle in the liquid D-MAC system (Gao et al., 2015b). Effects of operating parameters on system performance are analyzed.



Figure 3.11 Schematic diagram of liquid desiccant M-Cycle AC (D-MAC) system (reproduced from Gao et al., 2015b).



Figure 3.12 Effect of air temperature on the performance of the liquid D-MAC system (reproduced from Gao et al., 2015b).

According to the results, the dew-point effectiveness of about 1.20 is obtained when inlet air humidity ratio was ranging from 9 to 20 g/kg-DA. It is worth mentioning that the dew-point effectiveness exceeds from unity because the dew-point temperature by the dehumidified air is well below than the inlet air. It can be seen from Figure 3.12(a) and (b) that the cooling ability (ΔT) and dew-point effectiveness (ϵ_{dp}) increases with the increase in air inlet temperature. It is because of M-Cycle dependency on dew-point temperature (Khalatov et al., 2011) as explained in section 3.2. Therefore, the liquid D-MAC has a potential to achieve the sensible AC loads (after desiccant dehumidification) for various humid climates. The study concludes that the moisture removal capacity (in dehumidifier) and sensible heat (in M-Cycle unit) can be improved significantly at the same time by increasing the liquid desiccant flow rate or inlet concentration. Furthermore, a new conception of membrane AC based on liquid D-MAC is proposed recently (Burch et al., 2012; Kozubal et al., 2011; Kozubal et al., 2012; Woods and Kozubal, 2012; Woods and Kozubal, 2013a). Significance of membrane AC is well-known because of the isothermal dehumidification (El-Dessouky et al., 2000; Yang et al., 2015), though the efficiency is debatable. In the cited references (Burch et al., 2012; Kozubal et al., 2011; Kozubal et al., 2012; Woods and Kozubal, 2012; Woods and Kozubal, 2013a), the authors developed numerical model of the proposed membrane/desiccant-MAC system on the basis on their previous studies (Woods et al., 2009; Woods et al., 2011; Woods and Kozubal, 2013b), and successfully validated the model with the experimental data within $\pm 10\%$ error. It has been concluded that there is an inherent design tradeoff between COP and system size (Woods and Kozubal, 2012). According to the analysis, the system enables the energy saving potential as compared to conventional AC technologies. The detail energy saving potential will be demonstrated in the future after field installation of the system as reported by the authors (Woods and Kozubal, 2013b).

3.5 Nomenclature

air-conditioning
coefficient of performance [-]
desiccant air-conditioning
direct evaporative cooling
desiccant M-Cycle air-conditioning
ejector air-conditioning
ejector M-Cycle air-conditioning
effectiveness [-]
hybrid M-Cycle air-conditioning
heating, ventilation, and air-conditioning
indirect evaporative cooling
M-Cycle air-conditioning
Maisotsenko Cycle
temperature [°C or K]
vapor compression air-conditioning
humidity ratio [g/kg-DA]

Subscript

dp	dew-point
wb	wet-bulb

3.6 References

- Abdulateef, J.M., Sopian, K., Alghoul M.A, Sulaiman, M.Y., 2009. Review on solar-driven ejector refrigeration technologies. Renew Sustain Energy Rev. 13, 1338-1349. doi:10.1016/j.rser.2008.08.012.
- Alklaibi, A.M., 2015. Experimental and theoretical investigation of internal two-stage evaporative cooler. Energy Convers Manag. 95, 140-148. doi:10.1016/j.enconman.2015.02.035.
- Alsharif, A., Gadalla, M., Dincer, I., 2011. Energy and exergy analyses of Maisotsenko cycle. Proc. ASME 5th Int. Conf. Energy Sustain. ES2011, Washington, DC, USA. 1-7.

- Al-Zubaydi, A.Y.T., 2011. Solar Air-conditioning and Refrigeration with Absorption Chillers Technology in Australia – An Overview on Researches and Applications. J. Adv. Sci. Eng. Res. 1, 23-41.
- Amer, O., Boukhanouf, R., Ibrahim, H.G., 2015. A Review of Evaporative Cooling Technologies. Int. J. Environ. Sci. Dev. 6, 111-117. doi:10.7763/IJESD.2015.V6.571.
- Anderson, E., Antkowiak, M., Butt, R., Davis, J., Dean. J., Hillesheim, M., Hotchkiss, R., Hunsberger, A., Kandt, J., Lund, K., Massey, R., Robichaud, B., Stafford, C., Visser. A., 2011. Broad Overview of Energy Efficiency and Renewable Energy Opportunities for Department of Defense Installations. Colorado, USA: National Renewable Energy Laboratory (NREL).
- Anisimov, S., Pandelidis, D., 2011a. Heat- and mass-transfer processes in indirect evaporative air conditioners through the Maisotsenko cycle. Int. J. Energy Clean Environ. 12 273-286. doi:10.1615/InterJEnerCleanEnv.2012005770.
- Anisimov, S., Pandelidis, D., 2011b. Numerical study of perforated indirect evaporative air cooler. Int. J. Energy Clean Environ. 12, 239-250. doi:10.1615/InterJEnerCleanEnv.2013006668.
- Anisimov, S., Pandelidis, D., 2014. Numerical study of the Maisotsenko cycle heat and mass exchanger. Int. J. Heat Mass Transf. 75, 75–96. doi:10.1016/j.ijheatmasstransfer.2014.03.050.
- Anisimov, S., Pandelidis, D., Jedlikowski, A., Polushkin, V., 2014a. Performance investigation of a M (Maisotsenko)-cycle cross-flow heat exchanger used for indirect evaporative cooling. Energy. 76, 593-606. doi:10.1016/j.energy.2014.08.055.
- Anisimov. S., Pandelidis, D., Danielewicz, J., 2014b. Numerical analysis of selected evaporative exchangers with the Maisotsenko cycle. Energy Convers. Manag. 88, 426-441. doi:10.1016/j.enconman.2014.08.055.
- Anisimov, S., Pandelidis, D., Maisotsenko, V., 2014d. Numerical analysis of heat and mass transfer processes through the Maisotsenko cycle. Proc 10th Int. Conf. Heat Transf. Fluid Mech. Thermodyn. HEFAT-2014, Orlando, Florida. 634–642.
- Anisimov, S., Pandelidis, D., 2015. Theoretical study of the basic cycles for indirect evaporative air cooling. Int. J. Heat Mass Transf. 84, 974-989. doi:10.1016/j.ijheatmasstransfer.2015.01.087.

- Anisimov, S., Jedlikowski, A., Pandelidis, D., 2015a. Frost formation in the cross-flow plate heat exchanger for energy recovery. Int. J. Heat Mass Transf. 90, 201-217. doi:10.1016/j.ijheatmasstransfer.2015.06.056.
- Anisimov, S., Pandelidis, D., Danielewicz, J., 2015b. Numerical study and optimization of the combined indirect evaporative air cooler for air-conditioning systems. Energy. 80, 452-464. doi:10.1016/j.energy.2014.11.086.
- Aphornratana, S., Eames, I.W., 1998. Experimental investigation of a combined ejectorabsorption refrigerator. Int. J. Energy Res. 22, 195–207. doi:10.1002/(SICI)1099-114X(19980310)22:3<195::AID-ER346>3.0.CO;2-A.
- Artemenko, S.V., Buyadgie, D.I., Buyadgie, O.D., Drakhnia, O.Y., Sladkovsky, E.N., Chamchine, A.V., 2013. Analysis of refrigerants properties for the ejector refrigeration systems. Proc 4th IIR Conf. Thermophys. Prop. Transf. Process. Refrig., Delft, The Netherlands. 1-10.
- Ascione, F., Bellia, L., Capozzoli, A., Minichiello, F., 2009. Energy saving strategies in air-conditioning for museums. Appl. Therm. Eng. 29, 676-686. doi:10.1016/j.applthermaleng.2008.03.040.
- Ascione, F., Bellia, L., Capozzoli, A., 2013. A coupled numerical approach on museum airconditioning: Energy and fluid-dynamic analysis. Appl. Energy. 103, 416-427. doi:10.1016/j.apenergy.2012.10.007.
- Baniyounes, A.M., Rasul, M.G., Khan, M.M.K., 201. Experimental assessment of a solar desiccant cooling system for an institutional building in subtropical Queensland, Australia. Energy. Build. 62, 78-86. doi:10.1016/j.enbuild.2013.02.062.
- Bruno, F., 2011. On-site experimental testing of a novel dew point evaporative cooler. Energy Build. 43, 3475-3483. doi:10.1016/j.enbuild.2011.09.013.
- Buker, M.S., Riffat, S.B., 2015. Recent developments in solar assisted liquid desiccant evaporative cooling technology—A review. Energy Build. 96, 95–108. doi:10.1016/j.enbuild.2015.03.020.
- Burch, J., Woods, J., Kozubal, E., Boranian, A., 2012. Zero Energy Communities with Central Solar Plants using Liquid Desiccants and Local Storage. Energy Procedia. 30, 55-64. doi:10.1016/j.egypro.2012.11.008.

- Buyadgie, D., Buyadgie, O., Drakhnia, O., Sladkovskyi, Y., Artemenko, S., Chamchine, A.,
 2011. Theoretical study of the combined M-Cycle/Ejector air-conditioning system. Int.
 J. Energy Clean Environ. 12, 309-318. doi:10.1615/InterJEnerCleanEnv.2013005893.
- Buyadgie, D., Buyadgie, O., Artemenko, S., Chamchine, A., Drakhnia, O., 2012a. Conceptual design of binary/multicomponent fluid ejector refrigeration systems. Int. J. Low-Carbon Technol. 7, 120-127. doi:10.1093/ijlct/cts038.
- Buyadgie, D., Buyadgie, O., Drakhnia, O., Artemenko, S., Chamchine, A., 2012b. Solar Cooling Technologies Using Ejector Refrigeration System. Energy Procedia. 30, 912-920. doi:10.1016/j.egypro.2012.11.103.
- Buyadgie, D., Buyadgie, O., Drakhnia, O., Brodetsky, P., Maisotsenko, V., 2015. Solar low-pressure turbo-ejector Maisotsenko cycle-based power system for electricity, heating, cooling and distillation. Int. J. Low-Carbon Technol. 10, 157-164. doi:10.1093/ijlct/ctv012.
- Caliskan, H., Dincer, I., Hepbasli, A., 2011a. Exergetic and sustainability performance comparison of novel and conventional air cooling systems for building applications. Energy Build. 43, 1461-1472. doi:10.1016/j.enbuild.2011.02.006.
- Caliskan, H., Hepbasli, A., Dincer, I., Maisotsenko, V., 2011b. Thermodynamic performance assessment of a novel air cooling cycle: Maisotsenko cycle. Int. J. Refrig. 34, 980-990. doi:10.1016/j.ijrefrig.2011.02.001.
- Caliskan, H., Dincer, I., Hepbasli, A., 2012a. A comparative study on energetic, exergetic and environmental performance assessments of novel M-Cycle based air coolers for buildings. Energy Convers. Manag. 56, 69-79. doi:10.1016/j.enconman.2011.11.007.
- Caliskan, H., Dincer, I., Hepbasli, A., 2012b. Exergoeconomic, enviroeconomic and sustainability analyses of a novel air cooler. Energy Build. 55, 747-756. doi:10.1016/j.enbuild.2012.03.024.
- Chen, X., Omer, S., Worall, M., Riffat, S., 2013. Recent developments in ejector refrigeration technologies. Renew. Sustain. Energy Rev. 19, 629-651. doi:10.1016/j.rser.2012.11.028.
- Chen, J., Havtun, H., Palm, B., 2014. Screening of working fluids for the ejector refrigeration system. Int. J. Refrig. 47, 1-14. doi:10.1016/j.ijrefrig.2014.07.016.

- Chua, K.J., Chou, S.K., Yang, W.M., Yan, J., 2013. Achieving better energy-efficient airconditioning – A review of technologies and strategies. Appl. Energy. 104, 87-104. doi:10.1016/j.apenergy.2012.10.037.
- Cui, X., Chua, K.J., Yang, W.M., 2014a. Numerical simulation of a novel energy-efficient dew-point evaporative air cooler. Appl. Energy. 136, 979-988. doi:10.1016/j.apenergy.2014.04.040.
- Cui, X., Chua, K.J., Yang, W.M., 2014b. Use of Indirect Evaporative Cooling as Precooling Unit in Humid Tropical Climate: An Energy Saving Technique. Energy Procedia. 61, 176-179. doi:10.1016/j.egypro.2014.11.933.
- Cui, X., Chua, K.J., Islam, M.R., Ng, K.C., 2015. Performance evaluation of an indirect pre-cooling evaporative heat exchanger operating in hot and humid climate. Energy Convers. Manag. 102, 140-150. doi:10.1016/j.enconman.2015.02.025.
- Daou, K., Wang, R.Z., Xia, Z.Z., 2006. Desiccant cooling air-conditioning: a review. Renew. Sustain. Energy. Rev. 10, 55-77. doi:10.1016/j.rser.2004.09.010.
- Dirkes II, J.V., 2011. Energy simulation results for indirect evaporative-assisted DX cooling systems. Int. J. Energy Clean Environ. 12, 209-220. doi:10.1615/InterJEnerCleanEnv.2012005806.
- Duan, Z., 2011. Investigation of a novel dew point indirect evaporative air-conditioning system for buildings. PhD thesis. University of Nottingham.
- Duan, Z., Zhan, C., Zhang, X., Mustafa, M., Zhao, X., Alimohammadisagvand B, Hasan, A., 2012. Indirect evaporative cooling: Past, present and future potentials. Renew. Sustain. Energy. Rev. 16, 6823-6850. doi:10.1016/j.rser.2012.07.007.
- El-Dessouky, H.T., Ettouney, H.M., Bouhamra, W., 2000. A novel air-conditioning system: Membrane air drying and evaporative cooling. Chem. Eng. Res. Des. 78, 999-1009. doi:10.1205/026387600528111.
- Enteria, N., Yoshino, H., Mochida, A., Takaki, R., Satake, A., Yoshie, R., Mitamura, T., Bab, S., 2009. Construction and initial operation of the combined solar thermal and electric desiccant cooling system. Sol. Energy. 83, 1300-1311. doi:10.1016/j.solener.2009.03.008.
- Enteria, N., Yoshino, H., Satake, A., Mochida A, Takaki R, Yoshie R, Baba, S., 2010. Development and construction of the novel solar thermal desiccant cooling system

incorporating hot water production. Appl. Energy. 87, 478-486. doi:10.1016/j.apenergy.2009.08.026.

- FEMP report, 2012. NSIDC data center: energy reduction strategies. United States: U. S. Department of Energy, Energy Efficiency & Renewable Energy, Federal Energy Management Program (FEMP), Report no. DOE/GO-102012-3509. http://www.osti.gov/scitech/biblio/1041349.
- Gao, W., Worek, W., Konduru, V., Adensin, K., 2015a. Numerical study on performance of a desiccant cooling system with indirect evaporative cooler. Energy Build. 86, 16-24. doi:10.1016/j.enbuild.2014.09.049.
- Gao, W.Z., Cheng, Y.P., Jiang, A.G., Liu, T., Anderson, K., 2015b. Experimental investigation on integrated liquid desiccant – Indirect evaporative air cooling system utilizing the Maisotesenko – Cycle. Appl. Therm. Eng. 88, 288-296. doi:10.1016/j.applthermaleng.2014.08.066.
- Ghaddar, N., Ghali, K., Chakroun, W., 2013. Evaporative cooler improves transient thermal comfort in chilled ceiling displacement ventilation conditioned space. Energy Build. 61, 51-60. doi:10.1016/j.enbuild.2013.02.010.
- Gillan, L., 2008. Maisotsenko Cycle for Cooling Processes. Int. J. Energy Clean Environ. 9, 47-64. doi:10.1615/InterJEnerCleanEnv.v9.i1-3.50
- Guojie, Z., Chaoyu, Z., Guanghai, Y., Wu, C., 2012. Development of a new marine rotary desiccant air-conditioning system and its energy consumption analysis. Energy Procedia. 16, Part B:1095–101. doi:10.1016/j.egypro.2012.01.175.
- Hao, X., Zhang, G., Chen, Y., Zou, S., Moschandreas, D.J., 2007. A combined system of chilled ceiling, displacement ventilation and desiccant dehumidification. Build. Environ. 42, 3298-3308. doi:10.1016/j.buildenv.2006.08.020.
- Hasan, A., 2012. Going below the wet-bulb temperature by indirect evaporative cooling: Analysis using a modified ε-NTU method. Appl. Energy. 89, 237-245. doi:10.1016/j.apenergy.2011.07.005.
- He, S., Gurgenci, H., Guan, Z., Huang, X., Lucas, M., 2015. A review of wetted media with potential application in the pre-cooling of natural draft dry cooling towers. Renew. Sustain. Energy. Rev. 44, 407-422. doi:10.1016/j.rser.2014.12.037.

- Heidarinejad, G., Moshari, S., 2015. Novel modeling of an indirect evaporative cooling system with cross-flow configuration. Energy Build. 92, 351-362. doi:10.1016/j.enbuild.2015.01.034.
- Ismail, M.Z., Angus, D.E., Thorpe, G.R., 1991. The performance of a solar-regenerated open-cycle desiccant bed grain cooling system. Sol. Energy. 46, 63-70. doi:10.1016/0038-092X(91)90017-Q.
- Itani, M., Ghali, K., Ghaddar, N., 2015. Displacement Ventilation System Combined with a Novel Evaporative Cooled Ceiling for a Typical Office in the City of Beirut: Performance Evaluation. Proceeding Int. Conf. Renew. Energ. Power Qual. ICREPQ'15, La Coruña, Spain: Renewable Energy and Power Quality Journal (RE&PQJ).
- Jaber, S., Ajib, S., 2011. Evaporative cooling as an efficient system in Mediterranean region. Appl. Therm. Eng. 31, 2590-2596. doi:10.1016/j.applthermaleng.2011.04.026.
- Khalatov, A., Karp, I., Isakov, B., 2011. Prospects of the Maisotsenko thermodynamic cycle application in Ukraine. Int. J. Energy Clean Environ. 12, 141-157. doi:10.1615/InterJEnerCleanEnv.2012005916.
- Khazhmuradov, M., Fedorchenko, D., Rudychev, Y., Martynov, S., Zakharchenko, A., Prokhorets, S, Skrypnyk, A.I., Krugol, M., Yurkin, A., Lukhanin, A.A., Lukhanin, O., 2011. Analysis of the Maisotsenko cycle based cooling system for accumulator batteries. Int. J. Energy Clean Environ. 12, 95-99. doi:10.1615/InterJEnerCleanEnv.2012005979.
- Kozubal, E., Slayzak, S., 2010. Coolerado 5 Ton RTU Performance: Western Cooling Challenge Results (revised). Colorado, USA: National Renewable Energy Laboratory (NREL).
- Kozubal, E., Woods, J., Burch, J., Boranian, A., Merrigan, T., 2011. Desiccant enhanced evaporative air-conditioning (DEVap): evaluation of a new concept in ultra efficient air-conditioning. Golden, Colorado 80401, USA: National Renewable Energy Laboratory (NREL), Technical report: NREL/TP-5500-49722, http://www.nrel.gov/docs/fy11osti/49722.pdf.
- Kozubal, E., Woods, J., Judkoff, R., 2012. Development and analysis of desiccant enhanced evaporative air conditioner prototype. Golden, Colorado 80401, USA: National

Renewable Energy Laboratory (NREL), Technical report: NREL/TP-5500-54755, http://www.nrel.gov/docs/fy12osti/54755.pdf.

- La, D., Dai, Y.J., Li, Y., Wang, R.Z., Ge, T.S., 2010. Technical development of rotary desiccant dehumidification and air-conditioning: A review. Renew. Sustain. Energy Rev. 14, 130-147. doi:10.1016/j.rser.2009.07.016.
- La, D., Dai. Y., Li. Y., Ge. T., Wang, R., 2011. Case study and theoretical analysis of a solar driven two-stage rotary desiccant cooling system assisted by vapor compression air-conditioning. Sol. Energy. 85, 2997-3009. doi:10.1016/j.solener.2011.08.039.
- Lee, J., Lee, D.Y., 2011. Design, fabrication and testing of a compact regenerative evaporative cooler with finned channels. Int. J. Energy Clean Environ. 12, 221-237. doi:10.1615/InterJEnerCleanEnv.2012006393.
- Lee, J., Lee, D.Y., 2013. Experimental study of a counter flow regenerative evaporative cooler with finned channels. Int. J. Heat Mass Transf. 65, 173-179. doi:10.1016/j.ijheatmasstransfer.2013.05.069.
- Lee, S.H., Lee, W.L., 2013. Site verification and modeling of desiccant-based system as an alternative to conventional air-conditioning systems for wet markets. Energy. 55, 1076-1083. doi:10.1016/j.energy.2013.04.029.
- Mahmood, M.H., Sultan, M., Miyazaki, T., Koyama, S., 2015. Evaluation of desiccant airconditioning applicability for agricultural sector of Pakistan. The 90th annual JSRAE conference. 20-23 Oct, 2015. Waseda University, Japan.
- Mahmood, M.H., Sultan, M., Miyazaki, T., Koyama, S., 2016a. Desiccant Air-Conditioning System for Storage of Fruits and Vegetables: Pakistan Preview. Evergr. Jt. J. Nov. Carbon Resour. Sci. Green Asia Strategy. 3, 12-17.
- Mahmood, M. H., Sultan, M., Miyazaki, T., Koyama, S., & Maisotsenko, V. S. 2016b. Overview of the Maisotsenko cycle–A way towards dew point evaporative cooling. Renew. Sustain. Energy Rev. 66, 537-555.
- Maisotsenko, V.S., Gillan, L.E., Heaton, T.L., Gillan, A.D., 2002. Method and apparatus of indirect-evaporation cooling. US Patent No. US6497107 B2.
- Maisotsenko, V., Gillan, L., Heaton, T., Gillan, A., 2004. Power system and method. US Patent No. US20040103637 A1.

- Maisotsenko, V., Reyzin, I., 2005. The Maisotsenko Cycle for Electronics Cooling. Proc. IPACK05 Int. Electron. Packag. Tech. Conf. Exhib., California, USA: ASME. 415-424. doi:10.1115/IPACK2005-73283
- Maisotsenko, V.S., Gillan, L.E., Heaton, T.L., Gillan, A.D., 2005b. Evaporative duplex counterheat exchanger. US Patent No. US6948558 B2.
- Maisotsenko, V.S., Gillan, L.E., Heaton, T.L., Gillan, A.D., 2006. Power system and method. US Patent No. US7007453 B2.
- Maisotsenko, V., Treyger, I., 2011. Way to energy abundance can be found through the Maisotsenko cycle. Int. J. Energy Clean Environ. 12, 319-326. doi:10.1615/InterJEnerCleanEnv.2012005830.
- Mei, L., Dai, Y.J., 2008. A technical review on use of liquid-desiccant dehumidification for air-conditioning application. Renew. Sustain. Energy Rev. 12, 662-689. doi:10.1016/j.rser.2006.10.006.
- Miyazaki, T., Akisawa, A., Nikai, I., 2010a. Study on the Maisotsenko cycle evaporative cooler driven by the solar chimney. Proc. Renew. Energy. Conf. O-Th-2-4, Yokohama, Japan.
- Miyazaki, T., Oda, T., Ito, M., Kawasaki, N., Nikai, I., 2010b. The possibility of the energy cost savings by the electricity driven desiccant system with a high performance evaporative cooler. Int. Symp. Innov. Mater. Process. Energy Syst.
- Miyazaki, T., Akisawa, A., Nikai, I., 2011a. The cooling performance of a building integrated evaporative cooling system driven by solar energy. Energy Build. 43, 2211-2218. doi:10.1016/j.enbuild.2011.05.004.
- Miyazaki, T., Nikai, I., Akisawa, A., 2011b. Simulation analysis of an open-cycle adsorption air-conditioning system–numeral modeling of a fixed bed dehumidification unit and the Maisotsenko Cycle cooling unit. Int. J. Energy Clean Environ. 12, 341-354. doi:10.1615/InterJEnerCleanEnv.2012005977.
- Mohammad, A.T., Mat, S.B., Sulaiman, M.Y., Sopian, K., Al-abidi, A.A., 2013a. Survey of hybrid liquid desiccant air-conditioning systems. Renew. Sustain. Energy Rev. 20, 186-200. doi:10.1016/j.rser.2012.11.065.
- Mohammad, A.T., Mat, S.B., Sulaiman, M.Y., Sopian, K., Al-abidi, A.A., 2013b. Artificial neural network analysis of liquid desiccant regenerator performance in a solar hybrid

air-conditioning system. Sustain. Energy Technol. Assess. 4, 11-19. doi:10.1016/j.seta.2013.08.001.

- Mohammad, A.T., Mat, S.B., Sulaiman, M.Y., Sopian, K., Al-abidi, A.A., 2013c. Artificial neural network analysis of liquid desiccant dehumidifier performance in a solar hybrid air-conditioning system. Appl. Therm. Eng. 59, 389-397. doi:10.1016/j.applthermaleng.2013.06.006.
- Nagaya, K., Senbongi, T., Li, Y., Zheng, J., Murakami, I., 2006. High energy efficiency desiccant assisted automobile air-conditioner and its temperature and humidity control system. Appl. Therm. Eng. 26, 1545-1551. doi:10.1016/j.applthermaleng.2005.12.005.
- Ndukwu, M.C., Manuwa, S.I., 2015. Impact of evaporative cooling preservation on the shelf life of fruits and vegetable in South Western Nigeria. Res. Agric. Eng. 61, 122-128.
- Novoselac, A., Srebric, J., 2002. A critical review on the performance and design of combined cooled ceiling and displacement ventilation systems. Energy Build. 34, 497-509. doi:10.1016/S0378-7788(01)00134-7.
- Olosunde, W.A., Aremu, A.K., Onwude, D.I., 2015. Development of a solar powered evaporative cooling storage system for tropical fruits and vegetables. J. Food Process Preserv. DOI: 10.1111/jfpp.12605.
- Pandelidis, D., Anisimov, S., 2015a. Numerical analysis of the selected operational and geometrical aspects of the M-cycle heat and mass exchanger. Energy Build. 87, 413-424. doi:10.1016/j.enbuild.2014.11.042.
- Pandelidis, D., Anisimov, S., 2015b. Numerical analysis of the heat and mass transfer processes in selected M-Cycle heat exchangers for the dew point evaporative cooling. Energy Convers. Manag. 90, 62-83. doi:10.1016/j.enconman.2014.11.008.
- Pandelidis, D., Anisimov, S., Worek, W.M., 2015a. Performance study of the Maisotsenko Cycle heat exchangers in different air-conditioning applications. Int. J. Heat Mass Transf. 81, 207-221. doi:10.1016/j.ijheatmasstransfer.2014.10.033.
- Pandelidis. D., Anisimov, S., Worek, W.M., 2015b. Comparison study of the counter-flow regenerative evaporative heat exchangers with numerical methods. Appl. Therm. Eng. 84, 211-224. doi:10.1016/j.applthermaleng.2015.03.058.

- Qi, R., Lu, L., 2014. Energy consumption and optimization of internally cooled/heated liquid desiccant air-conditioning system: A case study in Hong Kong. Energy. 73, 801-808. doi:10.1016/j.energy.2014.06.086.
- Qi, R., Lu, L., Huang, Y., 2014. Energy performance of solar-assisted liquid desiccant airconditioning system for commercial building in main climate zones. Energy Convers. Manag. 88, 749-757. doi:10.1016/j.enconman.2014.09.006.
- Rees, S.J., Haves, P., 2013. An experimental study of air flow and temperature distribution in a room with displacement ventilation and a chilled ceiling. Build. Environ. 59, 358-368. doi:10.1016/j.buildenv.2012.09.001.
- Reznikov, M., 2011. Electrostatic enforcement of heat exchange in the Maisotsenko-cycle system. Int. J. Energy Clean Environ. 12, 117-127. doi:10.1615/InterJEnerCleanEnv.2012005850.
- Riangvilaikul, B., Kumar, S., 2010a. An experimental study of a novel dew point evaporative cooling system. Energy Build. 42, 637-644. doi:10.1016/j.enbuild.2009.10.034.
- Riangvilaikul, B., Kumar, S., 2010b. Numerical study of a novel dew point evaporative cooling system. Energy Build. 42, 2241-2250. doi:10.1016/j.enbuild.2010.07.020.
- Rogdakis, E.D., Koronaki, I.P., Tertipis, D.N., 2014. Experimental and computational evaluation of a Maisotsenko evaporative cooler at Greek climate. Energy Build. 70, 497–506. doi:10.1016/j.enbuild.2013.10.013.
- Saghafifar, M., Gadalla, M., 2015. Innovative inlet air cooling technology for gas turbine power plants using integrated solid desiccant and Maisotsenko cooler. Energy. 87, 663-677. doi:10.1016/j.energy.2015.05.035.
- Sarkar, J., Ejector enhanced vapor compression refrigeration and heat pump systems—A review. Renew. Sustain. Energy Rev. 6, 6647-6659. doi:10.1016/j.rser.2012.08.007.
- Sultan, M., Miyazaki, T., Koyama, S., Saha, B.B., 2014. Utilization of Desiccant Air-Conditioning System for Improvement in Greenhouse Productivity: A Neglected Area of Research in Pakistan. Int. J. Environ. 4, 1-10
- Sultan, M., El-Sharkawy, II., Miyazaki, T., Saha, B.B., 2015a. Koyama S. An overview of solid desiccant dehumidification and air-conditioning systems. Renew. Sustain. Energy Rev. 46:16-29. doi:10.1016/j.rser.2015.02.038.

- Sultan, M., 2015b. Study on sorption characteristics of water adsorbents for agricultural airconditioning systems. D.Eng. thesis. Kyushu University, Japan.
- Sun, D.W., 1997. Solar powered combined ejector-vapour compression cycle for airconditioning and refrigeration. Energy Convers. Manag. 38, 479–491. doi:10.1016/S0196-8904(96)00063-5.
- Taki, A.H., Jalil, L., Loveday, D.L., 2011. Experimental and computational investigation into suppressing natural convection in chilled ceiling/displacement ventilation environments. Energy Build. 43, 3082–3089. doi:10.1016/j.enbuild.2011.08.002.
- Tertipis, D., Rogdakis, E., 2015. Maisotsenko cycle: technology overview and energysaving potential in cooling systems. Energy Emiss. Control Technol. 3, 15–22. doi:http://dx.doi.org/10.2147/EECT.S62995.
- Tyagi, S.K., Wang, S., Park, S.R., Sharma, A., 2008. Economic considerations and cost comparisons between the heat pumps and solar collectors for the application of plume control from wet cooling towers of commercial buildings. Renew. Sustain. Energy Rev. 12, 2194-2210. doi:10.1016/j.rser.2007.03.012.
- Tyagi, S.K., Pandey, A.K., Pant, P.C., Tyagi, V.V., 2012. Formation, potential and abatement of plume from wet cooling towers: A review. Renew. Sustain. Energy Rev. 16, 3409-3429. doi:10.1016/j.rser.2012.01.059.
- Weerts, B., 2011. Coolerado and modeling an application of the Maisotsenko Cycle. Int. J. Energy Clean Environ. 12, 287-307. doi:10.1615/InterJEnerCleanEnv.2013005585.
- Weerts, B.A., Gallaher, D., Weaver, R., Van Geet, O., 2012. Green Data Center Cooling: Achieving 90% Reduction: Airside Economization and Unique Indirect Evaporative Cooling. IEEE Green Technol. Conf., Tulsa, Oklahoma: IEEE; 2012, p. 1–6. doi:10.1109/GREEN.2012.6200950.
- Weerts, B.A., 2012. NSIDC green data center project: Coolerado and modeling an application of the Maisotsenko cycle. M.S. Thesis. University of Colorado Boulder.
- Woods, J., Pellegrino, J., Kozubal, E., Slayzak, S., Burch. J., 2009. Modeling of a membrane-based absorption heat pump. J. Membr. Sci. 337, 113-124. doi:10.1016/j.memsci.2009.03.039.

- Woods, J., Pellegrino, J., Kozubal, E., Burch, J., 2011. Design and experimental characterization of a membrane-based absorption heat pump. J. Membr. Sci. 378, 85-94. doi:10.1016/j.memsci.2010.11.012
- Woods, J., Kozubal, E., 2012. Desiccant enhanced evaporative air-conditioning: parametric analysis and design. Proc. Second Int. Conf. Build. Energy Environ. COBEE-2012, Boulder, Colorado, USA.
- Woods, J., Kozubal, E., 2103a. A desiccant-enhanced evaporative air conditioner: Numerical model and experiments. Energy Convers. Manag. 65, 208-220. doi:10.1016/j.enconman.2012.08.007.
- Woods, J., Kozubal, E., 2013b. Heat transfer and pressure drop in spacer-filled channels for membrane energy recovery ventilators. Appl. Therm. Eng. 50, 868-876. doi:10.1016/j.applthermaleng.2012.06.052.
- Worek, W.M., Khinkis, M., Kalensky, D., Maisotsenko, V., 2012. Integrated Desiccant– Indirect Evaporative Cooling System Utilizing the Maisotsenko Cycle. Proc. ASME Summer Heat Transf. Conf., Puerto Rico: ASME. 21-28. doi:10.1115/HT2012-58039.
- Xuan, Y.M., Xiao, F., Niu, X.F., Huang, X., Wang, S.W., 2012. Research and application of evaporative cooling in China: A review (I) – Research. Renew. Sustain. Energy Rev. 16:3535–46. doi:10.1016/j.rser.2012.01.052.
- Yang, B., Yuan, W., Gao, F., Guo, B., 2015. A review of membrane-based air dehumidification. Indoor Built Environ. 24, 11–26. doi:10.1177/1420326X13500294.
- Zhan, C., Duan, Z., Zhao, X., Smith, S., Jin, H., Riffat, S., 2011a. Comparative study of the performance of the M-cycle counter-flow and cross-flow heat exchangers for indirect evaporative cooling – Paving the path toward sustainable cooling of buildings. Energy. 36, 6790-6805. doi:10.1016/j.energy.2011.10.019.
- Zhan, C., Zhao, X., Smith, S., Riffat, S.B., 2011b. Numerical study of a M-cycle cross-flow heat exchanger for indirect evaporative cooling. Build. Environ. 46, 657-668. doi:10.1016/j.buildenv.2010.09.011.
- Zhao, X., Li, J.M., Riffat, S.B., 2008. Numerical study of a novel counter-flow heat and mass exchanger for dew point evaporative cooling. Appl. Therm. Eng. 28, 1942-1951. doi:10.1016/j.applthermaleng.2007.12.006.

- Zhu, J., Chen, W., 2014. Energy and exergy performance analysis of a marine rotary desiccant air-conditioning system based on orthogonal experiment. Energy. 77, 953– 962. doi:10.1016/j.energy.2014.10.014.
- Zube, D., Gillan, L., 2011. Evaluating Coolerado Corportion's heat-mass exchanger performance through experimental analysis. Int. J. Energy Clean Environ. 12, 101-116. doi:10.1615/InterJEnerCleanEnv.2012005839.

CHAPTER 4

EXPERIMENTAL PERFORMANCE EVALUATION OF DESICCANT UNIT

CHAPTER 4

EXPERIMENTAL PERFORMANCE EVALUATION OF DESICCANT UNIT

This chapter presents the setup of an open-cycle experimental apparatus for desiccant dehumidification analysis. Honeycomb like cubical shaped polymer desiccant blocks are used for the purpose. The experiments are performed under different ambient conditions, and various regeneration air temperatures and switching time ratios (five cases). The generalized root of sum of squares method is used to calculate the experimental uncertainty. The net and average effective dehumidification performances of desiccant unit are determined and presented. Influence of regeneration temperature on equivalent heat of adsorption is ascertained. The optimized switching time ratio between regeneration and dehumidification of desiccant unit is suggested. A novel correlation is developed on the conception of modification of slope of dehumidification under varying regeneration temperature is also investigated. The correlation leads toward the steady-state analysis of desiccant air-conditioning systems.

4.1 Introduction

Low-cost energy-efficient air-conditioning systems for human and nonhuman applications are the dire need of the day due to increasing threats of global warming and climate change. The evaporative cooling technologies are considered low-cost options but
Chapter 4

cannot be used as standalone under varying climatic (humid) conditions (see appendices). In this regard, the desiccant dehumidification plays a pivotal role to enhance the scope of evaporative cooling in the form of desiccant air-conditioning. The experimental investigation of dehumidification performance of desiccant unit is always required for the designing of optimized DAC system for certain application. Moreover, the experimentally validated models/correlations need to be developed for the particular adsorbent. Such correlations ease the dehumidification analyses for certain applications under varying climates/regions. Therefore, in this chapter an open-cycle experimental apparatus was setup for the performance evaluation of hydrophilic polymer based desiccant blocks. Series of experiments are conducted for various: ambient air conditions, regeneration conditions, cycle time, and switching time. Generalized root sum of squares method is used to calculate the experimental uncertainty. A simplified correlation is developed, on the basis of bunch of experiments, by which real desiccant dehumidification process can be predicted on psychrometric chart for polymeric desiccants. The correlation leads toward the stead-state analysis of desiccant air-conditioning systems. The analyses conducted in the next chapter (chapter 5) are based on this novel and simplified correlation for various applications.

4.2 Experimental Section

4.2.1 Adsorbent/Desiccant

Honeycomb like cubical shaped desiccant blocks were used for desiccant AC application. Study utilizes eight desiccant blocks ($20 \times 20 \times 20 \times 20 \text{ cm}^3$ each) for the experimental analysis. The desiccant blocks were composed of hydrophilic organic polymer based sorbent (HPBS) (Mahmood et al., 2016, Sultan et al., 2016a) provided by Showa Manufacturing Co., Ltd., Japan. The pictorial and magnifying view of a desiccant block is shown in Figure 4.1. The parametric description of the desiccant block is given in Table 4.1 (Yoshida, 2014). Moreover, the close and open cycle kinetics parametric values of desiccant blocks are described in Table 4.2 (Sultan et al., 2016a). It can be seen (Figure 4.1) that each block contains series of honeycomb like channels. During the experimentation air at designated flow rate was passed through these successive channels.

The cubical shape of desiccant block facilitates easy integration in the system and may also provide ease in batch processing during regeneration and dehumidification.



Figure 4.1 Pictorial and magnifying view of the desiccant block.

Table 4.1 Parametric description of desiccant block (Yoshida, 2014).

Parameters description	Values
Dimensions of block [cm]	20 x 20 x 20
Half thickness of air layer [mm]	1.7
Half thickness of sorbent [mm]	0.31
Sorbent density [kg-sorbent/m ³]	1500
Sorption heat [kJ/kg-water]	2.5×10^3
Effective diffusion coefficient [m ² /sec]	7.0 x 10 ⁻⁴

Average	$F_0 \frac{D_s}{R^2 p}$	Average $F_0 \frac{D_s}{R^2p}$	Average	$F_0 \frac{D_s}{R^2 p}$	Average $F_0 \frac{D_s}{R^2 p}$
T _{ads} [K]	[value x 10 ⁻⁴]	[value x 10 ⁻⁴]	T _{ads} [K]	[value x 10 ⁻⁴]	[value x 10 ⁻⁴]
Open-cyc	le adsorption kir	netics	Close-cycle	adsorption kinet	ics
296.24	2.51		296.70	3.26	
296.22	2.47		296.66	2.23	
295.75	2.36	1.90	296.67	2.51	2 92
295.68	2.06	1.90	295.15	2.33	2.92
294.90	2.06		295.15	1.31	
294.30	1.17		295.10	5.89	

Table 4.2 Close and open cycle kinetics parameters values of desiccant block (Sultan et al., 2014a).

4.2.2 Experimental Setup

An open-cycle experimental apparatus was setup in this study. It consists of desiccant blocks, air filter, air flow valves, air blowers, water to air heat exchanger, temperature control water bath, water circulator, temperature and relative humidity sensors, differential pressure transmitter, data logger. On the basis of these components the experimental setup is divided into five units such as desiccant block unit (DBU), regeneration air unit (RAU), supplementary heating unit (SHU), process air unit (PAU), and flow rate control unit (FCU). As far DBU is concerned it consists of eight desiccant blocks (HPBS), air filter, inlet and outlet temperature and relative humidity sensors (VAISALA: HMT 333, $RH_{ac} = \pm 1-1.7\%$; $T_{ac} = \pm 0.2-0.3^{\circ}C$). The orientation of desiccant block unit in experimental setup is shown in Figure 4.2. The desiccant blocks are placed in the experimental setup in such a way (two rows and columns) that it has dimensions of (40 x 40 x 40 cm³). A netted filter was used in order to avoid the entry of any foreign material to the HPBS. The air temperature and relative humidity at the inlet and outlet of the HPBS is measured by sensors (VAISALA: HMT 333). The detail about other units is described in the relevant subsequent headings.



Figure 4.2 Detailed illustration of experimental setup.

4.2.3 Development of Equilibrium Conditions

The experimental setup was placed in an environmentally controlled room. The room was therefore equipped with central air-conditioning (AC) unit which maintains its temperature and relative humidity. The required room temperature and relative humidity are regulated through control panel located outside of the experimental room. The main interface of the control panel has the options to start/stop the system, stop buzzing alarm, reset alarm and menu. It also continuously displays the preset, and achieved value of temperature and relative humidity. The temperature and relative humidity values can also

be changed. It must always be confirmed before starting the AC unit that water supply valve connected to the humidifier is open. However, in case of any unusual operation/mishandling the unit starts buzzing alarm and indicate the reason of alarming. The buzzing of alarm can be due to the abnormality in: blower, heater, humidifier, air conditioner, upper/lower limit of temperature and relative humidity, and due to overheat. The indoor environmentally equilibrium conditions were achieved before start of each experiment as explained above. Such conditions permit to analyze the repeatable performance of the desiccant blocks.

4.2.4 Adjustment of Operating Conditions

4.2.4.1 Regeneration Air Flow

The desiccant blocks are substantially regenerated before the dehumidification process. The hot regeneration air (RA) is supplied for the purpose through RAU. The RAU consists of regeneration air circulator with temperature and relative humidity sensor (APSITE: PAU-H3200-6KHC, $T_{ac} = \pm 0.5^{\circ}$ C; $RH_{ac} = \pm 2\%$), supply and exhaust air duct, water source, and air flow valves. The orientation of RAU in experimental setup is shown in Figure 4.2. It can supply the regeneration air at temperature and relative humidity ranging from 0-55°C, and 0-85% respectively. The supply of water to the RAU must always be ensured in order to have active control over relative humidity. Moreover, the temperature and relative humidity sensors of RAU must be placed properly in its supply air duct or within the experimental setup. Its front panel contains temperature and relative humidity controller, RUN/STOP and Drain ON/OFF buttons/switches, and light indications about power, alarm, fan, heater, humidifier and water. The adjustments of regeneration air temperature and relative humidity was made by regulating the RAU as follows

- i. ensure that water supply valve is open and there is no water lock in supply line
- ii. ensure that drain switch is in OFF position
- iii. check that all the front control panel light indications turned green except alarm indication (which remains off) during normal working operation
- iv. adjust the temperature and relative humidity as set values (SV)

- v. preset values (PV) of temperature and relative humidity start increasing/decreasing as per required SV
- vi. wait until the PV matches to SV of temperature and relative humidity, during this period air was exhausted outside of experimental room through exhaust channel
- vii. regeneration air temperature and relative humidity as reached to desire set value, close the exhaust valve in order to pass the hot regeneration air through the desiccant blocks

It is important to mention that the RAU increases the temperature of the air up to 55°C, however further increase in air temperature was made by employing SHU. The SHU consists of air to water heat exchanger (HX), temperature control water bath (ADVANTEC: TBN402DA, $T_{ac} = \pm 0.1^{\circ}$ C), water circulator (EYELA CTP-3000, $T_{ac} = \pm 0.1^{\circ}$ C) and accessories. The temperature control water bath can raise the water temperature up to 90°C. The hot water from temperature control water bath is supplied to the HX through water circulator. Moreover, the heating mode provision of the water circulator enable to control any fluctuation in the temperature of water supplied to the HX. The hot water circulating in the HX exchanges heat with air passing over it. The warm water coming out from the HX is flowed to the temperature control water bath in order to maintain the hot water supply cycle.

4.2.4.2 Process Air Flow

The air supplied to the desiccant blocks during dehumidification mode is termed as process air (PA). It was supplied to the desiccant blocks through process air unit (PAU). The PAU consists of process air circulator with temperature sensor (APSITE: PAU-AZ1800SE, $T_{ac} = \pm 0.05 \cdot 0.1^{\circ}$ C), supply and exhaust air duct, and air flow valves. The orientation of PAU in experimental setup is shown in Figure 4.2. It can supply the air at temperature ranging from 8°C to 47°C (when damper fully opened), however the relative humidity is automatically adjusted as per the room ambient conditions. The temperature sensor needs to be place inside the outlet air duct of the PAC or inside the experimental setup. The process air temperature during experimentation was adjusted using SEL and Up/Down push buttons. The desired set value (SV) of the temperature was achieved in couple of minutes depending upon the preset value and ambient environmental conditions. The air was exhausted outside of the experimental room till it achieves the temperature set value. It is important to mention that the experimental setup also has option to use the process air circulator as regeneration air circulator. The temperature of the air thus can be increased by passing it through HX of SHU in order to use it as regeneration air. However, when both the PAU and RAU were employed, the process air by-passed the heat exchanger.

4.2.4.3 Air Flow Rate

The regeneration and process air flow rate during experiments is regulated by FCU. It consist of circular orifice, differential pressure transmitter (TESTO: 6349, $P_{ac} = 0.35Pa + 0.6\%$ of full scale) and variable speed blower (SHOWA: EC-100T-R313) with accessories as shown in Figure 4.3. The flow rate of PA and RA is regulated by changing the speed of blower and measuring the pressure difference (ΔP) across the circular orifice. The ΔP across the circular orifice was measured by differential pressure transmitter. In this regard, its positive and negative pressure hose are connected with the inlet and outlet hose of the orifice, respectively. Therefore, changes in the pressure across the orifice create movement in its membrane spring which oscillates between two inductive transmitters to indicate the ΔP . The flow rate of PA and RA are then calculated using particular orifice Eq. (4.1) (ASHRAE, 2009).

$$m = C_d A \sqrt{2 \rho \Delta P} \tag{4.1}$$

where *m* is air mass flow rate [kg/sec], C_d is the discharge coefficient [-], *A* is the area of the orifice [m²], ΔP is air pressure difference between inlet and outlet of the orifice [Pa] and ρ is the density of the air [kg/m³]. The diameter of the orifice and discharge coefficient is taken as 8.0 cm and 0.6, respectively.



Figure 4.3 Illustration of flow control unit of experimental setup.

4.2.5 Experimental Procedure

The ambient conditions of the experimental room are always maintained before start of experiments as explained in heading 4.2.3. Afterwards, experiments are started by performing initialization of desiccant blocks. During initialization desiccant blocks are first regenerated and then dehumidified in order to provide the thermodynamically equilibrium conditions for subsequent regeneration and dehumidification processes. It is important to mention that the experimental data taken after the initialization is used in the analyses. Referring to Figure 4.2 the regeneration air is supplied to the desiccant blocks through RAU. The regeneration was performed at varying temperature (40-60°C). The regeneration at low temperatures was selected in order to simulate the conditions that can be available through the use of low grade waste heat, solar energy (preferably solar thermal) and biogas. As RAU increases the regeneration air temperature up to 55°C, therefore further increase in regeneration air temperature (when required) was made using SHU. Such adjustments (T and RH) in regeneration air were made as explained in heading 4.2.4.1. After the regeneration of the desiccant blocks, the process air is supplied to the desiccant block through PAU. The varying process air conditions (Tin and RHin) were provided to the desiccants block in order to analyze the dehumidification performance of the desiccant blocks. The switching time ratios between regeneration and dehumidification modes were

also varied (as 1:1, 2:3, 1:2) during the experiments. The time ratio (1:1) means that after performing 60 minutes regeneration the desiccant blocks are switched for 60 minutes dehumidification. Moreover, the flow rate of PA and RA were regulated by FCU as explained in heading 4.2.4.3. The instantaneous values of temperature, relative humidity and differential pressure during experiments were recorded by data logger (DAQMASTER: MX 100, accuracy = $\pm 0.01\%$) connected with personal computer. It was used to record the data for every 10 second interval. The representation of experimental data recoding sheet is shown in Table 4.3.

 Table 4.3 Data recording sheet.

Time (sec)	ΔP (Pa)	$T_{in}(^{\circ}C)$	$T_{out}(^{\circ}C)$	$RH_{in}(\%)$	RH _{out} (%)
-	-	-	-	-	-
-	-	-	-	-	-
-	-	-	-	-	-
-	-	-	-	-	-

4.2.6 Uncertainty Analysis

The uncertainty analysis of the measured and/or calculated variables was carried out to determine the accuracy of the experimental data. The uncertainty in experimental data depends on the instrument type, instrument calibration, ambient conditions, observations, readings, data recordings, experiment planning and other associated factors (Hepbasli and Akdemir, 2004). The uncertainty in the resultant function (σ_R) of the independent variables can be determined by the generalized root of sum of squares method (Esen et al. 2006; Holman, 2012) as described in Eq. (4.2).

$$\sigma_R = \sqrt{\left(\frac{\partial R}{\partial N_1}\sigma_1\right)^2 + \left(\frac{\partial R}{\partial N_2}\sigma_2\right)^2 + \dots + \left(\frac{\partial R}{\partial N_n}\sigma_n\right)^2} \tag{4.2}$$

where $\sigma_1, \sigma_2, ..., \sigma_n$ represents the uncertainties in the independent variables $N_1, N_2, ..., N_n$. Whereas, the result *R* is given function of independent variables. The total uncertainties (σ_t) in RH, T, and ΔP measurements are however calculated by Eq. (4.3) as cited in the literature (Hepbasli and Akdemir, 2004; Uçkan et al. 2013).

$$\sigma_t = \sqrt{\sigma_{sr}^2 + \sigma_{cp}^2 + \sigma_{du}^2} \tag{4.3}$$

where σ_{sr} , σ_{cp} and σ_{du} describe the uncertainties of sensor, calibration process, and data acquisition unit, respectively. The values of these uncertainties are provided by the particular instrument's manufacturer. Thus, the total uncertainties in the measurements of relative humidity, temperature and pressure are calculated as ±1.97%, ±1.10% and ±2.47%, respectively. However, the percent uncertainty in the mass flow rate $(\frac{\sigma_m}{m})$ (Hepbasli and Akdemir, 2004; Holman, 2012) is calculated using the Eq. (4.5). The Eq. (4.5) is obtained by the algebraic manipulation of Eqs. (4.1), (4.2) and (4.4).

$$\sigma_m = \sqrt{\left(\frac{\partial m}{\partial \Delta P}\right)^2 \sigma_{\Delta p}^2 + \left(\frac{\partial m}{\partial \rho}\right)^2 \sigma_{\rho}^2} \tag{4.4}$$

$$\frac{\sigma_m}{m} = \sqrt{0.25 \left(\frac{\sigma_{\Delta P}}{\Delta P}\right)^2 + 0.25 \left(\frac{\sigma_{\rho}}{\rho}\right)^2} \tag{4.5}$$

The uncertainty in calculating the mass flow rate is found as $\pm 1.81\%$. It is important to mention that the uncertainty in the mass flow rate is mainly due to the uncertainty in the measurement of pressure difference across the orifice.

4.3 Theory and Methods

Performance evaluation of desiccant blocks is experimentally investigated using an open-cycle experimental setup (Figure 4.2) for various process and regeneration air

conditions. The experiments were initially conducted for two reference inlet air conditions case-A ($T_{in} \approx 31^{\circ}C$, $X_{in} \approx 6$ g/kg-DA) and case-B ($T_{in} \approx 13^{\circ}C$, $X_{in} \approx 6$ g/kg-DA) as briefly elaborated in Table 4.4. The inlet and outlet air conditions of desiccant unit are measured experimentally whereas rests of the conditions are calculated by Eqs. (4.6)-(4.14) (ASHRAE, 2009; Hyland and Wexler 1983; Stull, 2011). The development of experimental setup and its operating procedure is however discussed in heading 4.2.

$$X = 0.621945 \ \frac{P_v}{P_{atm} - P_v} \tag{4.6}$$

$$P_{\nu} = P_{\nu s} R H \tag{4.7}$$

$$lnP_{\nu s} = \left[\frac{\zeta_1}{T} + \zeta_2 + \zeta_3 T + \zeta_4 T^2 + \zeta_5 T^3 + \zeta_6 ln T\right] / 1000$$
(4.8)

$$T_{dp>0} = \zeta_7 + \zeta_8 \,\alpha + \zeta_9 \,\alpha^2 + \zeta_{10} \,\alpha^3 + \,\zeta_{11} \,(P_\nu)^{0.1984} \tag{4.9}$$

$$T_{dp<0} = 6.09 + 12.608 \,\alpha + 0.4959 \,\alpha^2 \tag{4.10}$$

$$\alpha = \ln P_{\nu} \tag{4.11}$$

$$\zeta_{1} = -5.8002206E + 03$$

$$\zeta_{2} = 1.3914993E + 00$$

$$\zeta_{3} = -4.8640239E - 02$$

$$\zeta_{4} = 4.1764768E - 05$$

$$\zeta_{5} = -1.4452093E - 08$$

$$\zeta_{6} = 6.5459673E + 00$$

$$\zeta_{7} = 6.54$$

$$\zeta_{8} = 14.526$$

$$\zeta_{9} = 0.7389$$

$$\zeta_{10} = 0.09486$$

$$\zeta_{11} = 0.4569$$

$$\rho = \frac{P}{R_{DA}T} \tag{4.12}$$

where X is humidity ratio [kg/kg-DA], P_v is partial pressure of water vapor [kPa], P_{vs} is saturation pressure of water vapor [kPa], P_{atm} is atmospheric air pressure [kPa], P is air pressure [Pa], ρ is air density [kg/m³], R_{DA} is specific gas constant of dry air [J/kg·K], *RH* is relative humidity [-], T is absolute temperature [K], T_{dp} is dew-point temperature [°C].

The mass flow rate of process air (m_{PA}) and regeneration air (m_{RA}) was determined (Eq. 4.1) as 0.1 kg/sec during the desiccant block dehumidification and/or regeneration open-cycle experiments. Therefore, the process to regeneration air flows ratio was 1:1 during experiments and analyses. The air enthalpy [ASHRAE, 2009] and wet-bulb temperature [Stull, 2011] are however calculated by Eq. (4.13) and Eq. (4.14), respectively.

$$h = 1.006 T + X (2501 + 1.86 T)$$
(4.13)

$$T_{wb} = T \tan^{-1} (0.151977 * \sqrt{RH} + 8.313659) + \tan^{-1} (T + RH) - \tan^{-1} (RH - 1.676331) + 0.00391838 (RH)^{3/2} \tan^{-1} (0.023101 RH) - 4.686035$$
(4.14)

where *h* is air enthalpy [kJ/kg-DA], T_{wb} is wet-bulb temperature [°C], *T* is air dry bulb temperature [°C], *RH* in case of calculating wet-bulb temperature is in percentage [%].

The net water vapors adsorption/desorption amount (ΔX) is calculated by Eq. (4.15). The average dehumidification ($\overline{\Delta X}$) is calculated by Eq. (4.16). Whereas the effective average dehumidification ($\overline{\Delta X}$) is calculated by Eq. (4.17).

$$\Delta X = X_{in} - X_{out} \tag{4.15}$$

$$\overline{\Delta X} = \frac{1}{\overline{n}} \sum_{i=1}^{\overline{n}} (X_{in} - X_{out})_i$$
(4.16)

$$\overline{\Delta X} = \frac{1}{\overline{\overline{n}}} \sum_{i=1}^{\overline{n}} (X_{in} - X_{out})_i$$
(4.17)

where \bar{n} represents number of experimental entries during dehumidification cycle and \bar{n} represents the number of experimental entries during dehumidification and regeneration cycle. In other words, $\overline{\Delta X}$ deals the dehumidification cycle whereas $\overline{\Delta X}$ presents the combined impression of dehumidification and regeneration.

4.4 **Results and Discussion**

4.4.1 Performance Evaluation of Desiccant Unit

An open-cycle desiccant block experimental setup was developed for the performance evaluation of desiccant blocks at different process and regeneration air conditions (Table 4.4 and Table 4.5). The study initially focuses two reference inlet air conditions i.e. case-A ($T_{in} \approx 31^{\circ}$ C, $X_{in} \approx 6$ g/kg-DA) and case-B ($T_{in} \approx 13^{\circ}$ C, $X_{in} \approx 6$ g/kg-DA).

The regeneration of desiccant blocks was made at low regeneration temperatures (Table 4.4) for both the cases (A and B) in order to simulate the conditions that can be available through the use of low grade waste heat, solar energy, bio-gas etc. The air mass flow rate for all experiments was used as 0.1 kg/sec. Whereas, the regeneration to process air flow ratio was maintained at 1:1. The experimental profiles of dynamic behavior of process and regeneration air streams of case-A and case-B at regeneration temperature (T_{reg}) of 60°C are shown by Figure 4.4(a) and 4.5(b), respectively. Similarly, the experimental temperature and relative humidity profiles at T_{reg} = 40°C for case-A and case-B are however shown in Figure 4.5(a) and 4.6(b), respectively. Referring to dehumidification cycle in Figure 4.4(b), ambient air enters in to desiccant block at RH~65-70% and exits at RH~12-25%. It shows huge potential of air dehumidification which can be



Figure 4.4 Experimental profiles of ambient air conditions on desiccant dehumidification performance at 60°C regeneration temperature: (a) case-A, and (b) case-B.



Figure 4.5 Experimental profiles of ambient air conditions on desiccant dehumidification performance at 40°C regeneration temperature: (a) case-A, and (b) case-B.

Process air		ocess air	Rege	neration air	Switching time (Reg:Deh)
Case	T _{in} [°C]	X _{in} [g/kg-DA]	T _{in} [°C]	X _{in} [g/kg-DA]	[min]
А	31	6	40-60	6	60:90
В	13			5	

 Table 4.4 Experiments under varying regeneration temperatures and constant switching time ratio.

utilized for various DAC applications. For example, return air from postharvest storage zone I to III (or ambient air) (see chapter 5) can be successively dehumidified to lower relative humidity in order to provide optimum storage conditions. Consequently, required level of respiration, transpiration and fermentation can be maintained by means of exposure of dehumidified return/fresh air. Similarly, desiccant based systems can successfully achieved required conditions of temperature and relative humidity for various applications e.g. agricultural greenhouses (Sultan et al., 2016b), animal AC (Sultan et al., 2017), wet markets (Lee and Lee, 2013). It is important to mention that desiccant dehumidifies the air at one side, and heated the air on the other side due to heat of adsorption/condensation. The present study presents the similar behaviour of desiccant dehumidification (Figure 4.4 and Figure 4.5) in which the dehumidified air is heated up more than the heat of condensation due to the isosteric heat of adsorption. Most of the DAC systems possess the desiccant unit followed by the heat exchanger (HX) in order to recover the sensible energy, consequently similar conception has also been considered in the proposed DAC systems (chapter 5). Hence adsorption heat generates can be somehow managed by HX unit.

From the above prospective, Figure 4.6 shows two identical experiments at regeneration temperature of 40°C and 60°C in order to investigate adsorption heat. In the present study effect of adsorption heat has been realized by equivalent heat of adsorption (q_{eq}) as expressed by Eq. (4.18).

 $q_{eq} = m \left(h_{out} - h_{in} \right) \tag{4.18}$

where *m* is air mass flow rate [kg/sec] and *h* is air enthalpy [kJ/kg]. The numerical values of q_{eq} in both cases is increasing with the regeneration temperature, however decreases temporarily at all regeneration temperatures. This phenomena of adsorption heat (Q_{st}) is obvious and can be understood from Clausius–Clapeyron relationship as expressed by Eq. (4.19) (Sultan et al., 2015).



Figure 4.6 Influence of regeneration temperature on equivalent adsorption heat for: (a) case-A, and (b) case-B.



Figure 4.7 Effect of regeneration temperature on net dehumidification and regeneration performance for: (a) case-A, and (b) case-B. Negative and positive values indicate regeneration and dehumidification cycle, respectively.

$$\frac{Q_{st}}{R_v} = -\left[\frac{\partial \ln P_v}{\partial \left(\frac{1}{T_{db}}\right)}\right]$$
(4.19)

where P_v , R_v , and T_{db} represent the partial water vapor pressure [kPa], water vapor gas constant [kJ/kgK], and dry bulb temperature [K], respectively. Referring to Clausius–Clapeyron equation the adsorption heat is the function of adsorbed water vapors, therefore variation in q_{eq} in both cases and different regeneration temperatures (Figure 4.6) is due to the amount of adsorbed water vapors. The q_{eq} line approaches to net value equals zero indicates the isenthalpic adsorption or zero adsorption phenomena. This representation is handy in order to quantify the desiccant unit switching time (t). For example in Figure 4.6(b) process air condition at t = 60 min indicates that the desiccant still has ability to dehumidify the air, therefore dehumidification cycle time can be increased accordingly at both regeneration temperatures. On the other hand in Figure 4.6(a), the q_{eq} of process air (t = 60 min) is approaches to zero which shows that the desiccant probably has been saturated and therefore should be switched for regeneration. Similarly in another study (Sultan et al., 2017) effect of adsorption heat on desiccant performance has been reported.

Effect of regeneration temperatures on net water vapors adsorption/desorption amount per kg of dry air has been investigated using Eq. (4.15) at T_{reg} of 40°C and 60°C as shown in Figure 4.7(a) and 4.7(b) for case-A and case-B, respectively. According to the thermodynamics laws total water vapor adsorption (dehumidification cycle) is equal to total water vapor desorption amount (regeneration cycle) at certain regeneration temperature. For example the area under the line (Figure 47) should be equal in dehumidification and regeneration cycle. The fact is quite clear for both cases and multiple regeneration temperatures. However, for both cases it is interesting to note that the total water vapor desorption amount per minute (regeneration cycle) is significantly higher than the total adsorption per minute (dehumidification cycle). Therefore, the dehumidification cycle must be longer than the regeneration cycle (particularly in case-B) in order to establish a cost effective DAC system. It can also be concluded that the regeneration cycle can be even shrinked by employing higher regeneration temperature. The reason of this behaviour is due to typical adsorption kinetics phenomena of polymer based sorbents as reported by Sultan et al., 2016c. It can be simply explained that as the water vapor sorption kinetics is higher at higher regeneration temperatures resulting the lower cycle time for regeneration as compare to the dehumidification. This is also one of the fact which motivate the block based DAC as compare to rotary DAC, which provides the provision of variable dehumidification and regeneration cycle time. However in case of rotary DAC regeneration and dehumidification are made simultaneously (Sultan et al., 2016b). Similarly the proposed systems (see chapter 5) are useful for dehumidification at night time and regeneration at day time preferably by solar radiation. In addition, the amount of dehumidification presented by Figure 4.7 is useful for designing the process/return air flow rates and ultimately the regeneration temperature for particular agricultural product storage (see Table 5.1).



Figure 4.8 Effect of regeneration temperature on average dehumidification of desiccant unit.

From the discussion point of view, the average dehumidification ($\overline{\Delta X}$) performed by desiccant blocks under case-A and case-B at $T_{reg} = 40^{\circ}$ C-60°C is compared and presented by Figure 4.8. It can be seen (Figure 4.8) that higher $\overline{\Delta X}$ is resulted by case-B at all the regeneration temperatures as desiccant blocks accommodates huge amount of water vapor adsorption due to higher RH (case-B). It is obvious higher regeneration temperatures bring deep dehumidification though it might not be effective in terms of net COP of DAC system. Moreover, $\overline{\Delta X}$ increases with the increase in the regeneration temperature (40°C-60°C). On the other hand relative dehumidification difference decreases with the increase in the regeneration temperature from



Figure 4.9 Experimental performance evaluation of desiccant unit under case-C: (a) temperature and relative humidity profiles (a) resulted desorption and adsorption profiles.



Figure 4.10 Experimental performance evaluation of desiccant unit under case-D: (a) temperature and relative humidity profiles (a) resulted desorption and adsorption profiles.



Figure 4.11 Experimental performance evaluation of desiccant unit under case-E: (a) temperature and relative humidity profiles (a) resulted desorption and adsorption profiles.

 50° C - 60° C for both cases. This leads towards the optimum dehumidification performance of the desiccant blocks at low regeneration temperature ($\approx 50^{\circ}$ C). Therefore, the study is further extended by keeping the regeneration temperature constant (50° C) and varying the switching time ratios in order to optimize the cycle time as explained in coming paragraphs. The details of switching time are given in Table 4.5.

The experiments are further expanded for three different process air conditions (case-C, D, E) and for three switching time ratio (1:1, 2:3, 1:2) in order to optimize the switching time as explained in Table 4.5. The regeneration of desiccant blocks was made at 50°C and $X_{in} = 11$ g/kg-DA, respectively. The experimental results are repeated for two cycles (cycle-I and cycle-II) in order to analyze the experimental repeatability. Such repeatable performance of desiccant blocks is verified for all cases as shown by Figures 4.9-4.11. The corresponding dehumidification performance of desiccant unit is evaluated (Eq. 4.15) as shown in Figures 4.9(b)-4.11(b). Referring to Figures 4.9-4.11, it seems that desiccant blocks performed almost same dehumidification during both cycles (I and II). In addition, average dehumidification ($\overline{\Delta X}$) is calculated by Eq. (4.16) for both cycles and all cases. For example in case-C, the $\overline{\Delta X}$ of cycle-I and cycle-II is found as 2.114 g/kg-DA and 2.023 g/kg-DA, respectively (Figure 4.9). Cyclic performance is found similar or even better in other operating conditions as shown by Figure 4.10 and Figure 4.11. Therefore, it has been concluded that experimental setup successfully reproduces 95-98% of the numerical values in terms of air temperature and humidity.

Table	4.5	Experiments	under	varying	switching	time	ratios	and	constant	regeneration
temper	ature									

	Pro	cess air	Regeneration air		Switching time [min]			
Case	T _{in} [°C]	X _{in} [g/kg-DA]	T _{in} [°C]	X _{in} [g/kg-DA]	Reg. = 60 Deh. = 60 (1:1)	Reg. = 60 Deh. = 90 (2:3)	Reg.= 60 Deh.= 120 (1:2)	
С	20				0	0	0	
D	25	6	50	11	0	0	0	
Е	28				0	0	0	



Figure 4.12 Effect of switching time ratios on net dehumidification.



Figure 4.13 Temporal variation in wet-bulb and dew-point temperature depression.



Figure 4.14 The resulted dehumidification comparison at different switching time ratios.

It is determined that the net adsorption (ΔX) per kg of air decreases with elapsed cycle time as shown in Figure 4.12. Higher ΔX was found for process air condition of case-C as compare to case-D and case-E due to possessing higher relative humidity (Figures 4.9 - 4.11). Therefore, higher wet-bulb $(T_{wb,out} - T_{wb,in})$ and dew-point temperature $(T_{dp,out} - T_{dp,in})$ differences were found for case-C throughout the cycle time as compare to case-E as shown in Figure 4.13. The sequential decrement of ΔX remained same for all process air condition and $\overline{\Delta X}$ decreases with the increase in switching time (Figure 4.12). The $\overline{\Delta X}$ decrement for all cases by increasing switching time is obvious. Therefore $\overline{\Delta X}$ is further compared with average effective dehumidification ($\overline{\Delta X}$) which is expressed by Eq. (4.17). It is interesting to note that $\overline{\Delta X}$ slightly increases for case-C and case-D by changing the switching time ratio from 1:1 to 2:3 and again decreases at 1:2 (Figure 4.14). Moreover, it keeps decreasing with increase in switching time ratio for case-E due to arising of equilibrium conditions. It is envisaged therefore that selection of switching time mainly depends on required level of ambient air dehumidification. In addition, it may also depend on certain AC applications, desiccant material, ambient and operating conditions, and the availability of energy source. Similarly, according to a recent study (Sultan, 2015) of activated carbon based DAC system for agricultural greenhouse AC, the longer switching time presents optimum operating condition. On the other hand, it may not guarantee the higher overall system performance. Therefore, it has been concluded from bunch of experiments that the switching time ratio of 1:2, 2:3 and 1:1 can be selected for the operation of DAC system for high, medium and low humidity operating conditions, respectively.

4.4.2 Development of Correlation

On the basis of above discussion, it is ascertained that dehumidified air conditions are crucial for the performance evaluation of desiccant unit. Once the air is passed through the desiccant unit the absolute amount of water vapors existing in the air reduces due to the water vapor adsorption onto desiccant blocks. Ultimately absolute humidity as well as relative humidity of the air at desiccant exist end decreases. On the other hand, temperature of dehumidified air increases due to water vapor adsorption heat and/or heat of adsorption. In case of ideal dehumidification process all the adsorption heat is produced due to the water vapors from the vapor phase adsorbed onto desiccant in condensed form. So, the net adsorption heat in this case should be equivalent to the heat of condensation per unit mass multiply by mass of water vapor adsorbed. If an adsorbent follows this kind of ideal adsorption phenomena then the dehumidification process line of the adsorbent on psychrometric chart should follow the isenthalpic behavior. On the other hand, it has been practically observed that the net adsorption heat is well above the ideal adsorption heat and consequently it has been reported by various studies (Sultan et al., 2017). The relationship of this kind of behavior can be expressed by the following relationship (Eq. 4.20) as described in the literature (Sultan et al., 2014).

$$\phi_{deh} = \frac{\delta}{Q_{st}} \phi_h \tag{4.20}$$

where ϕ_h and ϕ_{deh} represent the slope of enthalpy line and desiccant dehumidification line on the psychrometric chart, respectively. The δ and Q_{st} represent the heat of water vapor condensation [kJ/kg] and isosteric adsorption heat [kJ/kg], respectively. In ideal dehumidification process the Q_{st} remains zero and consequently ϕ_h follow the isenthalpic trend. However, the desiccant dehumidification process usually does not exhibit isenthalpic behaviour. The difference between ϕ_{deh} and ϕ_h depend upon the adsorbent/adsorbate pairs' interactions and nature of adsorption e.g. monolayer/multilayer etc.

In addition to the complex behavior of adsorption heat discussed above, the desiccant performance is extremely influence by the dynamic/instantaneous water vapor pressure deficit as well as thermal stresses, and it has been reported in various studies (Sultan et al., 2015, Sultan et al., 2017). Therefore, it is really complex to simulate the performance of desiccant due to extreme variation encountered conditions of process and regeneration air streams. Consequently, lot of techniques has been established for the performance evaluation of desiccant unit which can be characterized as steady-state and dynamic operations. For example, in case of steady-state the techniques/models developed by: Beccali et al., 2003; Jurinak, 1982; Koronaki et al., 2012; Panaras et al., 2010; Sultan et al., 2016b are important from subject point of view. In case of dynamics operations desiccant models presented by: Miyazaki et al. 2009, Sultan et al. 2017 are important from the view point of research conducted in our lab. In this regard this thesis mainly considers the steady-state behavior of adsorption for the performance evaluation of desiccant block. The current study explores the various scenarios of steady-state adsorption and figure out the possible options for the development of a novel and simplified correlation for desiccant



Figure 4.15 Experimental investigation of slope of dehumidification.



Figure 4.16 Experimental investigation of dehumidification slope at varying T_{reg}.

performance evaluation. It is important to mention here that the analyses conducted in the coming chapter (chapter 5) will be solely based on this novel and simplified correlation. The correlation was basically based on the conception of modification of isenthalpic slope of dehumidification line on psychrometric chart. Use of this conception is limited to polymeric sorbent (4.2.1) irrespective of regeneration temperatures.

On the basis of above explained conception, the slope of dehumidification line is experimentally determined for the studied polymer desiccant blocks. A simplified relationship is developed for determining the ϕ_{deh} as described by Eqs. (4.21)-(4.24)

$$\phi_{deh} = \frac{T_{out} - T_{in}}{X_{in} - X_{out}} \tag{4.21}$$

$$\phi_{deh} = \frac{T_{out} - T_{in}}{\left(\frac{h_{in} - C_p T_{in}}{h_w + C_{pv} T_{in}}\right) - \left(\frac{h_{out} - C_p T_{out}}{h_w + C_{pv} T_{out}}\right)}$$
(4.22)

$$\phi_{deh} = \frac{(T_{out} - T_{in})(h_w + C_{pv}T_{in})(h_w + C_{pv}T_{out})}{\left[(h_{in} - C_p T_{in})(h_w + C_{pv}T_{out}) - (h_{out} - C_p T_{out})(h_w + C_{pv}T_{in}) \right]}$$
(4.23)

$$\phi_{deh} = \frac{(T_{out} - T_{in}) [h_w^2 + C_{pv}^2 T_{in} T_{out} + h_w C_{pv} (T_{in} + T_{out})]}{[(h_{in} - C_p T_{in}) (h_w + C_{pv} T_{out}) - (h_{out} - C_p T_{out}) (h_w + C_{pv} T_{in})]}$$
(4.24)

where ϕ_{deh} represents the slope of real/experimental dehumidification line [-]. T_{in} , T_{out} and X_{in} , X_{out} represent the air temperature [°C] and humidity ratio [kg/kg-DA], respectively at the inlet and outlet of desiccant blocks (see Figure 4.2). h_{in} and h_{out} are the corresponding air enthalpy [kJ/kg-DA]. C_p and C_{pv} denote the specific heat of dry air [kJ/kg·K] and water vapor [kJ/kg·K], respectively whereas h_w represents the evaporation heat of water [kJ/kg].

The experimental results of ϕ_{deh} for varying process air conditions (case-C,D,E) at different switching time ratios are presented in Figure 4.15. Moreover, in order to realize the effects of increasing regeneration temperature on ϕ_{deh} the experiments are further conducted for process air conditions of case-C at $T_{reg} = 60^{\circ}$ C. The results of ϕ_{deh} are therefore compared at both the regeneration temperature (50°C and 60°C) as shown in Figure 4.16. It is ascertained that the ϕ_{deh} does not vary largely for about 3600 sec. Therefore its experimental value ($\phi_{deh} =$ 0.31) can be set as constant for analyzing dehumidification performance of desiccant unit (4.2.1). Moreover, it can be concluded that variation in regeneration temperature does not vary the ϕ_{deh} as evident from Figure 4.16.

4.5 Conclusions

An open-cycle experimental apparatus was setup for the performance evaluation of hydrophilic polymer based desiccant blocks. Series of experiments are conducted for various: ambient air conditions, regeneration conditions, cycle time, and switching time. Generalized root sum of squares method is used to calculate the experimental uncertainty, and experimental data can be reproduced within 2-3% error. It is examined that when humid ambient air passes through the desiccant unit deep dehumidification occurs. It shows huge potential of air dehumidification at low regeneration temperature (50-60°C). It is determined that under the relatively dry ambient air conditions equivalent heat of adsorption (q_{eq}) profile approaches to net value equals zero. The average effective dehumidification slightly increases under humid ambient air conditions by changing the switching time ratio from 1:1 to 2:3 due to higher process air relative humidity. However, it keeps decreasing with increase in switching time ratio for relatively dry ambient conditions. It has been found that the switching time depends on dehumidification amount, nature of application and operating conditions. From the bunch of experiments, it has been concluded that the switching time ratio of 1:2, 2:3 and 1:1 can be selected for the operation of

DAC system for high, medium and low humidity operating conditions, respectively. The desiccant dehumidification process should follow isenthalpic line on psychrometric chart in an ideal scenario when there is no adsorption heat. The slope of dehumidification line on psychrometric chart is therefore modified for the realization of real desiccant dehumidification process based on experimental data. Consequently a simplified correlation is developed by which real desiccant dehumidification process can be predicted on psychrometric chart for polymeric desiccant.

4.6 Nomenclature

$\overline{\Delta X}$	average dehumidification [g/kg-DA or kg/kg-DA]
$\overline{\Delta X}$	effective average dehumidification [g/kg-DA or kg/kg-DA]
P _{atm}	atmospheric air pressure [Pa or kPa]
P_{v}	partial pressure of water vapor [Pa or kPa]
P_{vs}	saturation pressure of water vapor [Pa or kPa],
Q_{st}	heat of adsorption [kJ/kg]
R_{DA}	specific gas constant of dry air [J/kg·K or kJ/kg·K]
R_{v}	specific gas constant of water vapors [J/kg·K or kJ/kg·K]
\overline{n}	number of experimental entries during dehumidification
$ar{n}$	number of experimental entries of regeneration and dehumidification
<i>q_{eq}</i>	equivalent heat of adsorption [W]
ϕ_h	slope of enthalpy line
ϕ_{deh}	slope of dehumidification line
Δ	difference (in - out)
ΔX	net water vapors adsorption/desorption amount [g/kg-DA or kg/kg-DA]
А	area of orifice [m ²]
AC	air conditioning
C _d	discharge coefficient [-]
COP	coefficient of performance [-]
DA	dry air
DAC	desiccant air-conditioning

DBU	desiccant block unit
FCU	flow control unit
h	air enthalpy [kJ/kg-DA or J/kg-DA]
HPBS	hydrophilic organic polymer based sorbent
HX	heat exchanger
PA	process air
PAU	process air unit
RA	regeneration air
RAU	regeneration air unit
RH	relative humidity [%]
SHU	supplementary heating unit
Т	temperature [°C or K]
Χ	humidity ratio [g/kg-DA or kg/kg-DA]
Р	air pressure [Pa or kPa]
RH	relative humidity [% or -]
т	air mass flow rate [kg/sec]
δ	heat of condensation of water [kJ/kg]
ρ	air density [kg/m ³]

Subscript

ac	accuracy
DA	dry air
PA	process air
RA	regeneration air

4.7 References

- ASHRAE, 2009. Handbook-Fundamentals. American Society of Heating, Refrigerating and Airconditioning Engineers, Inc., Atlanta.
- Beccali, M., Butera, F., Guanella, R., Adhikari, R.S., 2003. Simplified models for the performance evaluation of desiccant wheel dehumidification. Int. J. Energy Res. 27, 17-29.
- Esen, H., Inalli, M., Esen, M., 2006. Technoeconomic appraisal of a ground source heat pump system for a heating season in eastern turkey. Energy Convers. Manag. 47, 1281-1297.
- Hepbasli, A., Akdemir, O., 2004. Energy and exergy analysis of a ground source (geothermal) heat pump system. Energy Convers. Manag. 45, 737-753.
- Holman, J.P., 2012. Experimental Methods for Engineers, eighth ed. McGraw-Hill, New York.
- Hyland, R.W., Wexler, A., 1983. Formulations for the thermodynamic properties of the saturated phases of H2O from 173.15K to 473.15K. ASHRAE Transactions 89, 500-519.
- Jurinak J.J., 1982. Open-cycle desiccant cooling component models and system simulation. Ph.D. thesis. Univ. of Wisconsin, Madison, USA.
- Koronaki, I.P., Rogdakis, E., Kakatsiou, T., 2012. Thermodynamic analysis of an open-cycle solid desiccant cooling system using artificial neural network. Energy Convers. Manag. 60, 152-160.
- Lee, S.H., Lee, W.L., 2013. Site verification and modeling of desiccant-based system as an alternative to conventional air-conditioning systems for wet markets. Energy. 55, 1076-1083.
- Mahmood, M.H., Sultan, M., Miyazaki, T., Koyama, S., 2016b. Theoretical and experimental analysis of desiccant air conditioning system for storage of agricultural products. 16th International refrigeration and air conditioning conference at Purdue, July 11-14, 2016. Purdue University, USA.
- Miyazaki, T., Akisawa, A., Shindoh, S., Mori, H., 2009. Development of a ventilation and airconditioning system using fixed bed desiccant units. Trans. JSRAE. 26, 521-532.
- Panaras, G., Mathioulakis, E., Belessiotis, V., Kyriakis, N., 2010. Experimental validation of a simplified approach for a desiccant wheel model. Energy Build. 42, 1719-1725.
- Stull, R., 2011. Wet-bulb temperature from relative humidity and air temperature. J. Appl. Meteorol. Climatol. 50, 2267-2269.

- Sultan, M., El-Sharkawy, I.I, Miyazak, T., Koyama, S., Saha, B.B., 2014. Experimental study on carbon based adsorbents for greenhouse dehumidification. Evergr. Jt. J. Nov. Carbon Resour. Sci. Green Asia Strategy. 1, 5-11.
- Sultan, 2015. Study on sorption characteristics of water adsorbents for agricultural airconditioning systems. D.Eng. thesis. Kyushu University, Japan.
- Sultan, M., El-Sharkawy, I.I., Miyazaki, T., Saha, B.B., Koyama, S., Maruyama, T., Maeda, S. and Nakamura, T., 2015. Insights of water vapor sorption onto polymer based sorbents. Adsorption, 21, 205-215.
- Sultan, M., Mahmood, M.H., Miyazaki, T., Koyama S., Khan, Z.M., 2016a. Close and open cycle adsorption kinetics: development of correlation for desiccant air-conditioning. J. Engg. Appl. Sci. 35, 1-8. (ISSN 1023-862X)
- Sultan, M., Miyazaki, T., Saha, B. B., Koyama, S., 2016b. Steady-state investigation of water vapor adsorption for thermally driven adsorption based greenhouse air-conditioning system. Renew. Energy. 86, 785-795.
- Sultan, M., El-Sharkawy, I.I., Miyazaki, T., Saha, B.B., Koyama, S., Maruyama, T., Maeda, S. and Nakamura, T., 2016c. Water vapor sorption kinetics of polymer based sorbents: Theory and experiments. Appli. Therm. Eng. 106, 192-202.
- Sultan, M., Miyazaki, T., Koyama, S., Khan, Z. M., 2017. Performance evaluation of hydrophilic organic polymer sorbents for desiccant air-conditioning applications. Adsorption Sci. Technol. 1-16. (DOI: 10.1177/0263617417692338)
- Uçkan, İ., Yılmaz, T., Hürdoğan, E., Büyükalaca, O., 2013. Experimental investigation of a novel configuration of desiccant based evaporative air conditioning system. Energy Convers. Manag. 65, 606-615.
- Yoshida, K., 2014. Study on application of desiccant air conditioning system to agriculture sector. M.Eng. thesis. Kyushu University, Japan.

CHAPTER 5

Optimization of Desiccant Air-Conditioning Systems for Storage of Food Products
CHAPTER 5

Optimization of Desiccant Air-Conditioning Systems for Storage of Food Products

This chapter presents the development of ideal storage zones of agricultural products, dried fruits, dried foods & feeds in comparison with greenhouse growth and humans thermal comfort zones. Six different configurations of DAC systems are proposed and their performance evaluation is made under the ambient conditions of Fukuoka-Japan. A simplified methodology is developed for performance evaluation of proposed DAC systems under varying regeneration temperatures. The parametric and thermodynamic analysis of all the system configurations (S-I to S-VI) is made to investigate that which configurations could yield better system performance. The psychrometric evaluation of three optimized DAC systems is performed for the agricultural storage and parallel applications.

5.1 Introduction

The postharvest shelf/storage life of agricultural products can be extended through cold storage, drying, chemical additives and heat treatments etc. The cold storage is considered most adopted physical treatment to combat postharvest losses and ultimately to extend the product's storage life (Usall et al., 2016). The most of the cold storages available today are equipped with conventional vapor compression refrigeration machines. Such

systems are not only degrading the environment and consuming high primary energy but also cannot provide the optimal storage conditions (temperature and relative humidity) particularly to the tropical agricultural products and dried fruits (Mahmood et al., 2016; Sultan and Miyazaki, 2017). It is because of diversified nature of agricultural products, chilling injury, and complex mechanism of transpiration, respiration, fermentation etc. In this perspective, low cost environmental friendly evaporative cooling technologies like direct evaporative cooling (DEC), indirect evaporative cooling (IEC)/M-Cycle evaporative cooling (MEC) have shown potential to provide storage conditions for particular agricultural products (Lal-Basediya et al., 2013; Sultan and Miyazaki, 2017). However, these technologies cannot be used effectively for the storage of agricultural products under largely varying ambient conditions (particularly humid) due to limited cooling performance (Lal-Basediya et al., 2013; Sultan and Miyazaki, 2017). The scope of DEC and IEC/MEC for the storage of wide range of agricultural products under varying environmental condition can be extended by the integration of desiccant dehumidification. The desiccant air-conditioning (DAC) comprises of desiccant dehumidification cum evaporative cooling has ability to deal the latent and sensible load of air-conditioning distinctly. Such ability of the DAC system makes it promising to provide optimal conditions for the storage of agricultural products without affecting their quantitative, qualitative and nutritive attributes. The approach adopted for storage of food products through DAC system can be described by self-explanatory Figure 5.1.

The present chapter experimentally investigates the combined effect of desiccant dehumidification and sensible heat exchange (via heat exchanger and/or M-Cycle evaporative cooling) for the storage of agricultural products. Six different configurations of DAC systems are proposed and their performance evaluation is made under the ambient conditions of Fukuoka-Japan. The ideal storage zones of agricultural products, dried fruits, dried foods & feeds are developed and compared with greenhouse growth and humans thermal comfort zones. Finally the psychrometric evaluation of three optimized DAC systems is made for food products storage and parallel applications.



Figure 5.1 Simplified scheme of adopted approach for storage of food products.

5.2 Development of Ideal Storage Zones

It is worthy to mention that more than one crop should not be stored in the same storage room except only for short period (< 1 week) or during transportation (Thompson, et al., 1996). Camelo, 2004, stated that sharing of same storage space/room can cause differences in storage temperature and relative humidity, ethylene and chilling sensitivity, odor contamination and promotion of other quality and shelf life effecting factors.

Moreover, the highly incompatible products should not be stored in the same storage room for more than one or two days. However, compatible products storage groups have been suggested in the literature in order to optimize the storage space. These groups are developed with the assumption that the ambient ethylene concentration does not exceed 1 ppm (Thompson et al., 1996). In this regard, Thompson et al., 1996, developed three different temperature (T) and relative humidity (RH) compatible groups of fruits and vegetable for short term (7 days) storage. Tan, 1996, however recommends five different compatible groups for mixed storage of agricultural products. McGregor, 1989, on the other hand, suggested the seven different T and RH compatible groups of fruits and vegetables.

Table 5.1 Compatible storage groups (I, II and III) of the agricultural products (ASHRAE, 2010; Kitinoja and Kader, 2002).

Groups*	T [°C]	RH [%]	Products**
Ι	10	85-90	Cucumber, Eggplant, Okra (ladyfinger), Olive, Pepper, Potatoes-storage, Squash- summer
Π	13-15	85-90	Avocado, Banana, Coconut, Ginger root, Grapefruit, Guava, Jackfruit, Lemon, Lime, Mango, Melon, Papaya, Potato-new, Pineapple, Pumpkin, Rambutan, Squash- winter, Tomatoes-ripe
III	18-21	85-90	Jicama, Sweetpotatoes, Tomatoes-mature green, Watermelon, White sapote, Yams

*Psychrometric representation of groups is given on Figure 5.2. **For detail, refer to ASHRAE (2010) and Kitinoja and Kader (2002)

It is ascertained on the basis of above discussion that the storage compatibility of agricultural products is crucial to maintain the quantitative and qualitative attributes. In this regard, three different compatible groups of postharvest agricultural products (fruits and vegetables) according to their T and RH are established for the development of ideal storage zones on psychrometric chart as shown in Figure 5.2. The details of groups including product type, temperature and relative humidity are shown in Table 5.1 (ASHRAE, 2010; Kitinoja and Kader, 2002). Only the few products are presented in Table



Figure 5.2 Psychrometric comparison among agricultural products, dried fruits, dried foods/feed storage, greenhouse AC and humans' AC.

Table 5.2 Recommended moisture contents of dried fruits for safe storage, shipping and buying/consumption (Dauthy, 1995).

Dried fruits	Dried forms	Moisture [%]
Apricots	Caps	17-20
Prunes	Whole	18-20
Figs	Whole	21-25
Raisins	Whole	15-19

5.1, however the details about other agricultural product and storage life can be found from the cited references. The group I and II are selected because of their sensitivity to the chilling injury, whereas group III is selected due to the requirement of slightly warmth conditions.

As far as the storage of dried agricultural products is concerned, the recommended moisture contents contained by the dried fruits (apricots, prunes, figs and raisins) with additions of $\pm 2\%$ in figs and raisins are given in Table 5.2 (Dauthy, 1995). Such moisture content percentage is needed for their safe storage, shipping and buying/consumption by the end users (Dauthy, 1995). In case the moisture contents of the dried fruits increases than the recommended value then chances of fungi attack will be higher. On the other hand, if the moisture contents decreases then the weight of the dried fruits will decrease tremendously and also the products will become so hard which ultimately affect the taste of the product (Pahlevanzadeh and Yazdani, 2005). In order to maintain the equilibrium moisture content of the dried fruits, the ideal T and RH zone is determined using reference (Maroulis et al., 1988) along with the Guggenheim (1966), Anderson (1946) and De-Boer (1953) (GAB) model. The GAB model can be simply represented by Eq. (5.1). It has been well described in the literature for various applications e.g. food products storage (Akanbi et al., 2006; Wolf et al., 1984) and polymer based DAC (Sultan et al., 2015) etc. It is mostly employed for analysis of moisture sorption in fruits, vegetables and meats (Andrade et al., 2011). The GAB model as improved form of Brunauer-Emmett-Teller (BET) theory (Brunauer et al., 1938) which possesses an additional constant K. All of the constants (M_m, C, K) of GAB model are function of adsorption temperature and have physical meaning as well. The constants along with other parameters are explained by Eqs. 5.2-5.6.

$$M = \frac{M_m a_w CK}{(1 - Ka_w)(1 - Ka_w + CKa_w)}$$
(5.1)

$$M_m = M_0 \exp\left(\frac{l_m}{R T}\right) \tag{5.2}$$

$$C = C_{\rm o} \exp\left(\frac{\Delta H_c}{R T}\right) \tag{5.3}$$

$$K = K_{\rm o} \exp\left(\frac{\Delta H_k}{R T}\right) \tag{5.4}$$

$$\Delta H_c = H_{\rm m} - H_{\rm n} \tag{5.5}$$

$$\Delta H_k = \omega_{\rm w} - H_{\rm n} \tag{5.6}$$

where *M* is the moisture content of the material on dry basis [kgH₂O/kg-dry solids]; $M_{\rm m}$ is moisture contents on dry basis corresponding to an absorbed monolayer [kg/kg]; a_w is the water activity; *C* and *K* are constants related to the temperature effect, M_0 , $l_{\rm m}$, C_0 , and K_0 , are adjusted constants for the temperature effect, $\Delta H_{\rm C}$ and $\Delta H_{\rm k}$ are functions of the heat of sorption of water, $H_{\rm m}$ and $H_{\rm n}$ represent the heat of sorption of monolayer and multilayer respectively, $\omega_{\rm w}$ is heat of condensation water vapor [kJ/kg], *T* is absolute temperature [K] and *R* is gas constant [kJ/kg·K] (Maroulis et al., 1988; Sultan et al., 2015).

It is important to mention that GAB equation becomes BET equation when constant K equals to unity (i.e. $C_{BET} = C_{GAB} K$) (Sultan et al., 2015). The direct and indirect methods are used in order to estimate the constants of GAB model. In indirect method the three constants of GAB model (M_m , C, K) are preliminary estimated at each temperature and afterwards all the other constants (M_o , l_m , C_o , K_o , H_C and ΔH_k) are estimated. Such indirect method of estimating the constants is not preferred in literature. It is because the reliability of this method is based on confidence limit and regions of preliminary estimated constant (M_m , C, K) and are interconnected with each other (Maroulis et al., 1988, Sultan et al., 2015). On the other hand, in direct method all GAB constant are estimated by the algebraic manipulation of Eqs. 5.1-5.6.

Though the relative humidity for storage of dried fruits at temperature 15-25°C varies for each product (apricots, prunes, figs, raisins), however it falls between 50-70%. Therefore, the representative ideal storage zone for dried fruits is developed at temperature and relative humidity of 15-25°C and 50-70%, respectively, as shown in Figure 5.2. On the other hand, as per recommendations laid in ASHRAE, 2010, the bulk storage of dried foods and feeds (like dehydrated milk and alfalfa meal) require temperature and relative humidity of about 21-32°C and 40-60%, respectively, as shown in Figure 5.2. Their (dried

foods/feeds) storage life varies between three months to one year depending upon the provision of temperature and relative humidity conditions (ASHRAE, 2010). However, storage of dried fruits at 21°C enables one year life. Moreover, the further decrease in temperature results in higher storage life. But, lowering of temperature must be managed that it will not affect the recommended moisture contents of dried fruits. Otherwise, increasing and decreasing moisture contents can result in loss of quality (fungi attack) and quantity (weight), respectively. In this regard, the approximate shelf life (Γ) of food at fluctuating temperatures can be evaluated using Eq. (5.7) and Eq. (5.8) (Meurant, 2012; Toledo, 2007; Singh, 1994).

$$\Gamma_2 = \Gamma_1 \exp\left(-\theta(T_2 - T_1)\right) \tag{5.7}$$

$$\theta = \frac{\ln Q_{10}}{10} \tag{5.8}$$

where Γ_1 represents product known shelf life at reference temperature (T_1) and Γ_2 represents the expected product shelf life at target temperature (T_2) . The factor Q_{10} is temperature coefficient. In case of reactions like nonenzymatic browning (Samaniego-Esguerra, 1989) the value of Q_{10} can be taken as 2 (Toledo, 2007), therefore same value is considered for shelf life evaluation of dried fruits. The equations (5.7 and 5.8) are solved by taking the dried fruits shelf life as 12 months at reference temperature of 21°C (Dauthy, 1995). It was found that Γ increases with decreasing storage temperature as shown in Figure 5.3. It becomes double (24 months) with 10°C reduction in temperature (i.e. 11°C).

The ideal storage zones of agricultural products (Table 5.1), dried fruits (apricots, prunes, figs, raisins), dried foods & feeds (ASHRAE, 2010) are also compared with greenhouse growth (Sultan, 2015, Sultan et al., 2016) and humans thermal comfort zones (ASHRAE, 2009) as shown in Figure 5.2. It can be noticed that low temperature and high relative humidity is required for storage of agricultural products as compare to humans thermal comfort zone.



Figure 5.3 Effects of temperature on storage life of dried fruits.

5.3 Description of Proposed Systems

On the basis of desiccant block dehumidification performance analysis (chapter 4) the study is further extended by considering the incorporation of these blocks into desiccant air-conditioning (DAC) system. In this regard six different configurations of DAC systems are proposed. These system configurations are named here as System-I (S-I), System-II (S-II), System-II (S-II), System-IV (S-IV), System-V (S-V) and System-VI (S-VI). Their schematics along with psychrometric representation are shown in Figures 5.4(a)-(f).

The S-I consists of two desiccant unit, a heat exchanger (HX), a M-Cycle evaporative cooler (MEC), a direct evaporative cooler (DEC) and heat source. It is important to mention that each proposed system configuration commonly contains heat source and two desiccant units. Two units of desiccant blocks (DB) are used to enable their switching during regeneration and dehumidification. However, the other components are used alternatively as shown in the respective system configurations (Figure 5.4). It is worthy to mention that system configurations (S-I to S-IV) employ the MEC to achieve the



Figure 5.4 Schematics of proposed DAC systems along with psychrometric representation (a)-(f) represents system (I)-(VI), respectively.



Figure 5.4 Schematics of proposed DAC systems along with psychrometric representation (a)-(f) represents system (I)-(VI), respectively.

sensible load of AC. In contrast with conventional indirect evaporative cooling methods, the MEC utilizes novel dew-point evaporative cooling conception via M-Cycle (Mahmood et al., 2016). Therefore, MEC has an ability to cool the air below its wet-bulb temperature and approaches to the corresponding dew-point. The fundamental details and principles of M-Cycle operations are explained in chapter 3. Referring to Figure 5.4(a), the working principle of the proposed system is as follow: ambient/process air (1) is dehumidified by desiccant blocks that exits at condition (2). The dehumidified air (2) is initially cooled by HX followed by MEC which exits at condition (3) and (4) respectively. The supply air (SA) at condition (4) can be used for the relevant application. For the cyclic usage the desiccant blocks are required to be regenerated. Therefore, the working air at condition (5/1) initially passes through the DEC (6) followed by HX (7) in order to recovers the heat of adsorption of the dehumidified process air. The warm working air at condition (7) is further heated by the heat source. The hot working air (8) at regeneration temperature (T_{reg}) 50-80°C passes through the desiccant blocks for their regeneration. Finally, the regeneration air (9) is exhausted to the atmosphere. The methods used for the analyses of all the system configurations are explained in next heading.

5.4 Methods and Analyses

The phenomena of desiccant block dehumidification have been explained in detail in chapter 4 wherein a simple correlation is developed, on the basis of series of experiments, in order to find out slope of dehumidification line (ϕ_{deh}) (Eqs. 4.21-4.23). The detailed methodology adopted for the analysis of proposed systems is shown in Figure 5.5. When the ambient and/or return air pass through the desiccant unit it become dehumidified consequently its temperature increases and relative humidity decreases. In this regard, it is worthy to mention that dehumidification (latent load) performance of desiccant blocks in all system configurations is evaluated by the control strategy utilizing experimentally validated ϕ_{deh} (chapter 4). Moreover, the low temperature regeneration (50-80°C) of desiccant blocks was made as shown in Figure 5.5. On the other hand, in order to achieve the sensible load of air-conditioning the dehumidified air at optimized conditions (T and RH) is passed either through HX and/or MEC depending upon the system configuration (Figure 5.4). The conditioned supply air (SA) temperature by MEC is determined through simplified correlation (Eq. 5.9) validated by published experimental data (Sultan and Miyazaki, 2017). However, the rest of air conditions are determined by fundamental thermodynamic expressions (Eqs. 5.10-5.14) according to of each system configuration. Therefore, it is worthy to mention that the system process air analyses are determined by experimentally validated correlations however the regeneration/working air analyses are made by fundamental thermodynamic expressions.

$$T_{SA/(MEC,o)PA} = 6.70 + 0.2630 T_{(MEC,i)PA} + 0.5298 X_{(MEC,i)PA}$$
(5.9)

$$T_{(HX,o)PA} = T_{(DB,o)PA} - \varepsilon_{HX} \left[T_{(DB,o)PA} - T_{(HX,i)RA} \right]$$
(5.10)

$$T_{(DEC,o)RA} = T_{(DEC,i)RA} - \varepsilon_{DEC} \left[T_{(DEC,i)RA} - T_{(DEC,i,wb)RA} \right]$$
(5.11)

$$T_{(HX,o)RA} = T_{(HX,i)RA} + \varepsilon_{HX} \left[T_{(DB,o)PA} - T_{(HX,i)RA} \right]$$
(5.12)

$$Q_{reg} = m_{RA}C_p [T_{(HS,o)RA} - T_{(HS,i)RA}]$$
(5.13)

$$COP = \left[\frac{h_{(DB,i)PA} - h_{(MEC,o)PA}}{h_{(HS,o)RA} - h_{(HS,i)RA}}\right]$$
(5.14)

where *COP*, *T* and *X* represent the thermal coefficient of performance [-], air temperature [°C] and humidity ratio [g/kg-DA], respectively. Subscript *i*, *o* represent process (*PA*) and regeneration air (*RA*) conditions at the inlet (*i*)/outlet (*o*) of desiccant block (*DB*), direct evaporative cooler (*DEC*), heat source (*HS*), heat exchanger (*HX*), M-Cycle evaporative cooler (*MEC*). ε , *h* and subscript *wb* represent the effectiveness [-], enthalpy [J/kg-DA] and wet-bulb, respectively. *Q_{reg}* represents the heat input [kW] through heat sources for regeneration of desiccant blocks. *m* and *C_p* represent the air mass flow rate [kg/sec] and specific heat capacity [kJ/kg·K], respectively.



Figure 5.5 Schematics elaborated the methodology adopted for the analysis of proposed systems.



Figure 5.6 Ambient temperature and relative humidity profiles of Fukuoka, Japan.



Figure 5.7 Profiles of ambient temperature and relative humidity for the month of July.

The performance evaluation of the proposed systems is carried out for the ambient conditions of Fukuoka, Japan as explained above. In this regard the data of ambient conditions (T, RH, Solar radiation) for the month of July is obtained from "Meteonorm 7"; licensed version software developed by Meteotest, Switzerland. The month of July is selected because it is one of hot and humid month throughout the year as shown in Figure 5.6. However, the daily T and RH profile for the month of July are shown in Figure 5.7. The data of T and RH (Figure 5.6 and Figure 5.7) is based on ten years (2000-2009) records; however the solar radiation data is based on twenty years (1991-2010) records.

The supply air temperature and relative humidity are determined as explained above. The average supply air temperature (T_{SA}) and relative humidity (RH_{SA}) are then determined for the month of July. The net and average dehumidification (ΔX and $\overline{\Delta X}$) is determined by Eq. 4.15 and 4.16, respectively as explained in chapter 4. However, in calculating the $\overline{\Delta X}$ the parameter \overline{n} here represents the number of ambient conditions entries for the month of July. Similarly, the thermal COP of all DAC systems is calculated by the respective equation as depicted on their schematics (Figure 5.4). Moreover, wet-bulb (ε_{wb}) and dew-point (ε_{dp}) effectiveness of MEC integrated in DAC system is determined by Eq. 3.1 and Eq. 3.2 respectively, as elaborated in chapter 3. The heat input through heat sources for regeneration of desiccant blocks (Q_{reg}) during the month of July is calculated by fundamental thermodynamic expression (Eq. 5.13) (Mahmood et al. 2016). The results and discussion about the performance evaluation of proposed systems is explained in next heading.

5.5 **Results and Discussion**

5.5.1 Performance Evaluation of System-I and II

In system configurations (S-I and S-II), the supply air temperature decreases with increasing regeneration temperature. It is determined that the T_{SA} of S-I and S-II decreases from 20.01-17.93°C and 20.74-18.68°C, respectively by increasing the T_{reg} from 50°C to 80°C. The corresponding RH_{SA} also decreases as shown in Figure 5.8. Hence the cooling capacity of S-I is relatively slightly higher than S-II. It is due to the supply of cool working air, through DEC, to the HX of S-I during the regeneration of desiccant block. On the other hand, it is found that the desiccant blocks in system configuration of S-II enable comparatively higher net dehumidification than S-I at identical regeneration temperature as shown in Figure 5.9. It is because of configuration of S-II which permits the supply of working air at lower humidity ratio than S-I. Therefore, the average dehumidification of S-I and S-II at Treg of 50-80°C is determined as 4.26-8.94 g/kg-DA and 4.58-9.20 g/kg-DA, respectively. It is worthy to mention that S-II performs higher dehumidification but required less heat input for regeneration (Q_{reg}) desiccant blocks as compare to S-I (Figure 5.10). The Q_{reg} for S-II was determined as 688 kW and 1934 kW at regeneration temperature of 50°C and 80°C respectively. Whereas, it increases correspondingly for S-I as 785 kW and 2018 kW.

The thermal COP of S-II remained higher than S-I at $T_{reg} = 50-80^{\circ}$ C. However, it is found that the relative difference between thermal COP of S-I and S-II decreases with increasing regeneration temperature. Similarly, the corresponding Q_{reg} tends to decrease as shown in Figure 5.10.



Figure 5.8 Effect of regeneration temperature on supply air temperature and relative humidity of S-I and S-II.



Figure 5.9 Net dehumidification performance profiles of S-I and S-II under varying ambient conditions.



Figure 5.10 Effect of regeneration temperature on system's (S-I, S-II) thermal COP and heat input.



Figure 5.11 Dew-point and wet-bulb effectiveness comparison of MEC integrated in S-I and S-II.

The wet-bulb effectiveness (ε_{wb}) of MEC integrated in S-I and S-II was found almost same at $T_{reg} = 50^{\circ}$ C. Accordingly, the corresponding dew-point effectiveness (ε_{dp}) also remains same. However, both the effectiveness starts decreasing by increasing the regeneration temperature from 60°C to 80°C. Moreover, such decrement remained higher in S-I than S-II that means the ε_{wb} and ε_{dp} of S-II becomes higher than S-I at identical elevated T_{reg} (60-80°C) as shown in Figure 5.11.

On the basis of above prospective, it can be concluded that the system configuration (S-II) is more suitable due to the requirement of less regeneration heat input, higher dehumidification, higher wet-bulb and dew-point effectiveness, and almost same cooling capacity as of S-I. The S-II also does not involve the additional cost of DEC. Moreover, the supply of higher regeneration heat input in S-I incur variable cost.

5.5.2 Performance Evaluation of System-III and IV

In contrast to S-I and S-II, the supply air temperature increases with increasing regeneration temperature in S-III and S-IV. However, it is determined that T_{SA} (also RH_{SA}) remains same for both S-III and S-IV at identical T_{reg} as shown in Figure 5.12. It means that cooling capacity of both systems remains almost same. Therefore, such system configurations can be potentially operated at low regeneration temperature to achieve the less sensible load of AC. However, as far the latent load is concerned the dehumidification of desiccant blocks of S-III and S-IV increases with increasing regeneration temperature as shown in Figure 5.13. The dehumidification performance of S-III and S-IV was found consistent with S-I and S-II, respectively. It is due to systems' design which facilitates supply of working air at identical humidity ratio during the regeneration of desiccant blocks. However, the heat energy required for regeneration of desiccant blocks of S-III and S-IV is determined about two times greater than of S-I and S-II in order to get the same level of dehumidification. Such higher regeneration heat input is due to the elimination of heat exchanger in system configurations of S-III and S-IV. Therefore, the Q_{reg} for S-IV was calculated as 1683 kW and 3929 kW at regeneration temperature of 50°C and 80°C respectively. Whereas, it increases correspondingly for S-III as 1959 kW and 4205 kW. It is important to note that despite of utilizing less Q_{reg} (Figure 5.14), the S-IV enables higher dehumidification than S-III at same regeneration temperature (Figure 5.13).



Figure 5.12 Effect of regeneration temperature on supply air temperature and relative humidity of S-III and S-IV.



Figure 5.13 Net dehumidification performance profiles of S-III and S-IV under varying ambient conditions.



Figure 5.14 Effect of regeneration temperature on system's thermal COP and heat input.



Figure 5.15 Dew-point and wet-bulb effectiveness comparison of MEC integrated in S-I, S-II, S-III and S-IV.

Alike in S-I and S-II, the thermal COP of S-III and S-IV decreases with increasing regeneration temperature. However, the COP of S-III and S-IV was found much lesser than S-I and S-II due to higher Q_{reg} . Moreover, the COP of S-III remained higher than S-IV even at higher regeneration temperature (80°C) as shown in Figure. 5.14.

The wet-bulb effectiveness (ε_{wb}) of MEC integrated in S-III and S-IV was found almost same at the corresponding regeneration temperatures (50-80°C). Accordingly, the corresponding dew-point effectiveness (ε_{dp}) also remains same. However, both the effectiveness was found comparatively higher than S-I and S-II at elevated T_{reg} (60-80°C) as shown in Figure 5.15. Thus ε_{wb} and ε_{dp} of both systems (S-III and S-IV) is determined about 0.98 and 0.63, respectively at T_{reg} = 80°C.

It is ascertained from the above discussion that the system configuration (S-IV) is more suitable due to the requirement of less regeneration heat energy, higher dehumidification, almost same cooling capacity, and wet-bulb and dew-point effectiveness as of S-III. Alike in S-II, the S-IV also does not involve the additional cost of DEC. Moreover, the supply of higher regeneration heat in S-III incurs variable cost.

5.5.3 Performance Evaluation of System-V and VI

Alike in S-III and S-IV, the supply air temperature increases with increasing regeneration temperature in S-V and S-VI. However, in contrast to S-I and S-II, it is determined that T_{SA} of S-V and S-VI increases as 25.57-27.07°C and 29.0-30.47°C, respectively at $T_{reg} = 50-80$ °C as shown in Figure 5.16. Though the cooling capacity of S-V is higher than S-VI but much lesser than rest of the studied systems. Similar to other system configurations (S-I, S-II and S-III, S-IV) the S-VI performs relatively higher dehumidification than S-V. The average dehumidification ($\overline{\Delta X}$) of all the system configurations is determined at varying regeneration temperature in order to compare their dehumidification performance. It is ascertained that S-I, S-III and S-V enable similar $\overline{\Delta X}$ but higher than the S-II, S-IV and S-VI at corresponding regeneration temperature as shown in Figure 5.17. Moreover, $\overline{\Delta X}$ of all systems configuration also increases with increasing regeneration temperature.

The Q_{reg} required for regeneration of desiccant blocks of S-V and S-VI is determined about two times lesser than of S-III and S-IV in order to get the same level of



Figure 5.16 Effect of regeneration temperature on supply air temperature and relative humidity of S-V and S-VI.



Figure 5.17 Comparison of net dehumidification performance of all proposed systems.



Figure 5.18 Effect of regeneration temperature on system's (S-V, S-VI) thermal COP and heat input.

dehumidification. It means these systems (S-V and S-VI) required the same regeneration heat input as of S-I and S-II. The thermal COP of S-V and S-VI was found lower than S-I and S-II at same Q_{reg} and regeneration temperature (Figure 5.18). However, both the systems (S-V and S-VI) enable about two times higher COP than S-III and S-IV by utilizing two times lower Q_{reg} (see Figure 5.14 and Figure 5.18). Moreover, the COP of S-V in comparison to S-VI remained high even by increasing the regeneration temperature from 50°C to 80°C. It is ascertained that S-V can be better system configuration because of comparatively higher cooling capacity, RH_{SA} and COP than S-IV though it (S-V) require relatively higher Q_{reg} .

It is established on the basis of parametric and thermodynamic analysis of all the system configurations (S-I to S-VI) that S-II, S-IV and S-V could yield better system performance. Therefore, these system configurations are only considered for further psychrometric analysis in order to find out the appropriate system/technology for the storage of agricultural products and other parallel applications as explained in next heading.

5.5.4 Storage Systems

The performance evaluation of studied system configurations (S-I to S-VI) revealed that S-II, S-IV, and S-V could yield better system performance. In this regard supply air conditions of these systems against the ambient conditions of Fukuoka, Japan are determined at regeneration temperature of 50-80°C as explained in heading 5.4. Accordingly, psychrometric evaluation of proposed systems (S-II, S-IV, and S-V) is made for insights about the storage of agricultural products (Table 5.1), dried fruits (apricots, prunes, figs, raisins) and dried foods & feeds (ASHRAE, 2010) in comparison with zone of greenhouse growth (Sultan, 2015; Sultan et al., 2016) and human thermal comfort (ASHRAE, 2009). The development of such zones is explained in heading 5.2.



Figure 5.19 Psychrometric evaluation of supply air conditions of S-II for different applications.



Figure 5.20 Psychrometric evaluation of supply air conditions of S-IV for different applications.

The temporal variation in supply air conditions of S-II at $T_{reg} = 50-80^{\circ}C$ is presented in Figure 5.19. The psychrometric evaluation revealed that the S-II cannot achieve the latent and sensible load of AC for the storage of agricultural products at $T_{reg} =$ 50-70°C under the varying ambient conditions of Fukuoka. The system (S-II) achieves the latent load at $T_{reg} = 80^{\circ}C$ (Figure 5.19), however the sensible load can be achieved by further cooling through DEC (i.e. two-stage cooling: MEC+DEC). As the agricultural products required high storage relative humidity and low temperature therefore the integration of DEC after the MEC in S-II configuration can provide the optimum conditions for the storage of agricultural products. Therefore, the configuration of S-II need to be modify as two-stage cooling (MEC+DEC) rather than single-stage cooling (MEC) particularly for the agricultural products storage. But such integration will nullify the cost benefit of S-II over S-I as described above. On the other hand, S-II can provide the storage conditions for the dried fruits at low regeneration temperature ($T_{reg} = 70^{\circ}C$) as shown in Figure 5.19. Moreover, the S-II can be effectively used for the provisions of greenhouse growth zone AC even at low regeneration temperature depending upon the type and growth stage of the crop. However, the implication of S-II for rest of the applications can be seen from Figure 5.19.



Figure 5.21 Psychrometric evaluation of supply air conditions of S-V for different applications.

As far as psychrometric evaluation of S-IV for the storage of agricultural products is concerned, it is examined that S-IV cannot be effectively used for the storage of agricultural products at $T_{reg} = 50-80^{\circ}$ C due to non-provision of optimum temperature and

relative humidity conditions as shown in Figure 5.20. On the other hand, S-IV almost achieves the latent load at $T_{reg} = 70^{\circ}$ C but cannot achieves the sensible load of AC for the storage of dried fruits. The sensible load can be achieved by two-stage sensible cooling (MEC + MEC/IEC) in contrast to two-stage cooling (MEC + DEC) of modified S-II. It is because of requirement of less storage relative humidity for dried fruits as compare to agricultural products. Therefore, the configuration of S-IV needs to be modified by integrating one more sensible cooler (MEC/IEC). However, storage of dried food and feeds can be made by S-IV at low regeneration temperature (60°C). Moreover, the S-IV can assist the conventional AC systems for the provision of human thermal comfort.

The supply air conditions of S-V are presented by Figure 5.21. It can be envisaged that T_{SA} and RH_{SA} of S-V are higher and lower respectively in comparison to S-II and S-IV. In this regard, the S-V can be particularly used for the applications which required relatively high temperature and low relative humidity. Therefore it can be potentially employed for the storage of dried food and feeds at low regeneration temperature (60°C). However, the S-V required about two times less regeneration heat input (Q_{reg}) as compare to S-IV.

5.6 Concluding Remarks

The ideal storage zones of agricultural products, dried fruits, dried foods & feeds in comparison with greenhouse growth and humans thermal comfort zones are established on the psychrometric chart. The study experimentally investigates the combined effect of desiccant dehumidification and sensible cooling (via MEC or HX) for the storage of food products. In this regard, six different configurations of DAC systems are proposed and their performance evaluation is made under the ambient conditions of Fukuoka-Japan. It is established on the basis of parametric and thermodynamic analysis of all the system configurations (S-I to S-VI) that S-II, S-IV and S-V could yield better system performance. On the basis of the psychrometric evaluation of three optimized DAC systems (S-II, S-IV and S-V) following itemized concluding remarks can be made:

• S-II cannot achieve the latent and sensible load of AC for the storage of agricultural products at $T_{reg} = 50-70^{\circ}$ C under the varying ambient conditions of Fukuoka.

- S-II achieves the latent load at $T_{reg} = 80^{\circ}C$, however the sensible load can be achieved by two-stage cooling (MEC+DEC). Therefore, the modification of two-stage cooling in is suggested in configuration of S-II.
- S-II can provide the optimum conditions for the storage of dried fruits at low regeneration temperature ($T_{reg} = 70^{\circ}$ C). Moreover, the S-II can be effectively used for the provisions of greenhouse growth zone AC even at low regeneration temperature depending upon the type and growth stage of the crop.
- S-IV cannot be effectively used for the storage of agricultural products at all (50-80°C) regeneration temperatures.
- S-IV almost achieves the latent load at T_{reg} = 70°C but sensible load for storage of dried fruits can be achieved by two-stage sensible cooling (MEC + MEC/IEC). Therefore two-stage sensible cooling modification is suggested in configuration of S-IV.
- S-IV can assist the conventional AC systems for the provision of human thermal comfort.
- S-V can be potentially employed for the storage of dried food and feeds regeneration temperature (60°C). Moreover, it required about two times less regeneration heat input (Q_{reg}) as compare to S-IV.

Finally it is concluded that thermally driven DAC systems could yield advance agricultural storage system which can control temperature and humidity distinctly irrespective to conventional compressor based AC systems. Therefore, integration of evaporative cooling unit(s) into DAC system will lead towards energy-efficient and reliable low-cost AC systems for various applications. However optimum operational conditions will need to be determined and regulated for particular application. Herein it has been concluded that one or other DAC system could be efficiently utilized for the storage of agricultural products.

5.7 Nomenclature

Cp	specific heat capacity of air [kJ/kg·K]
<i>M</i> _m	moisture contents corresponding to an absorbed monolayer [kg/kg]
Q_{10}	temperature coefficient [-]
Q_{reg}	regeneration heat input [kW]
a_w	water activity [-]
Γ_1	product known shelf life [months]
ω _w	heat of condensation water vapor [kJ/kg]
ϕ_{deh}	slope of dehumidification line
ΔX	net dehumidification [g/kg-DA or kg/kg-DA]
AC	air-conditioning
COP	coefficient of performance [-]
DAC	desiccant air-conditioning
DB	desiccant block
DEC	direct evaporative cooling
3	effectiveness [-]
h	air enthalpy [kJ/kg-DA or J/kg-DA]
HS	heat source
HX	heat exchanger
IEC	indirect evaporative cooling
M-Cycle	Maisotsenko Cycle
MEC	M-Cycle evaporative cooling
RH	relative humidity [% or -]
Т	temperature [°C or K]
Х	humidity ratio [g/kg-DA or kg/kg-DA]
Μ	moisture content of the material on dry basis [kgH ₂ O/kg-dry solids]
R	gas constant [kJ/kg·K]
Г	product shelf life [months]

Subscript

dp	dew-point
PA	process air
RA	regeneration air
reg	regeneration
SA	supply air
wb	wet-bulb

5.8 References

- Akanbi, C.T., Adeyemi, R.S., Ojo, A. 2006. Drying characteristics and sorption isotherm of tomato slices. J. Food Eng. 73, 157–163.
- Anderson, R.B., 1946. Modification of the Brunauer, Emmet and Teller equation. J. Am. Chem. Sco. 68, 686-691.
- Andrade, P., Ricardo D., Lemus M.R., Perez, C., 2011. Models of sorption isotherms for food: uses and limitations. Vitae. 183, 325-334.
- ASHRAE, 2009. Handbook-Fundamentals. American Society of Heating, Refrigerating and Air-conditioning Engineers, Inc., Atlanta.
- ASHRAE, 2010. Handbook-Refrigeration. American Society of Heating, Refrigerating and Air-conditioning Engineers, Inc., Atlanta.
- Brunauer, S., Emmett, P.H., Teller, E., 1938. Adsorption of gases in multimolecular layers.J. Am. Chem. Soc. 60, 309-319.
- Camelo, A.F.L., 2004. Manual for the preparation and sale of fruits and vegetables: from field to market. FAO agricultural Services Bulletin No. 151. Food and Agriculture Organization of the United Nations, Rome (Italy).
- Dauthy, M.E., 1995. Fruit and vegetable processing. FAO agricultural Services Bulletin No. 119. Food and Agriculture Organization of the United Nations, Rome (Italy). http://postharvest.ucdavis.edu/Commodity_Resources/Storage_Recommendations/C ompatibility_Chart_for_Short-term_Transport_or_Storage/

De Boer, J.H., 1953. The dynamic character of adsorption. Clarendon, Oxford.

Guggenheim, E.A., 1966. Applications of statistical mechanics. Clarendon, Oxford.

- Kitinoja, L., Kader, A.A., 2002. Small-scale postharvest handling practices: a manual for horticultural crops, fourth ed. Postharvest Technology Research and Information Center, University of California, Davis.
- Lal Basediya, A., Samuel, D.V.K., Beera, V., 2013. Evaporative cooling system for storage of fruits and vegetables-a review, J. Food Sci. Technol. 50, 429-442.
- Mahmood, M.H., Sultan, M., Miyazaki, T., Koyama, S., 2016. Desiccant Air-Conditioning System for Storage of Fruits and Vegetables: Pakistan Preview. Evergr. Jt. J. Nov. Carbon Resour. Sci. Green Asia Strategy. 3, 12-17.
- Maroulis, Z. B., Tsami, E., Marinos-Kouris, D., Saravacos, G. D., 1988. Application of the GAB model to the moisture sorption isotherms for dried fruits. J. food Eng. 7, 63-78.
- McGregor, B.M., 1989. Tropical products handbook. USDA office of transportation. Agricultural handbook No. 668.
- Meurant, G., 2012. Atmospheric oxidation and antioxidants. Elsevier, USA.
- Pahlevanzadeh, H., M. Yazdani., 2005. Moisture adsorption isotherms and isosteric energy for almond. J. Food Process Eng. 28, 331-345.
- Samaniego-Esguerra, C.M., 1989. Shelf life prediction of dried fruit and vegetables: a quantitative approach. Ph.D. thesis. Massey University, New Zealand.
- Singh, R.P., 1994. Scientific principles of shelf life evaluation, in: Man, C.M.D., and Jones,A.A., (Eds.), Shelf life evaluation of foods, Springer, pp. 3-24.
- Sultan, M., 2015. Study on Sorption Characteristics of Water Adsorbent for Agricultural Air-conditioning, Dr.Eng. thesis. Kyushu University, Japan.
- Sultan, M., El-Sharkawy, I.I., Miyazaki, T., Saha, B.B., Koyama, S., Maruyama, T., Maeda, S., Nakamura, T., 2015. Insights of water vapor sorption onto polymer based sorbents. Adsorption. 21, 205-215.
- Sultan, M., Miyazaki, T., Saha, B.B., Koyama, S., 2016. Steady-state investigation of water vapor adsorption for thermally driven adsorption based greenhouse air-conditioning system, Renew. Energy. 86, 785-795.
- Sultan, M., Miyazaki, T., 2017. Energy-efficient air-conditioning systems for nonhuman applications, in: Erken, O., (Eds.), Refrigeration, InTech open publisher, Crotia-EU, 22 pages. (ISBN # 978-953-51-5417-4)

- Tan, S. C. 1996. Mixed storage of fruit and vegetables. Farmnote No. 3/96. https://www.agric.wa.gov.au/fruit/mixed-storage-fruits-andvegetables?page=0%2C0#smartpaging_toc_p0_s0_h2
- Thompson, J., Kader, A.A., Sylva, K., 1996. Compatibility chart for fruits and vegetables in short-term transport or storage. Division of Agriculture and Natural Resources. Publication, 21560.
- Toledo, R.T., 2007. Fundamentals of food process engineering. Springer, USA.
- Usall, J., Ippolito, A., Sisquella, M., Neri, F., 2016. Physical treatments to control postharvest diseases of fresh fruits and vegetables. Postharvest Biol. Technol. 122, 30-40.
- Wolf, W., Spiess, W.E.L., Jung, G., Weisser, H., Bizot, H., Duckworth, R.B., 1984. The water-vapour sorption isotherms of microcrystalline cellulose (MCC) and of purified potato starch. Results of a collaborative study. J. Food Eng., 3, 51-73.

CHAPTER 6

GENERAL CONCLUSIONS

CHAPTER 6

GENERAL CONCLUSIONS

The main objective of the work presented in this thesis was to investigate the applicability of environmentally benign thermally driven desiccant air-conditioning system for agricultural storage applications. In this regards following work is carried out to cope the objectives and scope of the thesis

- The key factors responsible for postharvest losses (PHL) are determined for insights about complex mechanism of respiration, transpiration and fermentation.
- The motivation towards the use of DAC for product storage instead of conventional vapor compression refrigeration and/or air-conditioning systems is ascertained from thermodynamic and product quality point of view.
- Extensive literature review about the existing agricultural product storage techniques and the cooling systems is made to highlight the associated implications. It leads towards the development of low cost energy efficient air-conditioning systems for agricultural storage applications.
- The potential applicability of Maisotsenko Cycle based evaporative cooling systems is analyzed for farm air-conditioning.
- An open-cycle desiccant block experimental apparatus is setup for dehumidification experimentation.
- Uncertainty analysis is made for measure and/or calculated variables in order to determine the efficacy of experimental data.

- Experimental investigation of dehumidification performance of desiccant block for different process/regeneration air conditions and switching time ratio is made (five cases).
- A simplified correlation is developed on the conception of modification of isenthalpic slope of dehumidification line on psychrometric chart.
- The experimental validation of slope of dehumidification line under varying regeneration temperature is also investigated.
- The ideal storage zones of agricultural products, dried fruits, dried foods & feeds in comparison with greenhouse growth and humans thermal comfort zones are established on the psychrometric chart.
- Six different configurations of DAC systems are proposed and their performance evaluation is made under the ambient conditions of Fukuoka-Japan.
- A simplified methodology is developed for performance evaluation of proposed DAC systems under varying regeneration temperatures.
- The psychrometric evaluation of three optimized DAC systems is performed for of agricultural storage and parallel applications.

6.1 General Conclusions

General conclusion from this thesis can be extracted as follows:

- The conventional HVAC systems are difficult to provide optimum conditions for the storage of agricultural products. It is mainly due to the uneven control of humidity and/or ventilation.
- Direct/Indirect evaporative cooling systems yield low cost and environmental friendly AC systems in order to control sensible load of AC.
- M-Cycle based evaporative cooling systems can cool the air below wet-bulb temperature and approaches to the dew-point. Therefore, its applicability has been found suitable for the storage of various agricultural products where main objective is to reduce the temperature sensibly.
- Evaporative cooling systems including direct, indirect, M-Cycle are applicable to dry regions/climate, whereas desiccant air-conditioning can be employed efficiently in humid climates/regions.
- The performance of DAC is mainly based on desiccant unit, therefore experiments have been conducted for the performance evaluation of polymer sorbent based honeycomb like desiccant block. The experimental setup can reproduce the data within 2-3% error.
- The equivalent heat of adsorption (q_{eq}) of case-B at both regeneration temperatures (40°C and 60°C) is found greater than case-A due to higher amount of adsorbed water vapors.
- Average effective dehumidification slightly increases for case-C and case-D by changing the switching time ratio from 1:1 to 2:3 due to higher process air relative humidity. However, it keeps decreasing with increase in switching time ratio for case-E due to saturation of the sorbent.
- It has been found that the switching time depends on dehumidification amount, nature of application and operating conditions. From the bunch of experiments it has been generally concluded that the switching time ratio of 1:2, 2:3 and 1:1 can be selected for the operation of block based DAC system employing the condition of high, medium and low humidity operating conditions, respectively.
- Higher regeneration temperature brings fast and deep dehumidification, though it does not guarantee the higher system COP. An optimized regeneration temperature will be required for each operating conditions in order to make the system cost effective.
- According to the experiments made on open-cycle DAC system, it has been found that the real desiccant dehumidification process does not follow the isenthalpic behavior due to the adsorption heat. In contrary the real desiccant dehumidification process can be imagined on psychrometric charts employing the slope of desiccant dehumidification process line ranging from 0.8-0.31. For dehumidification cycle of longer than 10 minutes the value of slope may be taken as 0.31 for the simple analysis.

- Six DAC system combinations are proposed for various storage applications for Fukuoka climates. It is established on the basis of parametric and thermodynamic analysis of all the system configurations (S-I to S-VI) that S-II, S-IV and S-V could yield better system performance.
- S-II cannot achieve the latent and sensible load of AC for the storage of agricultural products at $T_{reg} = 50-70^{\circ}$ C under the varying ambient conditions of Fukuoka.
- \circ S-II achieves the latent load at T_{reg} = 80°C, however the sensible load can be achieved by two-stage cooling (MEC+DEC). Therefore, the modification of twostage cooling in is suggested configuration of S-II.
- S-II can provide the optimum conditions for the storage of dried fruits at low regeneration temperature ($T_{reg} = 70^{\circ}$ C). Moreover, the S-II can be effectively used for the provisions of greenhouse growth zone AC even at low regeneration temperature depending upon the type and growth stage of the crop.
- S-IV cannot be effectively used for the storage of agricultural products at all (50-80°C) regeneration temperatures.
- \circ S-IV almost achieves the latent load at $T_{reg} = 70^{\circ}C$ but sensible load for storage of dried fruits can be achieved by two-stage sensible cooling (MEC + MEC/IEC). Therefore two-stage sensible cooling modification is suggested in configuration of S-IV.
- S-IV can assist the conventional AC systems for the provision of human thermal comfort.
- S-V can be particularly used for the storage of dried food and feeds because it required about two times less regeneration heat input as compare to S-IV.

The study concludes that the evaporative cooling (preferably M-Cycle) systems should be considered on top priority for agricultural storage applications wherever these are thermodynamically and meteorologically applicable. When these are not applicable, thermally driven DAC systems could yield advance agricultural storage system which can control temperature and humidity distinctly irrespective to conventional compressor based AC systems. Therefore, integration of evaporative cooling unit(s) into DAC system will lead towards energy-efficient and reliable low-cost AC systems for various applications.

However optimum operational conditions will need to be determined and regulated for particular application. Herein it has been concluded that one or other DAC system could be efficiently utilized for the storage of agricultural products.

6.2 Future Work

- Development and standardization for agricultural products storage zones and validation with the real data.
- Consideration of non-polymeric sorbents for the formation of desiccant blocks to be used for desiccant air-conditioning.
- Dynamic simulation of proposed six systems and validation with experiments for polymeric and non-polymeric materials.
- Size estimation for corresponding DAC systems along with their components for various agricultural storage buildings.

APPENDIX A

SCOPE OF NANO/MICRO POLYMERIC MATERIALS IN FARM AIR CONDITIONING APPLICATIONS

APPENDIX A

SCOPE OF NANO/MICRO POLYMERIC MATERIALS IN FARM AIR CONDITIONING APPLICATIONS

Significance of nano and micro materials is obvious in 21st century for various technological, medical and engineering applications. In this regards, carbon based nano/micro materials have shown huge potential in many adsorption based applications e.g. water and wastewater treatment. In addition, these are successfully utilized for various (methanol-, ethanol-, and ammonia- based) close-cycle adsorption heat pump systems. Significance of nano/micro polymers is also well-known in drug/medical industry, and therefore extensively studied for various aspects of adsorption. Present study focuses its application in open-cycle based dehumidification and air-conditioning for agriculture sector. Study comprises two kinds of carbon and four kinds of polymer materials for potential application in farm air-conditioning. Water vapor adsorption comparison has been made among nano/micro materials and conventional hydrophilic adsorbent i.e. silica-gel. The size of desiccant unit in desiccant air-conditioning system has been determined and compared accordingly. Results showed that 2-3 times smaller system size can be yielded under particular conditions while utilizing nano/micro materials intelligently. It has been found that nano/micro materials facilitate the formation of various conceptual designs (e.g. honeycomb like structure, coating on heat exchanger etc.) of desiccant unit promoting the heat mass transfer between desiccant and air sides. Moreover according to literature review these materials can be regenerated at relatively low regeneration temperature, providing an opportunity to utilize the renewable solar thermal energy, bio-gas and bio-diesel.

A.1 Introduction

The vital need of heating, ventilation and air-conditioning (HVAC) for human thermal comfort is well known in the literature. Many designs of HVAC systems have also been investigated for various applications like conventional vapor compression airconditioning (VAC) for residential, commercial, institutional and office buildings, data center cooling, electronics cooling, manufacturing and storage processes etc. (Mahmood et al., 2016c). However, the effective farm level applications of these systems have not been much explored in the literature. Its farm level applications may include (i) air-conditioning (AC) for human (farm residents/workers) thermal comfort, (ii) AC for farm livestock thermal comfort, (iii) AC for farm greenhouse, and (iv) AC for pre-cooling and storage of farm produce etc. The conventional vapor compression refrigeration and/or air-conditioning systems are not only responsible for high energy requirements and environmental pollution but also cannot be used to provide optimal storage conditions to the agricultural products (e.g. mango, tomato, leafy vegetables etc.) due to chilling injury, discoloration and offflavor (Mahmood et al., 2016b; Ndukwu and Manuwa, 2015; Olosunde et al., 2016). Moreover, these systems are not best suited to control the latent and sensible load of greenhouse environment (Sultan, 2015). The complex mechanism of transpiration, respiration, fermentation in stored agricultural product and photosynthesis, evapotranspiration in growing greenhouse plants required active consideration about the provisions of ventilation rate, temperature and relative humidity control (Mahmood et al., 2016b; Sultan, 2015). On the other hand, the much higher ventilation rate is required in animal houses to remove the odors and ammonia. It may go up to maximum 60 air changes per hour in order to maintain the indoor air quality, temperature, and relative humidity (ASHRAE, 2007). The ammonia is most dangerous and chronic contaminant gas produced within the animal house due to the decomposition of manure. The conventional vapor compression refrigeration and/or air-conditioning systems are considered unfeasible and/or uneconomical for animal houses due to the requirements of higher air changes per hour (ASHRAE, 2007). The high ventilation rate is essential to maintain the indoor air quality by removing odor, ammonia and heat. Therefore, there is dire need of low cost, environmental friendly air-conditioning system/package that can be effectively employed for all farm level applications. In this regard, evaporative cooling technologies like direct evaporative cooling (DEC), indirect evaporative cooling (IEC)/M-Cycle evaporative cooling (MEC) have shown potential for farm level applications (Mahmood et al. 2016a; Sultan, 2015). However, these technologies do not perform efficiently under largely varying ambient conditions (particularly humid) due to limited cooling performance. The scope of DEC and IEC/MEC for various agricultural applications under varying environmental condition can be extended by the integration of desiccant dehumidification. The desiccant air-conditioning (DAC) comprises of desiccant dehumidification cum evaporative cooling has ability to deal the latent and sensible load of air-conditioning distinctly. It also gives opportunity to operate it by low grade waste heat, and renewable energy options like solar energy (particularly solar thermal) bio-gas, bio-diesel etc. Therefore, energy and environmental friendly DAC system represent zero global warming (GWP) and ozone depletion potential (ODP). In this regard, lots of practical airconditioning systems based on desiccant has been established and working all over the world. Mostly studied desiccant air-conditioning systems are standalone DAC systems with aid of simplified heat exchangers, direct and/or indirect evaporative cooling assisted DAC systems, vapor compression based hybrid DAC systems, solar energy operated DAC systems, bio-gas operated DAC systems, single/multi stage DAC systems, solid/liquid based DAC systems, wheel/block based DAC systems, heat and pressure regeneration based DAC systems and many more. Such characteristics of the DAC system make it promising for the widespread applications like agricultural product storage/preservation, greenhouses, marine ships, automobiles, buildings, museums, hospitals etc. (Ascione et al., 2009; Ascione et al., 2013; Enteria et al., 2009; Enteria et al., 2010; Guojie et al., 2012; Lee and Lee, 2013; Longo and Gasparella, 2015; Nagaya et al., 2006; Sultan et al., 2014a; Zhu and Chen, 2014).

In purview of above discussion, it is ascertain that the DAC system can be a promising solution for all farm level air-conditioning applications. Additionally, it can be operated by harvesting the renewable energy sources (solar energy, bio-gas, bio-diesel etc.)

abundantly available at the farm. The details about the typical system operation can be found from reference (Sultan et al., 2015a). The scopes of nano/micro material in desiccant air-conditioning are explained in heading A.2.

A.2 Scope of the Study

The desiccant air-conditioning despite of energy saving and eco-friendly technology still could not break the barriers for its market penetration. The main barriers are high initial cost, long payback period and system size (Sultan et al., 2015a). The one of the main component in DAC system is desiccant unit by which system size and operational cost are associated. There is dire need of research for the development of small size, low cost, nontoxic and non-corrosive adsorbent bed. The adsorbent should have high adsorption uptake with higher adsorption kinetics. In this regard nano/micro materials can play role towards the optimization of DAC system size. The significance of nano and micro materials is obvious for various technological, medical and engineering applications. Physiochemical and thermodynamic properties of nano/micro materials have been effectively realized for various adsorption based applications e.g. wastewater treatment, adsorption heat pump systems etc. Carbon based nano/micro materials have shown huge potentials in numerous adsorption applications particularly for water and wastewater treatment. It has been also successfully utilized for various (methanol-, ethanol-, and ammonia- based) close cycle adsorption heat pump systems. However, the present study focused towards the open cycle desiccant air-conditioning. The utilization of nano/micro materials in DAC and its further application in agriculture sector is relatively new in the literature. Therefore, .the study comprises two kinds of carbon and four kinds of polymer materials for potential application in farm air-conditioning. Water vapor adsorption comparison has been made among nano/micro materials and conventional hydrophilic adsorbent i.e. silica-gel. The size of desiccant unit in desiccant air-conditioning system has been determined and compared accordingly. The detailed methodology adopted in this regard is explained in heading A.3.

A.3 Materials and Methods

A.3.1 Materials

The study comprises different adsorbents such as polymer based, carbon based, and conventional silica-gel. The polymer based adsorbents include polymer sorbent-I (PS-I), polymer sorbent-II (PS-II) (Sultan et al., 2015b; Sultan et al., 2016a), and polystyrene sulfonic acid, sodium salt - mono sulfonated (PSS-MS) (Toribio et al., 2004; Toribio et al., 2005) and polystyrene sulfonic acid, sodium salt - full sulfonated (PSS-FS) (Toribio et al., 2004; Toribio et al., 2004; Toribio et al., 2005). The carbon based adsorbents include activated carbon powder (ACP) (Sultan et al., 2016b), and activated carbon fiber (ACF) (Sultan et al., 2016b). The particle size of these adsorbents is given in Table A.1 (Sultan et al., 2015b; Sultan et al., 2016a; Sultan et al., 2016b; Toribio et al., 2004; Toribio et al., 2005). All these adsorbents have specific adsorption characteristics. Silica-gel is mostly used in the designing of DAC systems due to its strong affinity towards water (Sultan et al., 2015a).

Adsorbents	Particle diameter (µm)	References
PS-I	2.6	Sultan et al., 2015b; Sultan et al., 2016a
PS-II	19.3	Sultan et al., 2015b; Sultan et al., 2016a
PSS-MS	10.0	Toribio et al., 2004; Toribio et al., 2005
PSS-FS	10.0	Toribio et al., 2004; Toribio et al., 2005
ACF	13.0	Sultan, 2015; Sultan et al., 2016b
ACP	105	Sultan, 2015; Sultan et al., 2016b
Silica-gel	1000	Sultan, 2015; Sultan et al., 2016b
PS-II PSS-MS PSS-FS ACF ACP Silica-gel	19.3 10.0 10.0 13.0 105 1000	Sultan et al., 2015b; Sultan et al., 2016a Toribio et al., 2004; Toribio et al., 2005 Toribio et al., 2004; Toribio et al., 2005 Sultan, 2015; Sultan et al., 2016b Sultan, 2015; Sultan et al., 2016b Sultan, 2015; Sultan et al., 2016b

Table A.1 The particle diameter of studied adsorbents.

The silica-gel has higher bulk density than PS-I but less than PS-II. It has been determined that the PS-II possesses about 1.5 times higher bulk density than PS-I and about 1.16 times higher than silica-gel (Sultan et al., 2015b). The microscopic images of PS-I and PS-II particles are shown in Figure A.1(a) and (b) respectively (Sultan, 2015). The illustration these of materials in to desiccant block, desiccant rotor, desiccant coated heat

exchangers is shown in Figure A.1. The other hydrophilic polymer adsorbents (PSS-MS and PSS-FS) were manufactured (by Polysciences, Europe) with the sulfonation of styrene (Toribio et al., 2004). The measured density of mono and full sulfonated PSS at room conditions was 1400 kg/m³ (Toribio et al., 2004; Toribio et al., 2005). It is important to mention that the density of adsorbent is crucial towards the optimization of system size for different farm level applications. The carbon based adsorbents (ACP, ACF) also have typical properties. The ACP is a highly porous adsorbent, whereas, ACF as a fibrous adsorbent possesses high porosity, and ease in handling [Sultan et al., 2016b]. The detailed information can be found from the cited literature.



Figure A.1 Few particles microscopic image of (a) PS-I, and (b) PS-II (Sultan et al., 2015b, Sultan et al., 2016a).

A.3.2 Data Analysis and Procedure

The isotherm plots of adsorbents (Table A.1) are drawn and compared in order to determine the adsorption uptake at respective temperature on the basis of data from references (Sultan, 2015; Sultan et al., 2015b; Sultan et al., 2016a; Sultan et al., 2016b; Toribio et al., 2004; Toribio et al., 2005). The isosteric heat of adsorption (Q_{st}) is also compared for PS-I, PS-II and silica-gel (Sultan, 2015; Sultan et al., 2015a; Sultan et al., 2015b). The performance of the adsorption heat pump systems (e.g. DAC system) mainly depends upon the amount of Q_{st} . Moreover, the contribution of the nano/micro polymeric material towards the system size reduction is investigated in comparison to conventional silica-gel. In this regard, the adsorbent to air mass fraction (MF_{A-A}) is calculated by the following relationship as given in the literature [Sultan et al., 2014b; Sultan et al., 2016b].

$$MF_{A-A} = \left(\frac{\Delta W}{\Delta M}\right)_{\Delta RH} = \left(\frac{W_1 - W_2}{M_1 - M_2}\right)_{\Delta RH}$$
(A.1)

where, MF_{A-A} describes the adsorbent to air mass fraction (g_{ads}/kg_{DA}), W_1 and W_2 are the humidity ratios (g_{H2O}/kg_{DA}), M_1 and M_2 are the adsorption uptakes (g_{H2O}/g_{ads}). The MF_{A-A} of the adsorbents (PS-I, PS-II, silica-gel) is calculated for high, medium and low humidity applications/conditions. The psychrometric representation of these applications is shown in Figure A.2.

The ambient air (T, RH) is considered same for all the adsorbents (PS-I, PS-II, silica-gel) in the respective humidity application. The state (1) in Figure A.3 represents the ambient air for medium humidity applications. Moreover, the states (2,3,4,5) represents the different stages of the simplified typical DAC cycle for different levels (a,b,c) of dehumidification and regeneration. The process/ambient air (1) becomes dehumidified as it passes through the adsorbents. The dehumidified air (2) is further passed through the sensible heat exchanger and/or evaporative coolers to supply the conditioned air. On the regeneration side the same ambient air (1) is used for the purpose. The regeneration air becomes hot by exchanging heat (via heat exchanger) with the process air. This hot regeneration air is then further heated (state 4) through external heat source to pass it though the adsorbent and finally exhaust (state 5) to the environment.

Appendix A Nano/Micro Polymeric Materials: Farm Air Conditioing Applications

The study towards the application side mainly focuses on the air-conditioning needs at the farm level. The psychrometric evaluation of thermal comfort zones for human and livestock, ideal growth zone for agricultural greenhouse, and ideal storage zones for agricultural products revealed that all these application required certain level of latent and sensible load of air-conditioning. In this regard DAC has shown potential to control the latent and sensible load of AC distinctly (Mahmood et al., 2016a; Mahmood et al., 2016b; Sultan, 2015; Sultan et al., 2015a).



Dry-bulb temperature [°C]



A.4 Results and Discussion

The water vapor adsorption isotherms are drawn on Figure A.4(a) using the data available in the literature for two carbon based and four polymer based nano/micro materials in comparison with silica-gel (Sultan et al., 2015b; Sultan et al., 2016a; Sultan et al., 2016b; Toribio et al., 2004; Toribio et al., 2005). It can be noticed that carbon based

adsorbents i.e. ACP and ACF shows hydrophobic behaviour at low relative pressure range, therefore may be neglected for the further investigation. However it is important to mention that such kind of water vapor adsorption might interesting for various high humidity based DAC applications e.g. agricultural greenhouses, storage of fresh fruits and likewise. This kind of adsorption is based on water vapor condensation using multilayer adsorption phenomena (Sultan et al., 2016a). The water vapor adsorption isotherms of ACP and ACF resemble to the IUPAC type-III and type-V isotherms, respectively. On the other hand, the adsorption isotherms of polymer adsorbents (PSS-MS, PSS-FS) represent the IUPAC type-II isotherm. The lower water vapor adsorption uptake has been observed in both PSS-MS and PSS-FS at low relative pressure. However, the hydrophilic nature of these polymers at higher relative pressure also depends upon the temperature (Toribio et al., 2004; Toribio et al., 2005).



Figure A.3 Schematic of the DAC system with psychrometric representation of ideal DAC cycle for medium humidity applications.



Figure A.4 Comparison of adsorption isotherms at 30°C for (a) PS-I, PS-II, ACP, ACF, PSS-FS, PSS-MS; (b) PS-I and PS-II with Silica-gel.

The full sulfonation of the polystyrene sulfonic acid, sodium salt (PSS) increased its adsorption capacity by 50% than mono sulfonation at particular conditions (Toribio et al., 2004). PSS-MS and PSS-MF enable the similar adsorption trend as achieved by PS-I and PS-II, but possesses considerable lower equilibrium adsorption uptake. Therefore the investigation of nano/micro materials in this study will be focused to PS-I and PS-II. In this regard PS-I and PS-II are compared with conventional silica-gel distinctly as depicting on Figure A.4(b). It can be seen from Figure A.4(b) that the water vapor adsorption uptake by silica-gel at saturation conditions (on 30°C) is about 0.40 kg/kg. But, the adsorption uptake by PS-I and PS-II under the same conditions is almost 2.0 and 2.5 times higher than silicagel adsorption uptake, respectively (Sultan, 2015). It is noticed that the isotherms of both the PS-I and PS-II have concave shape at the start (lower relative pressure) and then turned to convex shape as the relative pressure increases. Such shape of isotherms is due to two principal classes/fractions of the sorbed water (Smith, 1974; Sultan et al., 2015b). The first is bound water on the inner/outer surface of the adsorbent by the forces in excess of the normal forces which results in concave shape. The second fraction is responsible for convex shape because it is normally condensed within the adsorbent as the function of relative pressure. The summation of these fractions results in sigmoid shape polymer isotherms (Smith, 1974; Sultan et al., 2015b). It is worthy to mention that the sigmoid shape isotherms of polymeric material (PS-I and PS-II) are important for the designing of open cycle desiccant air-conditioning systems. Moreover, the bar graphs shown in Figure A.5 mainly represent the adsorption uptake fraction (ϕ) of PS-I and PS-II over silica-gel. The adsorption amounts of PS-I, and PS-II are divided by the adsorption amounts of the silica-gel in order to determine the Φ_{PS-I} and Φ_{PS-II} , respectively. It can be seen from Figure A.5 that Φ_{PS-II} is higher than the Φ_{PS-I} over the entire range of relative humidity. Therefore, it leads towards the better adsorption performance by PS-II under varying relative humidity conditions. The Q_{st} of the polymer adsorbents (PS-I, PS-II) and conventional silica-gel is also compared (Figure A.6) in order to further investigate the insights about these adsorbents. It represents the strength of adsorbent-refrigerant interaction and the performance of the adsorption heat pump systems mainly depend on it (Sultan 2015). The Q_{st} of the silica-gel largely vary as compare to PS-I and PS-II (Figure A.6). However, in case of PS-I and PS-II the isosteric heat of adsorption decreases relatively rapidly than silica-gel. Therefore, it can be concluded from this detailed discussion that the PS-I and PS-II can be potential adsorbents for the optimization of DAC system.



Figure A.5 Adsorption uptake fraction of adsorbents over silica-gel.



Figure A.6 Comparison of isosteric heat of adsorption of PS-I and PS-II with silica-gel.



Figure A.7 The adsorbent to air mass fraction under varying regeneration temperature for (a) high; (b) medium; and (c) low humidity conditions.

The contribution of the nano/micro polymeric material (PS-I, PS-II) towards the system size reduction is investigated in comparison to conventional silica-gel using Eq. (1) as explained in methodology section. It can be noticed that the MF_{A-A} increases with increasing regeneration temperature for all the adsorbents and under all humidity applications as shown in Figure A.7(a),(b),(c). It is determined that the MF_{A-A} for PS-II is well lower than silica-gel as compare to PS-I. In this regard, it can be envisaged that the PS-II can dehumidify the same amount of air with 2-3 times less adsorbent as compare to silica-gel under particular regeneration temperature. Therefore, the utilization of nano/micro polymeric material (PS-II) can leads towards the reduction in system size by 2-3 times under particular conditions as far as desiccant unit is concerned. Moreover, the nano/micro polymeric materials can be shaped in to honeycomb like structure (desiccant block, desiccant rotor, desiccant coated heat exchangers etc.). The polymeric materials can be regenerated at low temperature. It gives opportunity to harvest the renewable energy sources like solar energy (preferably solar thermal), bio-gas and bio-diesel (etc.) usually amply available at the farm. Therefore, the compact sized DAC system can be operated more economically for different farm applications by performing regeneration through renewable energy sources.

A.5 Conclusions

This study provides water vapor adsorption comparison among different nano/micro materials and conventional hydrophilic adsorbent, silica gel. The studied polymeric and carbon based nano/micro adsorbents include the PS-I, PS-II, PSS-MS, PSS-FS, ACP and ACF. It is investigated that ACP and ACF shows hydrophobic behaviour at low relative pressure range, therefore neglected for the further investigation in this study. However it is important to mention that such kind of water vapor adsorption might interesting for various high humidity based DAC applications e.g. agricultural greenhouses, storage of fresh fruits and likewise. On the other hand, PSS-MS and PSS-MF enable the similar adsorption trend as achieved by PS-I and PS-II, but possesses considerable lower equilibrium adsorption uptake. Therefore the investigation of nano/micro materials in this study mainly focused to

PS-I and PS-II. Moreover, the following generalized conclusions can be highlighted on the basis of this study.

- ✓ Carbon based adsorbents enable adsorption uptake at very high relative humidity range due to their adsorption isotherms of IUPAC type III and V.
- ✓ PSS-FS and PSS-MS enable IUPAC type II isotherm similar to PS-I and PS-II, however enable lower equilibrium adsorption uptake.
- ✓ Polymeric nano/micro adsorbents (PS-I and PS-II) enable higher water vapor adsorption uptake with IUPAC adsorption isotherm type II.
- ✓ The nano/micro polymeric can be well shaped into channels and honeycomb like structure (desiccant block, desiccant rotor, desiccant coated heat exchangers etc.).
- ✓ Polymeric nano/micro adsorbents facilitate low temperature regeneration, providing an opportunity to run the system on renewable energy (like solar, bio-gas, bio-diesel etc.).
- ✓ PS-II reduces the desiccant unit size (in the DAC system) by 2-3 times as compare to conventional silica-gel under particular conditions.

The study provides fundamental guidelines of polymer nano/micro materials application in air-conditioning, and therefore will be worthy for the researchers/scientists working for the development of advance polymer based nano or micro materials.

A.6 Nomenclature

AC	air-conditioning
ACF	activated carbon fiber
ACP	activated carbon powder
DAC	desiccant air-conditioning
Μ	adsorption uptakes [g _{H2O} /g _{ads}]
MF_{A-A}	adsorbent to air mass fraction [gads/kgDA]
P_a	partial vapor pressure [kPa]
P_s	saturation vapor pressure [kPa]
PS	polymer sorbent
PSS-FS	polystyrene sulfonic acid, sodium salt - full sulfonated

PSS-MS	polystyrene sulfonic acid, sodium salt - mono sulfonated
Q_{st}	isosteric heat of adsorption [kJ/kg]
RH	relative humidity [%] or [-]
Т	temperature [°C or K]
W	humidity ratio [g _{H2O} /kg _{DA}]
Φ	adsorbents adsorption uptake fraction over silica-gel

Subscripts

DA	dry air
ads	adsorbent

A.7 References

- Ascione, F., Bellia, L., Capozzoli, A., Minichiello, F., 2009. Energy saving strategies in airconditioning for museums. Appl. Therm. Eng. 29, 676-686.
- Ascione, F., Bellia, L., Capozzoli, A., 2013. A coupled numerical approach on museum air conditioning: Energy and fluid-dynamic analysis. Appl. Energy. 103, 416-427.
- ASHRAE, Handbook-HVAC Applications, 2007. American Society of Heating, Refrigerating and Air-conditioning Engineers, Inc., Atlanta.
- Enteria, N., Yoshino, H., Mochida, A., Takaki, R., Satake, A., Yoshie, R., Mitamura, T., Baba, S., 2009. Construction and initial operation of the combined solar thermal and electric desiccant cooling system. Sol. Energy., 83, 1300-1311.
- Enteria, N., Yoshino, H., Satake, A., Mochida, A., Takaki, R., Yoshie, R., Baba, S., 2010. Development and construction of the novel solar thermal desiccant cooling system incorporating hot water production. Appl. Energy. 87, 478-486.
- Guojie, Z., Chaoyu, Z., Guanghai, Y., Wu, C. 2012. Development of a new marine rotary desiccant air conditioning system and its energy consumption analysis. Energy Procedia. 16, 1095-1101.
- Lee, S.H., Lee, W.L., 2013. Site verification and modeling of desiccant-based system as an alternative to conventional air-conditioning systems for wet markets. Energy. 55, 1076-1083.

- Longo, G.A., Gasparella, A., 2015. Three years experimental comparative analysis of a desiccant based air conditioning system for a flower greenhouse: assessment of different desiccants. Appl. Therm. Eng. 78, 584-590.
- Mahmood, M.H., Sultan, M., Miyazaki, T., Koyama, S., 2016a. Desiccant air-conditioning system for storage of fruits and vegetables: Pakistan preview. Evergr. Jt. J. Nov. Carbon Resour. Sci. Green Asia Strategy. 3, 12-17.
- Mahmood, M.H., Sultan, M., Miyazaki, T., Koyama, S., 2016b. Theoretical and experimental analysis of desiccant air conditioning system for storage of agricultural products. Proc. of 16th Int. Refrig. Air Cond. Conf., Purdue University, USA, Paper No. 2261, 1-10.
- Mahmood, M.H., Sultan, M., Miyazaki, T., Koyama, S., Maisotsenko, V.S., 2016c. Overview of the Maisotsenko cycle–a way towards dew point evaporative cooling. Renew. Sustain. Energy Rev. 66, 537-555.
- Nagaya, K., Senbongi, T., Li, Y., Zheng, J., Murakami, I., 2006. High energy efficiency desiccant assisted automobile air-conditioner and its temperature and humidity control system. Appl. Therm. Eng. 26, 1545-1551.
- Ndukwu, M. C., Manuwa, S. I., 2015. Impact of evaporative cooling preservation on the shelf life of fruits and vegetable in South Western Nigeria. Res. Agric. Eng. 61, 122-128.
- Olosunde, W.A., Aremu, A.K., Onwude, D.I., 2016. Development of a solar powered evaporative cooling storage system for tropical fruits and vegetables. J. Food Process Preserv. 40, 279-290.
- Smith, S.E., 1947 The sorption of water vapor by high polymers. J. American Chem. Soc., 69, 646-651.
- Sultan, M., Miyazaki, T., Koyama, S., Saha, B.B., 2014a. Utilization of desiccant airconditioning system for improvement in greenhouse productivity: a neglected area of research in Pakistan. Int. J. Environ., 4, 1-10.
- Sultan, M., Miyazaki, T., Saha, B.B., Koyama, S., 2014b. Experimental study on carbon based adsorbents for greenhouse dehumidification. Evergr. Jt. J. Nov. Carbon Resour. Sci. Green Asia Strategy. 01,.5-11.

- Sultan, M., 2015. Study on Sorption Characteristics of Water Adsorbents for Agricultural Air-Conditioning Systems, Dr.Eng. Thesis, Kyushu University, Japan.
- Sultan, M., El-Sharkawy, I. I., Miyazaki, T., Saha, B. B., Koyama, S., 2015a. An overview of solid desiccant dehumidification and air conditioning systems. Renew. Sustain. Energy Rev. 46, 16-29.
- Sultan, M., El-Sharkawy, I.I., Miyazaki, T., Saha, B.B., Koyama, S., Maruyama, T., Maeda, S., Nakamura, T., 2015b. Insights of water vapor sorption onto polymer based sorbents. Adsorpt. 21, 205-215.
- Sultan, M., El-Sharkawy, I.I., Miyazaki, T., Saha, B.B., Koyama, S., Maruyama, T., Maeda, S., Nakamura, T., 2016a. Water Vapor Sorption Kinetics of Polymer Based Sorbents: theory and experiments. Appl. Therm. Eng. 106, 1359-4311.
- Sultan, M., Miyazaki, T., Saha, B.B., Koyama, S., 2016b. Steady-state investigation of water vapor adsorption for thermally driven adsorption based greenhouse airconditioning system. Renew. Energy. 86, 785-795.
- Toribio, F., Bellat, J.P., Nguyen, P.H., Dupont, M., 2004. Adsorption of water vapor by poly (styrenesulfonic acid), sodium salt: isothermal and isobaric adsorption equilibria. J. Colloid Interface Sci. 280, 315-321.
- Toribio, F., Bellat, J.P., Nguyen, P.H., Dupont, M., 2005. Adsorption of Water Vapor by Poly (styrene) Sulfonic Acid Sodium Salt—Isothermal and Isobaric Adsorption Equilibria. HVAC&R Res. 11, 305-317.
- Zhu, J., Chen, W., 2014. Energy and exergy performance analysis of a marine rotary desiccant air-conditioning system based on orthogonal experiment. Energy. 77, 953-962.

APPENDIX B

STRENGTHS, WEAKNESSES, OPPORTUNITIES, THREATS (SWOT) ANALYSIS OF LOW COST COOLING TECHNOLOGIES FOR AGRICULTURAL PRODUCT STORAGE

APPENDIX B

STRENGTHS, WEAKNESSES, OPPORTUNITIES, THREATS (SWOT) ANALYSIS OF LOW COST COOLING TECHNOLOGIES FOR AGRICULTURAL PRODUCT STORAGE

This chapter investigates the prospective and challenges associated with low cost, energy and environmental friendly appropriate cooling technologies for the storage of agricultural products. In this regard, strengths, weaknesses, opportunities, threats (SWOT) analysis is carried out for the purpose. It is used to investigate the strengths, weaknesses, opportunities and threats related to the evaporative cooling, desiccant air-conditioning and compressor based air-conditioning technologies in general and in particular when employed for the storage of agricultural products. The study comprises on extensive literature review, experimental and theoretical studies published in the literature and discussion with the researchers working in the parallel field. The resultant performance by all kinds of airconditioning systems were based on ambient air conditions as well as nature of storage application. However according to the findings, it has been concluded that the one or other technology (evaporative cooling or desiccant air-conditioning) could yield low cost airconditioning system for the storage of agricultural products.

B.1 Introduction

The global production of agricultural products (fruits and vegetables) has crossed 1.8 billion tons (www.fruitlogistica.de). However, the large amount of fruits and vegetables becomes lost due to postharvest losses. The postharvest losses (PHL) are the losses of the quantity and quality of the agricultural product throughout the postharvest chain. The PHL are accounted about 35-55% of the fresh agricultural produce depending upon the countries' geo-economic condition (Sanzani et al., 2016). The most adopted physical treatment to combat such losses is the cold storage (Usall et al., 2016). The most of the cold storages available today are equipped with refrigeration machines. Such systems are not only degrading the environment and consuming high primary energy but also cannot provide the optimal storage conditions (temperature and relative humidity) particularly to the tropical agricultural products and dry fruits (Mahmood et al., 2015; Mahmood et al., 2016b). It is because of diversified nature of agricultural products, chilling injury, and complex mechanism of transpiration, respiration and fermentation etc. In this perspective, low cost environmental friendly evaporative cooling technologies like direct evaporative cooling (DEC), indirect evaporative cooling (IEC)/M-Cycle evaporative cooling (MEC) have shown potential to provide storage conditions for particular agricultural products (Lal Basediya et al., 2013; Mahmood et al., 2016c). However, these technologies cannot be used effectively for the storage of agricultural products under largely varying ambient conditions (particularly humid) due to limited cooling performance (Lal Basediya et al., 2013). The scope of DEC and IEC/MEC for the storage of wide range of agricultural products under varying environmental condition can be extended by the integration of desiccant dehumidification. The desiccant air-conditioning (DAC) comprises of desiccant dehumidification cum evaporative cooling has ability to deal the latent and sensible load of air-conditioning distinctly. Such ability of the DAC system makes it promising to provide optimal conditions for the storage of agricultural products without affecting their quantitative, qualitative and nutritive attributes. The objective of this study is to find out the low cost, energy and environmental friendly appropriate cooling technologies for the storage of agricultural products and for other parallel agricultural applications.

B.2 Procedural Details

Strengths, Weaknesses, Opportunities, Threats (SWOT) analysis is carried out in this study. It is used to investigate the strengths, weaknesses, opportunities and threats



Figure B.1 Hierarchy of SWOT analysis of three air-conditioning dilemma for agricultural products storage.

associated with the evaporative cooling (DEC, IEC/MEC), desiccant air-conditioning (DAC) and compressor based air-conditioning (VAC) technologies in general and in particular when employed for the storage of agricultural products. SWOT is a strategic analysis tool and consists of internal and external factors/environment (Oxizidis and Papadopoulos, 2008). The internal factors examine the strengths and weaknesses of the technology with respect to its users and competitive systems, whereas the external factors (opportunities and threats) represent the individual perceptions about the technology future footings (Oxizidis and Papadopoulos, 2008). The key purpose of the SWOT analysis is to achieve the study objective by considering external and internal factors. Therefore, in the present study, SWOT analysis is carried out on the basis of extensive literature review

(Ming et al., 2014; Oxizidis and Papadopoulos, 2008; Paschalidou et al., 2016), experimental and theoretical studies published in the literature (Lal Basediya et al., 2013; Mahmood et al., 2015a; Mahmood et al., 2016a; Mahmood et al., 2016b; Mahmood et al., 2016c; Sultan et al., 2015) and discussion with the researchers working in the parallel field. The hierarchy of the SWOT analysis adopted in this study is shown Figure B.1.

B.3 Results and Discussion

In the present study the SWOT analysis is preformed to identify the prospects and challenges associated with agricultural products storage. The procedural details about SWOT analyses are explained in heading B.2. The strengths, weaknesses, opportunities and threats of different air-conditioning technologies are briefly elaborated in the Table B.1. The major strengths of the evaporative cooling technologies for storage of agricultural products include its suitability for the storage of most of agricultural products without causing chilling injury, discoloration, off-flavor (Lal Basediya et al., 2013; Mahmood et al., 2016a), low cost, and supply of 100% fresh air [Mahmood et al., 2016a; Mahmood et al., 2016c]. Consequently fermentation is avoided by means of ventilation which maintains the quantitative, qualitative and nutritive attributes of the stored products (Mahmood et al., 2016a; Mahmood et al., 2016b). However, these technologies consume water to supply the fresh cooled air therefore the availability of water at site must be ensured during the storage period. The direct evaporative cooling also cannot be promptly used to provide storage conditions for the dry agricultural products. Moreover, the cooling performance of evaporative cooling technologies depends on the ambient conditions. In human thermal comfort, mostly DEC and IEC are considered suitable technologies when ambient air wet bulb temperature is $\leq 20^{\circ}$ C and $\leq 25^{\circ}$ C, respectively (Lal Basediya et al., 2013). However, the ideal design conditions for agricultural product storage may vary depending upon the nature of the storage. According to the literature MEC can work efficiently when ambient air dry bulb temperature and humidity ratio are up to 45°C and 11.2 g/kgDA, respectively (Mahmood et al., 2016c). Their (DEC/IEC/MEC) effectiveness decreases greatly in moderate and humid conditions. In humid areas the potential of all kind of evaporative cooling systems is limited, however, desiccant air-conditioning system could be a feasible



Table B.1 SWOT analysis of air-conditioning technologies.



choice in such scenario. In other words, one or other technology (evaporative cooling or desiccant air- conditioning) could yield low cost air-conditioning system. Studies have shown the potential of DAC system in humid and semi humid climatic conditions. It also gives opportunity to utilize low grade waste heat and renewable energy options particularly solar thermal energy. Lots of practical air-conditioning systems based on desiccant has been established and working all over the world. Mostly studied desiccant air-conditioning systems are standalone DAC systems with aid of simplified heat exchangers, direct and/or indirect evaporative cooling assisted DAC systems, vapor compression based hybrid DAC systems, solar energy operated DAC systems, bio-gas operated DAC systems, single/multi stage DAC systems, solid/liquid based DAC systems, wheel/block based DAC systems, heat and pressure regeneration based DAC systems and many more. The detail insights about DAC system can be found from reference (Sultan et al., 2015). The theoretical and experimental evaluation of the DAC system has been performed at low regeneration temperature (50-60°C) for storage of agricultural products under varying climatic conditions (Mahmood et al., 2015; Mahmood et al., 2016a, Sultan et al., 2014). It has been found in these studies that DAC can achieve the latent and sensible load of air-conditioning for the storage of agricultural products. Moreover, in case of higher sensible loads hybrid DAC system can be employed. In contrast to DAC and DEC/IEC/MEC, the refrigeration machines are being mostly used in the cold storages. Such systems, aside from high energy consumption and environmental degradation, cannot be used for the storage of agricultural products. It is due to uneven control of ambient relative humidity which ultimate effects quality attributes of the agricultural products. In addition due to excessive cooling chilling injury may occur in the product.

B.4 Conclusions

This study analyses the strengths, weaknesses, opportunities, threats (SWOT) related to the evaporative cooling, desiccant air-conditioning and compressor based air-conditioning technologies in general and in particular when employed for the storage of agricultural products. The resultant performance by all kinds of air-conditioning systems were based on ambient air conditions as well as nature of storage application. However

according to the findings, it has been concluded that the one or other technology (evaporative cooling or desiccant air-conditioning) could yield low cost air-conditioning system for the storage of agricultural products. Moreover, these technologies (evaporative cooling or desiccant air-conditioning) can be used for the storage of agricultural products without causing adverse effects on the product qualitative and nutritive attributes.

B.5 Nomenclature

ΔP	pressure drop
DAC	desiccant air-conditioning
DEC	direct evaporative cooling
Е	effectiveness [-]
h	enthalpy [kJ/kg]
IEC	indirect evaporative cooling
MEC	M-Cycle evaporative cooling
PHL	postharvest losses [%] or [-]
RH	relative humidity [%]
Т	temperature [°C or K]
VAC	vapor compression air-conditioning
X	humidity ratio [g/kgDA]

Subscript

d/dp	dew-point
w/wb	wet-bulb

B.6 References

- Lal Basediya, A., Samuel D.V.K., Beera V., 2013. Evaporative cooling system for storage of fruits and vegetables-a review. J. Food Sci. Technol. 50, 429-442.
- Mahmood, M.H., Sultan, M., Miyazaki, T., Koyama, S., 2015. Evaluation of desiccant airconditioning applicability for agricultural sector of Pakistan. The 91th Annual JSRAE conference (JSRAE-2015), October 20-23, 2015. Waseda University, Japan.

- Mahmood, M.H., Sultan M., Miyazaki, T., Koyama, S., 2016a. Theoretical and experimental analysis of desiccant air conditioning system for storage of agricultural products. 16th International refrigeration and air conditioning conference at Purdue, July 11-14, 2016. Purdue University, USA.
- Mahmood, M.H., Sultan, M., Miyazaki, T., Koyama, S., 2016b. Desiccant air-conditioning system for storage of fruits and vegetables: pakistan preview. Evergr. Jt. J. Nov. Carbon Resour. Sci. Green Asia Strategy. 3, 12-17.
- Mahmood, M.H., Sultan, M., Miyazaki, T., Koyama, S., Maisotsenko, V.S., 2016c. Overview of the Maisotsenko cycle–A way towards dew point evaporative cooling. Renew. Sustain. Energy Rev. 66, 537-555.
- Ming, Z., Shaojie, O., Yingjie, Z., Hui S., 2014. CCS technology development in china: status, problems and countermeasures—based on SWOT analysis. Renew. Sustain. Energy Rev. 39, 604-616.
- Oxizidis, S., Papadopoulos, A.M., 2008. Solar air conditioning: a review of technological and market perspectives. Adv. Build. Energy Res. 2, 123-157.
- Paschalidou, A., Tsatiris, M., Kitikidou, K., 2016. Energy crops for biofuel production or for food?-SWOT analysis (case study: Greece). Renew. Energy. 93, 636-647.
- Sanzani, S.M., Reverberi, M., Geisen, R., 2016. Mycotoxins in harvested fruits and vegetables: Insights in producing fungi, biological role, conducive conditions, and tools to manage postharvest contamination. Postharvest Biol. Technol. 122, 95-105.
- Sultan, M., El-Sharkawy, I.I., Miyazaki, T., Saha, B.B., Koyama, S., 2015. An overview of solid desiccant dehumidification and air conditioning systems. Renew. Sustain. Energy Rev. 46, 16-29.
- Usall, J., Ippolito, A., Sisquella, M., Neri F., 2016. Physical treatments to control postharvest diseases of fresh fruits and vegetables. Postharvest Biol. Technol. 122, 30-40.

APPENDIX C

ENERGY RECOVERY IN CONDENSERS AND COOLING TOWERS

APPENDIX C

ENERGY RECOVERY IN CONDENSERS AND COOLING TOWERS

C.1 Maisotsenko Cooling Tower

Cooling tower is a heat rejection device which rejects waste heat to the atmosphere through the cooling of a water stream to a lower temperature (He et al., 2015; Tyagi et al., 2008; Tyagi et al., 2012). The most common applications for cooling towers are providing cooled water for AC, manufacturing and power generation processes. The schematic diagrams for conventional (CCT) and Maisotsenko (MCT) cooling towers are shown in Figure C.1(a)-(d) (Anisimov et al., 2014c; Gillan et al., 2011b; Khalatov et al., 2011; Morosuk et al., 2012). In addition, the corresponding psychrometric processes are elaborated on Figure C.2(a) and C.2(b) (Anisimov et al., 2014c; Gillan et al., 2014c; Gillan et al., 2011b), respectively. It can be seen that the CCT cools the hot water from the cooling tower to the ambient air wet-bulb temperature using DEC technique (Anisimov et al., 2014c; Qureshi and Zubair, 2006). On the other hand the MCT cools the water stream towards the dewpoint of the ambient air (Anisimov et al., 2012; Wicker, 2003).



Figure C.1 Schematic diagram for: (a) general cooling tower flow scheme, (b) CCT, (c) close circuit MCT, and (d) open circuit MCT (reproduced from Anisimov et al., 2014c; Gillan et al., 2011b; Khalatov et al., 2011; Morosuk et al., 2012).

In both open and close circuit based MCTs, the air in dry-channel is sensibly cooled before it enters to the wet-channel for evaporative cooling which enables it to cool the water near to the dew-point temperature of the ambient air (Maisotsenko et al., 2005a; Morozyuk and Tsatsaronis, 2011; Sverdlin et al., 2011; Wicker, 2003). Furthermore, it also helps to reduce the pressure drop and fan power. The major differences between CCT and MCT are air flow route and water distribution through the fill (packing). Therefore, MCT fill can be designed as retrofit to the operational CCT in order to substantially decrease cooled water temperature.

The open and close circuit based MCTs are compared with the CCTs in order to analyze the M-Cycle applicability in cooling towers (Gillan et al., 2011b). In comparison with open circuit, the close circuit MCT has no direct contact of air with the process fluid i.e. water or a glycol/water mixture. Results showed that the both MCTs cooled the water at lower temperature than the CCT and enable double evaporation rate. Furthermore, the open circuit MCT has been preferably proposed by the authors because of its simple fill structure. and has been analyzed by many other researchers (Anisimov et al., 2014c; Khalatov et al., 2011; Morozyuk and Tsatsaronis, 2011; Sverdlin et al., 2011). Morosuk et al., 2012, simulated the COP of the MCT for various ambient air conditions. According to the results, the MCT yields the COP from 0.20 to 0.90 depending on the available conditions. Unlike the CCTs the COP of the MCT increases with the increase in ambient air temperature which distinguishes its applicability in hot climates. In another simulation based study (Anisimov et al., 2014c) the COP of the MCT was found up to 0.86 along with 2.20% evaporation rate. Furthermore, Sverdlin et al., 2011, analyzed the M-Cycle base cooling tower using a simulation program which was validated on the basis of real field data from existing cooling towers. It was concluded that the M-Cycle based cooling towers can conditionally reduce the water temperature up to the dew-point of the ambient air.


Figure C.2 Psychrometric representation of cooling tower operation for: (a) CCT; and (b) close/open circuit MCT (reproduced from Anisimov et al., 2014c; Gillan et al., 2011b).

C.2 Maisotsenko Condenser

One of the key component in vapor compression, absorption cooling and refrigeration system is the condenser, which could be air-cooled, water-cooled or evaporative condenser (Hajidavalloo and Eghtedari, 2010; Hosoz and Kilicarslan, 2004; Hwang et al., 2001; Li et al., 2009). The M-Cycle as an advance IEC process can be efficiently applied to the condensers of the air-conditioning, refrigeration and power producing systems (Maisotsenko et al., 2003; Maisotsenko et al., 2007). The Maisotsenko condenser (M-Condenser) is an evaporative condenser that can considerably enhance the energy efficiency of the system by rejecting the heat efficiently (Gorshkov, 2012; Khalatov et al., 2011; Maisotsenko, 2006; Idalex Technologies, 2006; Wani and Ghodke, 2011; Wani et al., 2012).



Figure C.3 Experimental setup for performance comparison between the air-cooled and M-Condenser (reproduced from Gillan et al., 2011a; Idalex Technologies, 2006; Maisotsenko, 2006).



Figure C.4 Effect of ambient air conditions on the performance of air-cooled and M-Condenser (reproduced from Gillan et al., 2011a). Lines are obtained from the best fit of experimental data.

Gillan et al., 2011a, performed an experimental study on M-Condenser in order to compare its performance with the conventional air-cooled condenser. Figure C.3 (Gillan et al., 2011a; Maisotsenko, 2006; Idalex Technologies, 2006) shows the experimental setup which mainly consists of: (1) compressor, (2) air-cooled condenser, (3) M-Condenser, (4) fans, and (5) valves. The arrangement of values facilitated in determination of performance parameters of air-cooled and M-Condenser independently. The M-Condenser was composed of aluminum based micro channels for refrigerant flow, whereas the product and working channels are realized by cellulose sheets and plastic coating as shown in Figure C.3. Effect of ambient conditions on energy efficiency ratio (EER) was determined for air-cooled and M-Condenser remains stable by increasing the ambient air temperature from 26.7°C to 43.3°C whereas it decreases in case of air-cooled condenser. Furthermore, the M-Condenser obtained 30% higher EER (on an average) as compared to air-cooled condenser with a minimum of 9% and a maximum of 58%.

C.3 Nomenclature

ССТ	conventional cooling tower
COP	coefficient of performance
DEC	direct evaporative cooling
EER	energy efficiency ratio
IEC	indirect evaporative cooling
МСТ	Maisotsenko cooling rower

Subscript

dp	dew-point
wb	wet-bulb

C.4 References

- Anisimov, S., Kozlov, A., Glanville, P., Khinkis, M., Maisotsenko, V., Shi, J., 2014c. Advanced Cooling Tower Concept for Commercial and Industrial Applications. Proc ASME 2014 Power Conf., vol. 2, Maryland, USA: ASME. V002T10A001 (7pages). doi:10.1115/POWER2014-32020.
- Gillan, L., Gillan, A., Kozlov, A., Kalensky, D., 2011a. An advanced evaporative condenser through the Maisotsenko cycle. Int. J. Energy Clean Environ. 12, 251-258. doi:10.1615/InterJEnerCleanEnv.2013006619.
- Gillan, L., Glanville, P., Kozlov, A., 2011b. Maisotsenko-Cycle enhanced cooling towers. Proc 2011 Cool. Technol. Inst. Annu. Conf., Texas, USA: Cooling Technology Institute. 1-7, (paper no. TP11–3).
- Gorshkov, V., 2012. Systems based on Maisotsenko cycle: Coolerado coolers. Bachelor's Thesis. Mikkeli University of Applied Sciences.
- Hajidavalloo, E., Eghtedari, H., 2010. Performance improvement of air-cooled refrigeration system by using evaporatively cooled air condenser. Int. J. Refrig. 33, 982-988. doi:10.1016/j.ijrefrig.2010.02.001.
- He, S., Gurgenci, H., Guan, Z., Huang, X., Lucas, M., 2015. A review of wetted media with potential application in the pre-cooling of natural draft dry cooling towers. Renew. Sustain. Energy. Rev. 44, 407-422. doi:10.1016/j.rser.2014.12.037.

- Hosoz, M., Kilicarslan, A., 2004. Performance evaluations of refrigeration systems with air-cooled, water-cooled and evaporative condensers. Int. J. Energy Res. 28, 683-696. doi:10.1002/er.990.
- Hwang, Y., Radermacher, R., Kopko, W., 2001. An experimental evaluation of a residential-sized evaporatively cooled condenser. Int. J. Refrig. 24, 238-249. doi:10.1016/S0140-7007(00)00022-0.
- Idalex Technologies, 2006. The Idalex evaporative condenser: Harnessing the power of nature to redefine energy efficiency. Business opportunity report, Idalex Technologies, Inc. Arvada, CO 80003, USA.
- Khalatov, A., Karp, I., Isakov, B., 2011. Prospects of the Maisotsenko thermodynamic cycle application in Ukraine. Int. J. Energy Clean Environ. 12, 141-157. doi:10.1615/InterJEnerCleanEnv.2012005916.
- Li, Y., Wu, J., Shiochi, S., 2009. Modeling and energy simulation of the variable refrigerant flow air conditioning system with water-cooled condenser under cooling conditions. Energy Build. 41, 949-957. doi:10.1016/j.enbuild.2009.04.002.
- Maisotsenko, V., Gillan, L.E., Heaton, T.L., Gillan. A.D., 2003. Method and plate apparatus for dew point evaporative cooler. US Patent No. US6581402 B2.
- Maisotsenko, V., Gillan, L.E., Heaton, T.L., Gillan, A.D., 2005a. Method of evaporative cooling of a fluid and apparatus therefor. US Patent No. US6854278 B2.
- Maisotsenko, V.S., 2006. Evaporative cooled micro-channel condenser project through the Maisotsenko cycle. Idalex Inc. Denver CO 80207, USA.
- Maisotsenko, V., Gillan, L.E., Heaton, T.L., Gillan, A.D., 2007. Method and plate apparatus for dew point evaporative cooler. US Patent No. US7197887 B2.
- Morozyuk, T., Tsatsaronis, G., 2011. Advanced cooling tower concept based on the Maisotsenko-cycle - an exergetic evaluation. Int. J. Energy Clean Environ. 12, 159-173. doi:10.1615/InterJEnerCleanEnv.2012006013.
- Morosuk, T., Tsatsaronis, G., Maisotsenko, V., Kozlov, A., 2012. Exergetic Analysis of a Maisotsenko-Process-Enhanced Cooling Tower. vol. 6, Texas, USA: ASME. 189-194. doi:10.1115/IMECE2012-87581.
- Qureshi, B.A., Zubair, S.M., 2006. A complete model of wet cooling towers with fouling in fills. Appl. Therm. Eng. 26, 1982-1989. doi:10.1016/j.applthermaleng.2006.01.010.

- Sverdlin, B., Tikhonov, A., Gelfand, R., 2011. Theoretical possibility of the Maisotsenko cycle application to decrease cold water temperature in cooling towers. Int. J. Energy Clean Environ. 12, 175-185. doi:10.1615/InterJEnerCleanEnv.2012005876.
- Tyagi, S.K., Wang, S., Park, S.R., Sharma, A., 2008. Economic considerations and cost comparisons between the heat pumps and solar collectors for the application of plume control from wet cooling towers of commercial buildings. Renew. Sustain. Energy Rev. 12, 2194-2210. doi:10.1016/j.rser.2007.03.012.
- Tyagi, S.K., Pandey, A.K., Pant, P.C., Tyagi, V.V., 2012. Formation, potential and abatement of plume from wet cooling towers: A review. Renew. Sustain. Energy Rev. 16, 3409-3429. doi:10.1016/j.rser.2012.01.059.
- Wani, C., Ghodke, S., 2011. Performance analysis of a Maisotsenko cycle-based energyefficient evaporative air conditioner. Int. J. Energy Clean Environ. 12, 327-340. doi:10.1615/InterJEnerCleanEnv.2013006192.
- Wani, C., Ghodke, S., Shrivastava, C., 2012. A review on potential of Maisotsenko cycle in energy saving applications using evaporative cooling. Int. J. Adv. Res. Sci. Eng. Technol. 1, 15–20.
- Wicker, K. 2003. Life below the wet bulb: The Maisotsenko cycle. Power. 147, 29-31.

APPENDIX D

Advances of Psychrometric Renewable Energy in Power Turbines

APPENDIX D

Advances of Psychrometric Renewable Energy in Power Turbines

D.1 Maisotsenko Cycle Conception in Gas Turbines

The gas turbine usually consists of: (1) a compressor, (2) a combustion chamber, (3) a turbine, and (4) a generator (Cengel and Boles, 1994; Cohen et al., 1996). A simple layout of open cycle gas turbine operating on ideal Brayton cycle is shown in Figure D.1(a). Referring to T-S diagram shown in Figure D.1(b), the ideal Brayton cycle is made up of four completely irreversible processes i.e. $(1 \rightarrow 2)$ isentropic compression; $(2 \rightarrow 3)$ constant pressure heat addition; $(3 \rightarrow 4)$ isentropic expansion; and $(4 \rightarrow 1)$ constant pressure heat rejection. The detail insight of the operational principle can be found from the ref. [Yee et al., 2008].

The gas turbine with the working fluid of air-water mixture enables high electric efficiency and high specific power output to specific investment cost below that of combined cycles (Jonsson and Yan, 2005; Poullikkas, 2005). Various humidified gas turbine cycles have been proposed in the literature (e.g. direct water-injected cycles, steam-injected cycles, and evaporative cycles with humidification towers etc.), though only few are available commercially. In addition, the Maisotsenko gas turbine conceptions based on humidified gas turbine cycles have been recently proposed (Buyadgie et al., 2015; Gillan and Maisotsenko, 2003; Maisotsenko et al., 2004; Maisotsenko et al., 2005; Saghafifar and Gadalla, 2015a; Saghafifar and Gadalla, 2015b; Saghafifar and

Gadalla, 2015c). The details about particular gas turbine type can be found from the cited references. The key features of the humidified gas turbine can be explained as (Jonsson and Yan, 2005). Addition of water or steam increases the combustion chamber efficiency as well as mass flow rate passing through the turbine, and consequently enhances the specific power output. In this regard, the compressor work remains constant (if the water/steam is added after the compressor), and least work is required to increase the pressure of a liquid as compared to a gas. The overall cycle efficiency increases by utilizing the turbine exhaust heat for generating hot water and/or steam in order to preheat the combustion air in the recuperator. Addition of water before combustion reduces the compressed air temperature at the inlet to the recuperator, which improves the energy recovery rate. Furthermore, humidifying the gas turbine working fluid helps in reduction of NO_x formation during the combustion process.

From the above prospective the M-Cycle roles in the improvement of gas turbine efficiency can be listed as follows:

- ✓ To supply hot and humid air into the combustion chamber that improves the cycle efficiency, and consequently augments the power generation.
- ✓ To recover the thermal energy from the exhaust gasses in order to improve the cycle efficiency.
- ✓ To provide cooled air to the compressor for turbine inlet air cooling which improves the compressor efficiency.
- ✓ To reduce the NO_x formation during combustion process by providing excessive moist air.

In order to achieve the above mentioned M-Cycle roles, the coming headings explain the recently proposed three different Maisotsenko gas turbine cycles. It is worthy to mention that the M-Cycle is a heat recovery conception, and is not limited to the discussed cycles only. Its applications could be more versatile, though the study focuses only few cases because of the limited literature.





D.2 Maisotsenko Humid Air Turbine Cycle

The evaporative gas turbine or humid air turbine (HAT) cycle involves in injection of water in the humidification tower with a water loop recirculation system (Saghafifar and Gadalla, 2015a; Jonsson and Yan, 2005). The importance of HAT cycle is well-known in the literature and considered as one of the most efficient humidified gas turbine cycle (Jonsson and Yan, 2005). Gallo, 1997, compared the HAT cycle performance with the other gas turbine power cycles. According to the results the HAT cycle obtained optimum efficiency at moderate pressure ratio whereas the specific work output increases with the increase in pressure ratio (Saghafifar and Gadalla, 2015a; Gallo, 1997). Furthermore, it achieved the highest efficiency of 54.8% among the other studied cycles when the turbine inlet temperature and pressure ratio was about 1300°C and 12, respectively. Similarly another study (Lazzaretto and Segato, 2002) showed the maximum efficiency of 54.6% for the total pressure ratio of 20.

The Maisotsenko humid air turbine (M-HAT) cycle is one of recently proposed humidified gas turbine cycle (Gillan and Maisotsenko, 2003; Jenkins et al., 2014a; Jenkins et al., 2014b; Khalatov et al., 2011; Maisotsenko et al., 2004; Maisotsenko et al., 2005; Maisotsenko et al., 2006; Reyzin, 2011; Saghafifar and Gadalla; 2015a; Saghafifar and Gadalla; 2015b). The simplified schematic diagram of the M-HAT cycle is shown in Figure D.2(a). It works on evaporative gas turbine principle in which the humidification tower is replaced by the M-Cycle air saturator assembly (Khalatov et al., 2011; Reyzin, 2011; Saghafifar and Gadalla; 2015a) e.g. shell and tube air saturator (Gillan and Maisotsenko, 2003; Saghafifar and Gadalla; 2015a), as shown in Figure D.2(a). The operational scheme of the M-Cycle shell and tube air saturator is shown in Figure D.2(b) [Gillan and Maisotsenko, 2003]. Referring to Figure D.2, air at state (1) is compressed adiabatically to state (2) by the compressor. The compressed air enters into the M-Cycle air saturator where it is heated and humidified at state (3) by utilizing the waste heat from the turbine exhaust gases, and consequently supplied to the combustion chamber. The turbine exhaust gases enter into the M-Cycle air saturator at state (5) and exit at state (6). The detailed operational procedure can be found from (Gillan and Maisotsenko, 2003; Saghafifar and Gadalla; 2015a).







Saghafifar and Gadalla, 2015a, performed the detailed analysis of M-HAT power cycle with a comprehensive model of air saturator. The comparative analysis between HAT and M-HAT cycles has been conducted by the authors in order to optimize the effect of

compressor pressure ratio, inlet air temperature, combustor outlet temperature, water inlet temperature, and air saturator degree of humidification. Maximum efficiency by the M-HAT was achieved when water addition in the upper section of the air saturator (Figure D.2b) was limited to cool the exhaust gases. It has been concluded that air saturator in comparison with conventional heat exchanger can increase the plant efficiency and specific work output by 7% points and 44.4%, respectively. In addition, the improvement results in 13,000 tonnes of natural gas fuel saving per year. Moreover, the M-HAT cycle possesses greater efficiency than the HAT cycle at higher pressure ratios. It is worth mentioning that the specific work output by the M-HAT cycle was continuously greater than one achieved by the HAT cycle regardless of compression pressure ratio. Similarly various studies (Gillan and Maisotsenko, 2003; Jenkins et al., 2014a; Jenkins et al., 2014b; Khalatov et al., 2011; Reyzin, 2011) conclude that the thermodynamic efficiency of M-HAT cycle is significantly higher than any humidified gas turbine cycle including the HAT cycle. Consequently, the present study concludes that the M-HAT cycle has challenged the HAT cycle for the title of optimal humidified gas turbine cycle.

D.3 Maisotsenko Air Bottoming Cycle

Conventional combined power cycles operating on gas turbine topping cycle and a steam turbine bottoming cycle are considered as the most efficient combined power cycles (Cengel and Boles, 1994; Cohen et al., 1996; Saghafifar and Gadalla, 2015b). However, it has been reported that the fact is not applicable for the small scale power plants with capacity of \leq 50MWe (Poullikkas, 2005), because of the condenser and heat recovery steam generator in the steam bottoming cycle (Saghafifar and Gadalla, 2015b). In this regard, organic Rankine cycle (ORC) (Bianchi and De Pascale, 2011; Chacartegui et al., 2009; Clemente et al., 2013;; Kaikko et al., 2001) and air bottoming cycle (ABC) (Ghazikhani et al., 2014; Korobitsyn, 2002; Pierobon and Haglind, 2014; Saghafifar and Poullikkas, 2014) has been extensively studied in order to replace the steam bottoming cycle. The ABC cycle was proposed in the late 1980s and is well-known in the literature due to its simplicity and compact design (Saghafifar and Gadalla, 2015b).



Figure D.3 Schematic diagram of the Maisotsenko air bottoming cycle (M-ABC) (reproduced from Saghafifar and Gadalla, 2015b).

Maisotsenko air bottoming cycle (M-ABC) (Saghafifar and Gadalla, 2015b) is a recently proposed ABC cycle which is an integration of Maisotsenko gas turbine cycle (Saghafifar and Gadalla, 2015a) as a bottoming cycle to a topping simple gas turbine cycle (Cengel and Boles, 1994; Cohen et al., 1996). In the M-ABC, the conventional air heat exchanger of the ABC is replaced by the M-Cycle air saturator cum heat exchanger. The schematic diagram of the M-ABC cycle is shown in Figure D.3 (Saghafifar and Gadalla, 2015b). It can be seen that the hot exhaust gases from turbine No. 1 enters into the M-Cycle

air saturator cum heat exchanger at state (4) where it transfer its heat using the M-Cycle principle and exit at state (5). On the other hand the compressed air at state (7) is heated and humidified simultaneously by means of M-Cycle principle, and supplied to the turbine No. 2 at state (11). A comprehensive analysis of M-ABC is reported by Saghafifar and Gadalla 2015b. The authors developed a thermodynamic model for M-ABC with the detailed air saturator model. It has been concluded that the M-ABC enables higher efficiency as compared to conventional ABC at the optimum operating conditions. The specific work output by the M-Cycle air saturator was 43% higher than the conventional heat exchanger. Consequently, the efficiency was enhanced by 3.7%, which is equivalent to 2600 tonnes of natural gas fuel saving per year. Furthermore, the authors reported that the replacement of conventional ABC power plants with the M-ABC can yield savings of 0.655 US\$M every year for the plant operating hours of 8000 hr/year and fuel cost of 5.5 US\$/GJ. Hence, it has been concluded that the M-ABC has thermodynamic superiority over ABC, which can setup an advance combined power cycle.

D.4 Maisotsenko Sub-Atmospheric Brayton Cycle

The modern gas turbine engines are based on well-known open Brayton cycle (see Figure D.1) (Cengel and Boles, 1994; Cohen et al., 1996), which begins with air compression (in the compressor) followed by constant pressure heat addition (in the combustion chamber) and finally terminates in the gas turbine at the ambient pressure conditions. The overall disadvantage of the cycle is the significant amount of waste heat discharged into the atmosphere which results in poor thermal efficiency (Khalatov et al., 2015). In this regard, inverse Brayton cycle (Wilson and Korakianitis, 1984) have been investigated with different configuration in order to increase the overall cycle performance (Agnew et al., 2003; Alabdoadaim et al., 2006; Besarati et al., 2010; Zhang et al., 2009; Zhang et al., 2012a, Zhang et al., 2012b). In this cycle heated working medium at atmospheric pressure is initially expanded in the gas turbine. After that the working medium heat is recovered by the heat exchanger, and finally the cooled gas is sucked by the compressor to the atmospheric pressure. The cycle works below the atmospheric pressure and referred as sub-atmospheric cycle (Khalatov et al., 2015). It is believed that the reverse

Brayton cycle is not commercially feasibly because of the greater compressor size and employed higher operational energy (Khalatov et al., 2015). On the other hand, the M-Cycle as an innovative humidifying recuperator can significantly improve the cycle performance by providing extremely saturated hot air to the combustion chamber (before turbine) and cooled air to the compressor (after turbine) simultaneously (Buyadgie et al., 2015; Khalatov et al., 2015; Maisotsenko et al., 2004; Maisotsenko et al., 2006). Consequently, it will improve the fuel combustion efficiency as well as compressor efficiency at the same time. Furthermore, simple designs of atmospheric combustion chamber and cheaper materials could be employed in the turbine industry (Khalatov et al., 2015).

The Maisotsenko sub-atmospheric Brayton (M-SAB) cycle conception was realized recently by (Maisotsenko et al., 2004; Maisotsenko et al., 2006) in which the authors proposed various possible configurations of M-SAB cycle. On the basis of available literature, the present study discuses two kinds of M-SAB cycle which are based on: (1) compressor (Khalatov et al., 2015), and (2) ejector (Buyadgie et al., 2015). The compressor based M-SAB cycle is similar to the conventional reverse Brayton cycle. However, the compressor is replaced by the steam-air ejector in case of ejector based M-SAB cycle. The compressor based M-SAB cycles one by one.

D.4.1 Compressor based System

Khalatov et al. 2015, analyzed the compressor based M-SAB cycle while recovering the turbine waste heat. The schematic diagram of the cycle is shown in Figure D.4 (Khalatov et al. 2015; Maisotsenko et al., 2004; Maisotsenko et al., 2006). The cycle configuration is similar to the (Maisotsenko et al., 2004; Maisotsenko et al., 2006), however an additional solar energy utilization unit is proposed for pre-heating (process $1 \rightarrow 2$) in order to improve the cycle efficiency. The air is heated and humidified simultaneously by the M-Cycle assembly (process $2 \rightarrow 3$) while recovering the turbine waste heat (process $5 \rightarrow 6$). The saturated hot air improves the combustion efficiency as well as reduces the NO_x emission when used in combustion chamber (process $3 \rightarrow 4$). The combust hot gases at atmospheric conditions are expanded in the gas turbine (process $4 \rightarrow 5$). The energy from the hot gases at state (5) is recovered before it goes to the compressor by means of M-Cycle assembly (process $5 \rightarrow 6$) and an additional heat exchanger (process $6 \rightarrow 7$), which consequently improves the compressor efficiency. Analysis showed that the M-SAB cycle can achieve the thermal efficiency of 0.45 to 0.82 at pre-heating (T₂) and combustion temperature (T₄) of 40-90 °C and 160-340 °C, respectively. It is worth mentioning that the pre-heating shows significant improvement in thermal efficiency by the M-SAB cycle because of the versatile features of M-Cycle at higher temperature as explained in heading D.2. Unlike open Brayton cycle the higher regeneration rate promotes the thermal efficiency of the M-SAB cycle possesses higher efficiency as compared to conventional open Brayton cycle at certain conditions.





D.4.2 Ejector based System

Buyadgie et al. 2015, proposed the ejector based M-SAB cycle and investigated its performance for various applications. The schematic diagram of the turbo-ejector based M-SAB cycle is shown in Figure D.5 (Buyadgie et al. 2015).



Figure D.5 Schematic diagram of the ejector based M-SAB cycle (reproduced from Buyadgie et al., 2015).

The principle operation of the cycle is similar the one based on compressor as explained in Figure D.4 (Khalatov et al. 2015; Maisotsenko et al., 2004; Maisotsenko et al., 2006), though the compressor is replace by the steam-air ejector. Each process of the cycle is labelled on the Figure D.5 which gives the detail insight of the cycle. According to the results the replacement of mechanical compressor with the steam-air ejector results in 2-4 times higher power generation, and yields 15% to 20% capital cost reduction of the system. In addition, the electricity used to operate the fans for the M-Cycle assembly decreases two times per power unit. The authors concluded that the turbo-ejector M-SAB cycle design is the optimum choice when the electricity price is high and heat price is low. Furthermore, it

is more beneficial when the power generation and low temperature cooling is required simultaneously despite of the available heat cost.

From the above prospective the present study concludes that the M-Cycle possesses huge energy recovery potential in various power producing gas turbines. It addition to provide hot and humidified air for combustion, the M-Cycle recovers the turbine waste heat efficiently as compared to conventional heat exchangers. Furthermore, the nature of the M-Cycle helps to provide the cooled air to the compressor simultaneously, which increases the compressor efficiency. Another silent feature of the M-Cycle is the pollution control by reducing NO_x formation during combustion which can lead towards an environment friendly gas turbine power cycle.

D.5 Nomenclature

M-Cycle	Maisotsenko Cycle
HAT	humid air turbine
M-HAT	Maisotsenko humid air turbine
ABC	air bottoming cycle
M-ABC	Maisotsenko air bottoming cycle
M-SAB	Maisotsenko sub-atmospheric Brayton

D.6 References

- Agnew, B., Anderson, A., Potts, I., Frost, T.H., Alabdoadaim, M.A., 2003. Simulation of combined Brayton and inverse Brayton cycles. Appl. Therm. Eng. 23, 953-963. doi:10.1016/S1359-4311(03)00019-X.
- Alabdoadaim, M.A., Agnew, B., Potts, I., 2006. Performance analysis of combined Brayton and inverse Brayton cycles and developed configurations. Appl. Therm. Eng. 26, 1448-1454. doi:10.1016/j.applthermaleng.2006.01.003.
- Besarati, S.M., Atashkari, K., Jamali, A., Hajiloo, A., Nariman-zadeh, N., 2010. Multiobjective thermodynamic optimization of combined Brayton and inverse Brayton cycles using genetic algorithms. Energy Convers. Manag. 51, 212-217. doi:10.1016/j.enconman.2009.09.015.

- Bianchi, M., De Pascale, A., 2011. Bottoming cycles for electric energy generation: Parametric investigation of available and innovative solutions for the exploitation of low and medium temperature heat sources. Appl. Energy. 88, 1500–1509. doi:10.1016/j.apenergy.2010.11.013.
- Buyadgie, D., Buyadgie, O., Drakhnia, O., Brodetsky, P., Maisotsenko, V., 2015. Solar low-pressure turbo-ejector Maisotsenko cycle-based power system for electricity, heating, cooling and distillation. Int. J. Low-Carbon Technol. 10, 157-164. doi:10.1093/ijlct/ctv012.
- Cengel, Y.A., Boles, M.A., 1994. Thermodynamics: An Engineering Approach. McGraw-Hill, New York.
- Chacartegui, R., Sánchez, D., Muñoz, J.M., Sánchez, T., 2009. Alternative ORC bottoming cycles FOR combined cycle power plants. Appl. Energy. 86, 2162-2170. doi:10.1016/j.apenergy.2009.02.016.
- Clemente, S., Micheli, D., Reini, M., Taccani R., 2013. Bottoming organic Rankine cycle for a small scale gas turbine: A comparison of different solutions. Appl. Energy. 106, 355-364. doi:10.1016/j.apenergy.2013.02.004.
- Cohen, H., Rogers, G.F.C., Saravanamuttoo, H.I.H., 1996. Gas Turbine Theory 4th Edition. Reading MA: Addison Wesley Longman.
- Gallo, W.L.R., 1997. A comparison between the hat cycle and other gas-turbine based cycles: Efficiency, specific power and water consumption. Energy Convers. Manag .38, 1595-1604. doi:10.1016/S0196-8904(96)00220-8.
- Ghazikhani, M., Khazaee, I., Abdekhodaie, E., 2014. Exergy analysis of gas turbine with air bottoming cycle. Energy. 72, 599-607. doi:10.1016/j.energy.2014.05.085.
- Gillan, L., Maisotsenko, V., 2003. Maisotsenko Open Cycle Used for Gas Turbine Power Generation. ASME Turbo Expo 2003 Collocated Int. Jt. Power Gener. Conf., vol. 3, Georgia, USA. 75-84. doi:10.1115/GT2003-38080.
- Jenkins, P., Cerza, M., Saaid, M.A., 2014a. Analysis of Using the M-cycle Regenerative-Humidification Process on a Gas Turbine. J. Energy Power Eng. 8, 1824-1837.
- Jenkins, P.E., Cerza, M., Saaid, M.M. Al., 2014b. Analysis of Using the M-Cycle Regenerative-Humidification Process on a Gas Turbine. ASME Turbo Expo 2014

Turbine Tech. Conf. Expo., vol. 3A, Düsseldorf, Germany. 1-11. doi:10.1115/GT2014-25178.

- Jonsson, M., Yan, J., 2005. Humidified gas turbines—a review of proposed and implemented cycles. Energy. 30, 1013-1078. doi:10.1016/j.energy.2004.08.005.
- Kaikko, J., Hunyadi, L., Reunanen, A., Larjola, J., 2001. Comparison between air bottoming cycle and organic rankine cycle as bottoming cycles. Proc. Second Int. Heat Powered Cycles Conf. HPC. 1, 195.
- Khalatov, A., Karp, I., Isakov B., 2011. Prospects of the Maisotsenko thermodynamic cycle application in Ukraine. Int. J. Energy Clean Environ. 12, 141-157. doi:10.1615/InterJEnerCleanEnv.2012005916.
- Khalatov, A.A., Severin, S.D., Brodetsky, P.I., 2015. Maisotsenko VS. Brayton's subatmospheric inverse cycle with regeneration of output heat by Maisotsenko's cicle. Dopovidi Natsional'noi akademii nauk Ukrainy (Reports of the National Academy of Sciences of Ukraine). 1, 72-79, ISSN: 1025-6415 (in Russian language), http://dopovidi-nanu.org.ua/en/archive/2015/1/11 n.d.
- Korobitsyn, M., 2002. Industrial applications of the air bottoming cycle. Energy Convers. Manag. 43, 1311-1322. doi:10.1016/S0196-8904(02)00017-1.
- Lazzaretto, A., Segato, F., 2002. A thermodynamic approach to the definition of the HAT cycle plant structure. Energy Convers. Manag. 43, 1377-1391. doi:10.1016/S0196-8904(02)00022-5.
- Maisotsenko, V., Gillan, L., Heaton, T., Gillan, A., 2004. Power system and method. US Patent No. US20040103637 A1.
- Maisotsenko, V.S., Gillan, L.E., Heaton, T.L., Gillan, A.D., 2005. Evaporative duplex counterheat exchanger. US Patent No. US6948558 B2.
- Maisotsenko, V.S., Gillan, L.E., Heaton, T.L., Gillan, A.D., 2006. Power system and method. US Patent No. US7007453 B2.
- Pierobon, L., Haglind, F., 2014. Design and optimization of air bottoming cycles for waste heat recovery in off-shore platforms. Appl. Energy, 118, 156-165. doi:10.1016/j.apenergy.2013.12.026.

- Poullikkas, A., 2005. An overview of current and future sustainable gas turbine technologies. Renew. Sustain. Energy Rev. 9, 409-443. doi:10.1016/j.rser.2004.05.009.
- Reyzin, I., 2011. Evaluation of the Maisotsenko power cycle thermodynamic efficiency. Int.J.EnergyClean.Environ.12,129-139.doi:10.1615/InterJEnerCleanEnv.2012005808.
- Saghafifar, M., Poullikkas, A., 2014. Comparative analysis of power augmentation in air bottoming cycles. Int. J. Sustain. Energy. 1–14.
- Saghafifar, M., Gadalla, M., 2015a. Analysis of Maisotsenko open gas turbine power cycle with a detailed air saturator model. Appl. Energy. 149, 338-353. doi:10.1016/j.apenergy.2015.03.099.
- Saghafifar, M., Gadalla, M., 2015b. Analysis of Maisotsenko open gas turbine bottoming cycle. Appl. Therm. Eng. 82, 351-359. doi:10.1016/j.applthermaleng.2015.02.032.
- Saghafifar, M., Gadalla, M., 2015c. Innovative inlet air cooling technology for gas turbine power plants using integrated solid desiccant and Maisotsenko cooler. Energy. 87, 663-677. doi:10.1016/j.energy.2015.05.035.
- Wilson, D.G., Korakianitis, T., 1984. The design of high-efficiency turbomachinery and gas turbines. London, England: MIT Press, Cambridge.
- Yee, S.K., Milanovic, J.V., Hughes, F.M., 2008. Overview and Comparative Analysis of Gas Turbine Models for System Stability Studies. IEEE Trans. Power Syst. 23, 108-118. doi:10.1109/TPWRS.2007.907384.
- Zhang, W., Chen, L., Sun, F., 2009. Power and efficiency optimization for combined Brayton and inverse Brayton cycles. Appl. Therm. Eng. 29, 2885-2894. doi:10.1016/j.applthermaleng.2009.02.011.
- Zhang, W., Chen, L., Sun, F., 2012a. Thermodynamic optimization principle for open inverse Brayton cycle (refrigeration/heat pump cycle). Sci. Iran. 19, 1638-1652. doi:10.1016/j.scient.2012.09.007.
- Zhang, Z., Chen, L., Sun, F., 2012b. Energy performance optimization of combined Brayton and two parallel inverse Brayton cycles with regeneration before the inverse cycles. Sci. Iran. 19, 1279-1287. doi:10.1016/j.scient.2012.07.009.